

City of Salem

Natural Hazard Mitigation Plan





SALEM NATURAL HAZARDS MITIGATION PLAN



Source: Gary Halvorson, Oregon State Archives

Prepared by:

Oregon Partnership for Disaster Resilience

June 2002 Revised May 2003 Updated August 2008

Special Thanks & Acknowledgements

The Salem Natural Hazard Mitigation Plan was developed by the Oregon Partnership for Disaster Resilience at the University of Oregon's Community Service Center.

Project Steering Committee:

- Roger Stevenson, Salem Emergency Management
- Joe Parrott, Salem Fire Department
- Cliff Sabin, Salem Hospital
- Charlie Crawford, Salem Hospital
- Lyle Gembala, City of Salem Police
- Sara Jondahl, City of Salem Public Works- Development Services Division
- Mike Gotterba, City of Salem Public Works Emergency Preparedness Division
- Ken Roley, City of Salem Public Works Operations Division
- Steve Downs, City of Salem Public Works Utilities Planning Section
- Bryce Bishop, City of Salem Community Development Planning Division
- John Vanderzanden, Marion County Emergency Management

Project Managers:

Krista Dillon, Associate Director, Oregon Partnership for Disaster Resilience A. Gregoor Passchier, Research Intern, Oregon Partnership for Disaster Resilience Nick Kraemer, Research Intern, Oregon Partnership for Disaster Resilience

Table of Contents

1.0	INTRODUCTION	1
1.1	Purposes of This Plan	1
1.2	Hazard Mitigation	1
1.3	City of Salem	2
1.4	Planning Process Summarized	4
1.4	4.1 Development of the 2003 Salem Natural Hazard Mitigation Plan	4
1.4	4.2 Development of the 2008 Salem Natural Hazard Mitigation Plan Update	6
1.4	4.3 Development of the 2013 Salem Natural Hazard Mitigation Plan Update	7
1.5	Plan Implementation	8
1.5	5.1 Convener	8
1.5	5.2 Coordinating Body	8
1.5	5.2.1 Coordinating Body Members	9
1.5	5.3 Project Prioritization Process	9
1.6 F	Plan Maintenance	12
1.6	6.1 Continued Public Involvement	12
1.6	6.2 Five-Year Review process	12
1.7 N	Vatural Hazard Mitigation Plan Components	13
2.0	NATURAL HAZARDS	14
2.1	Principal Natural Hazard Descriptions	14
2.1	1.1 Flood Hazard	14
2.1	1.2 Earthquake Hazard	17
2.1	1.3 Landslide Hazard	20
2.1	1.4 Severe Weather Hazards	22
2.1	1.5 Urban-Wildland Interface Fire Hazard	22
2.1	1.6 Volcano Eruption Hazard	24
2.1	1.7 Drought	24
2.1	1.8 Hazardous Materials Release Hazard	25
3.0	GOALS AND OBJECTIVES	27
4.0	ALL HAZARDS RISK ASSESSMENT	28
4.1	Purpose	28
4.2	Hazard Assessments and Rankings	28
4.3	Risk Assessment Table	30
5.0	MITIGATION ACTION ITEMS	36
5.1	Updated Mitigation Action Items	36
5.´	1.1 2008 Action Item Update	36
5.1	1.2 Coordination with Existing Plans and Programs	36
5.2 A	Action Item Worksheets	37
5.2	2.1 Rationale or Key Issues Addressed:	37
6.0	REFERENCES	75
6.1	Principal References	76
APPEN	NDICES - INFORMATION AND RESOURCE COMPENDIUM	A1
Appe	endix A. Additional Federal and State Government Mitigation Information	A1
Appe	endix B. ORS 455.477 (excerpt) - Essential Facility, Major Structure, Hazardou	S
	Facility, and Special Occupancy Structures	A7

Appendix C.	Benefit-Cost Methodologies	A8
Appendix D.	2003 Salem Natural Hazards Mitigation Plan Action Items	A12
Appendix E.	2008 Plan Update Memo	A31
Appendix F.	Public Process	A39

1.0 INTRODUCTION

Salem has experienced and is vulnerable to losses caused by natural hazards. These hazards include floods, landslides, earthquakes, severe wind and ice storms, urban-wildland interface fires, volcanic eruptions, drought, and hazardous materials incidents. All hazards are potential threats capable of causing significant future life and property losses. Some events may lead to "compound disasters," such as earthquakes that trigger landslides and hazardous materials releases.

The City of Salem has benefited from its natural setting, situated in the heart of the valley with the Willamette River flowing through the center of the City and flanked by the Salem Hills and Eola Hills to the south and west.

It is a well-accepted tenet that natural activities, such as rain, cold and heat are by themselves beneficial or, at worst, benign. These activities are part of the natural functions of the ecosystem. Floods replenish floodplains with nutrients and recharge aquifers, and wildfires help preserve and restore appropriate plant life. These "hazards" do not cause problems except where humans are affected.

Disasters occur when natural hazards affect human development, especially in urban areas. We are recognizing the adverse consequences of occupying hazardous areas (such as floodplains and steep hillsides) and following construction practices that do not account for natural activities (such as earthquakes, erosion, wind, and wildfire). This lesson was brought home to Salem's residents in February and November 1996 when floods and landslides caused extensive damage to homes, businesses, public utilities, and other infrastructure.

1.1 Purposes of This Plan

The general purposes of the Salem Natural Hazard Mitigation Plan are to:

- Define the scope of and experience with natural hazards affecting the community;
- Assess the ongoing hazard mitigation activities in the community, including approaches that the City is currently utilizing;
- Evaluate additional mitigation measures that should be undertaken;
- Define a strategy for implementation of mitigation projects and activities by all stakeholders (including cities, the County, citizens, businesses, others);
- Serve as a qualifying document for various hazard mitigation programs which are coordinated through the Office of Oregon Emergency Management (OEM); and
- Act as a resource document, subject to change, as the community refines hazard mitigation goals, strategies, and implementation.

1.2 Hazard Mitigation

The Federal Emergency Management Agency (FEMA) defines mitigation as "sustained action taken to reduce or eliminate long-term risk to people and their property from hazards and their effects." Example strategies include policy changes, such as updated ordinances; projects, such as seismic retrofits to critical facilities; and education and outreach to targeted audiences, such as Spanish speaking residents, or the elderly. Mitigation is the responsibility of individuals, private businesses and industries, state and local governments, and the federal government.

Engaging in mitigation activities provides jurisdictions with a number of benefits; including reduced loss of life, property, essential services, critical facilities and economic hardship; reduced short-term and long-term recovery and reconstruction costs; increased cooperation and communication within the community through the planning process; and increased potential for state and federal funding for recovery and reconstruction projects.

The Disaster Mitigation Act of 2000 (DMA 2000) is the latest federal legislation addressing mitigation planning. It reinforces the importance of mitigation planning and emphasizes planning for natural hazards before they occur. As such, this Act established the Pre-Disaster Mitigation (PDM) grant program and new requirements for the national post-disaster Hazard Mitigation Grant Program (HMGP). Section 322 of the Act specifically addresses mitigation planning at the state and local levels. State and local jurisdictions must have approved mitigation plans in place in order to qualify to receive Pre-Disaster Mitigation and post-disaster Hazard Mitigation Grant Program funds. Mitigation plans must demonstrate that their proposed mitigation measures are based on a sound planning process that accounts for the risk to the individual and their capabilities. In addition, the plan must coordinate with floodplain management planning credit criteria for FEMA's Community Rating System (CRS), Flood Mitigation Assistance (FMA) program, and the Department of Land Conservation and Development's (DLCD) Statewide Planning Goal 7. Refer to Appendix A for updates on the Stafford Act and more information about federal and state mitigation activities, laws, and programs.

1.3 City of Salem

Local authority for hazard mitigation is principally derived from state law and local ordinances that are within the delegated powers of local government. Specific hazard mitigation plans and Interagency Hazard Mitigation Team reports often recommend local administrative and legislative actions, and other recommendations focusing on needed state legislative, regulatory, or administrative actions.

Within the City of Salem, the Salem Area Comprehensive Plan (SACP) and Map, Salem Revised Code (SRC), Oregon Model Flood Damage Prevention Ordinance (adopted by the City), and the City Public Works Department Design Standards and Standard Construction Specifications lend themselves to planning and implementation of hazard mitigation. In addition, a series of master plans and issue specific plans/studies have been developed and also apply to hazard mitigation.

Comprehensive Plan & Map. The SACP is a long-range plan for guiding development in the Salem urban area and its relationship with the greater Salem/Keizer urban area. The plan was first adopted by Salem, Marion, and Polk Counties in 1973 through a process that involved broad-based participation by area citizens. The plan was revised in 1982 to conform to State Land Use Goals. A 1986 plan amendment reduced the urbanizable area, encompassed by the Urban Growth Boundary, (UGB) by 2,400 acres. The plan has been periodically reviewed and updated since 1988. The current SACP was adopted in 1992 and revised in March 1997, November 2000, July 2002, November 2003, April 2004, and January 2005.

SACP Map designations include single family residential, multifamily residential, developing residential, commercial, Central Business District, mixed-use, industrial, industrial/commercial, employment center, parks, open space and outdoor recreation, community services, river-oriented mixed uses, and farming and resource management (applied to areas zoned "Exclusive Farm Use"). In the Comprehensive Plan, recognized "special conditions" that affect development are floodplains, geologic conditions, soils, aggregate resources, fish and wildlife, the Willamette Greenway Boundary, historic landmarks, and airspace obstructions (around McNary Field).

A 1977 report by the State of Oregon and site specific reports are referenced in the plan pertinent to geologic conditions. The State report addresses steep slopes, landslide topography, and high groundwater. Soil Conservation Service maps and a 1974 report by the State Soil Conservationist are referenced. A 1961 State report is referenced pertinent to aggregate resources. State jurisdiction over fish spawning streams (including Mill and Pringle Creeks, and Shelton Ditch), the State Fish & Wildlife Commission establishment of the Brown Island and Minto Island Wildlife Refuge, and the Audubon Society's maintenance of a 22-acre heron rookery on the north end of Minto Island are referenced as significant fish and wildlife habitat. The Willamette Greenway boundary was adopted in 1976 and delineated by the Oregon Department of Transportation (ODOT) on 1"=400 foot maps.

The Salem Revised Code (Chapter 120A) designates historic landmarks, including those on the National Historic Register, which the City may wish to protect from hazard impacts. The Salem Revised Code (Chapter 125) addresses airspace concerns.

• <u>Salem Revised Code</u>. Title I of the Code (Chapters 6 and 20) apply to the organization and implementation of the Salem Planning Commission and Design Review Board, respectively.

Title V (Community Development Standards) includes a long list of codes, of which Building Codes (Chapter 56), dangerous buildings (Chapter 56), comprehensive planning (Chapter 64) and urban growth management (Chapter 66) are a part. Chapter 64 details procedures for amending or otherwise changing the comprehensive plan and neighborhood plans, as well as lists of "detailed" plans adopted by the City. Such detailed plans pertain to master planning related to parks, water, wastewater, stormwater, the Willamette Greenway, transportation corridors, neighborhoods, and public facilities. Chapter 69 includes the Landslide Hazard Ordinance intended to guide the review of development in landslide hazard areas. Chapters within Title V also include sections on subdivisions, excavation and filling (including areas near or within designated waterways or a floodplain overlay zone), maintenance of brush that is a fire hazard, trees and vegetation.

Chapters within Title X pertain to zoning, including the effect of other regulations (Chapter 110), setbacks and hillside lots (Chapter 130), all the various zoning designations, and special "overlay" zones for floodplains (Chapter 140), the Willamette Greenway (Chapter 141), and several areas/neighborhoods. The City adopted a new Erosion Prevention and Sediment Control Program (Chapter 75), effective September 2001.

<u>Oregon Model Flood Damage Prevention Ordinance</u>. All Oregon cities utilized and adopted this model ordinance to comply with standards for participation in the National Flood Insurance Program (NFIP). This model includes standards and provisions that "encourage sound floodplain management and, if implemented, allows property owners to obtain flood insurance at a more affordable rate". The model includes a minimum requirement for non-residential construction that the lowest floor is elevated one foot above the base flood or that the lowest floor is flood proofed.

The Salem Public Works Department published a post-flood report for the 1996 February floods that identified areas of inundation during the flood and compared the City data to FEMA Flood Insurance Rate Map (FIRM) data. There were significant differences between the City's 1996 data and FIRM data. Compared with the plotted 100-year flood level on the FIRMs, 1996 flood levels documented by the City ranged from an equivalent 42 year (Willamette River at Salem)- to 143-year (Santiam River at Jefferson) flood recurrence interval. Revised FIRM data for Mill Creek was received from FEMA and review was completed in the Fall of 2002. City Council adopted the revised January 2, 2003 FIRM on March 3, 2003.

The most significant flooding occurred along Mill Creek. A 1990 draft US Army Corps of Engineers (COE) flood damage reduction study for Mill and Pringle Creeks and Shelton Ditch was inconclusive on the best option for mitigation. However, upstream storage and increasing conveyance capacity were the general solutions evaluated.

Several projects spun off of this study and principally the February 1996 flood event, including draft revisions of the FIRMs in 1998 and 2000, and the COE's *Section 205 Flood Damage Reduction*

Study and associated Hydrologic and Hydraulic Analysis. The latter effort focused on developing an updated computer hydraulic model (UNET) of Mill Creek and its tributaries as part of the comprehensive Section 205 Flood Damage Reduction Study, completed in February 2002. A version of the model included some significant modifications of the floodway and floodplain, which were reflected by the revised January 2, 2003 FIRM.

- Public Works Department Design Standards and Standard Construction Specifications. A review of the Design Standards and Standard Construction specifications (two documents) suggests that, as intended, the documents are a guide for City permit/design review staff and design consultants during the permit/design review process. The Stormwater Management Design Standards are now being updated to reflect current industry-accepted and regulatory driven practices to control both stormwater quantity and quality.
- <u>Master Plans</u>. Several master plans exist and portions of these plans directly or indirectly apply to hazard mitigation. Master plans have been developed for the stormwater, water, wastewater, transportation, and parks systems within the City, and each is periodically reviewed and updated to ensure they remain current with respect to the City's current and future infrastructure needs
- Other Plans/Reports. Other plans and reports have been prepared including the Seismic Vulnerability Evaluation of Existing Water and Wastewater Facilities, City of Salem, Oregon. This 1996 report for Public Works identified areas of relative seismic hazard. In this report, thirteen known faults in the Willamette Valley were identified that could affect the City of Salem study area. The significant local faults in the study area include the Turner Fault, the Mill Creek Fault, and the Waldo Hills Frontal Range Fault. Earthquakes of magnitude greater than 6.5 are not expected from these features. Significant ground shaking could occur within the project area due to proximity of the faults. Also, the February 28, 2001 deep-seated "subduction zone" event near Olympia is a recent reminder of the potential for larger earthquakes. The report also identified many improvements the City could make to improve the survival of sewer and water facilities in the event of an earthquake.

HAZUS-MH: Earthquake Event Reports were completed in March 2005 to estimate potential losses from two different earthquake scenarios in Polk and Marion Counties. The primary purpose of HAZUS is to provide a methodology and software application to develop earthquake losses at a regional scale. These loss estimates would be used primarily by local, state, and regional officials to plan and stimulate efforts to reduce risks from earthquakes and to prepare emergency response and recovery.

1.4 Planning Process Summarized

Determining what mitigation strategies and measures are best for an area is done through a planning process. During this process, the various hazards are inventoried, the risks from each are judged, the full range of possible loss prevention measures are reviewed, current mitigation measures are identified, and the most appropriate and affordable new ones are recommended for implementation. The following is a description of how the Salem Natural Hazards Mitigation Plan was developed.

1.4.1 Development of the 2003 Salem Natural Hazard Mitigation Plan

In March of 1998, the City of Salem formed a Technical Advisory Committee led by the Community Development Department to guide development of the Natural Hazard Mitigation Plan. Participants in the Natural Hazard Mitigation Plan Technical Advisory Committee included:

- City of Salem Fire;
- City of Salem Public Works;
- City of Salem Human Resources;
- City of Salem Police;
- City of Salem Emergency Preparedness;
- City Manager's Office;
- City of Salem Legal Departments;
- Marion County;
- Polk County;
- Mid-Willamette Valley Council of Governments;
- City of Keizer;
- Salem/Keizer School District;
- State of Oregon Department of Geology and Mineral Industries;
- State of Oregon Department of Land Conservation and Development;
- Oregon State Police Office of Emergency Management (OEM);
- State of Oregon Administrative Services; and
- Federal Emergency Management Agency.

The Technical Advisory Committee held thirteen work sessions beginning in 1998 and ending in 2001.

There were several opportunities for public involvement throughout the planning process. A total of three open houses were held at various phases in the planning process to obtain citizen feedback and provide the public with an update on the development of the Plan. The first open house, held on December 15, 1999, provided the public with an opportunity to comment on the natural hazards inventory and to provide comments regarding the goals, objectives and direction the Plan should follow. The second open house, held on March 15, 2001, provided another opportunity to obtain citizen feedback and educate interested citizens on the Plan's progress. Comments and ideas regarding possible mitigation measures were also sought at this meeting. The final open house, held on May 2, 2001, provided the public an opportunity to review and provide comments on a completed draft of the Salem Natural Hazards Mitigation Plan. Handouts, maps/graphics and staff presentations were made available at these meetings. Surveys were also used to obtain additional public comments. An example of a public outreach announcement can be seen in Appendix E Public Process.

Other opportunities for public involvement included public meetings held before the City of Salem's Planning Commission and City Council. Public meetings and planning activities were announced through press releases to local radio stations, the Salem Statesman Journal newspaper, City newsletter (*Current Issues*), the City's web site, and staff reports to City Council and Planning Commission members.

The 2003 plan was adopted by the City Council in May 2003.

Since the adoption of the Natural Hazards Mitigation Plan, the Mitigation Coordinating Committee has met twice; September 9, 2005 and April 4, 2006. At the first meeting, the Mitigation Coordinating Committee reviewed the planning and maintenance process set forth in the Natural Hazards Mitigation Plan and discussed implementation of

the action items. During the second meeting, the Mitigation Coordinating Committee evaluated the progress of the seventeen action items and delegated tasks to support the completion of actions not yet addressed.

1.4.2 Development of the 2008 Salem Natural Hazard Mitigation Plan Update

In February 2008, the City of Salem contracted with the Oregon Partnership for Disaster Resilience (the Partnership) at the University of Oregon's Community Service Center to document and facilitate the five-year update of the Salem Natural Hazards Mitigation Plan. Salem Emergency Management re-convened the Mitigation Coordinating Committee (MCC) to guide the development of the 2008 plan update. During the plan update, the MCC's title changed to the Salem Natural Hazards Mitigation Committee (the Committee). The following organizations were represented and served on the Committee during the 2008 update of the Natural Hazards Mitigation Plan:

- Salem Emergency Management
- Salem Fire Department
- Salem Hospital
- City of Salem Police
- City of Salem Public Works- Development Services Division
- City of Salem Public Works Emergency Preparedness Division
- City of Salem Public Works Operations Division
- City of Salem Public Works Utilities Planning Section
- City of Salem Community Development Planning Division
- Marion County Emergency Management

A list of those organizations invited to participate on the Committee can be found in Appendix E Public Process.

The Partnership facilitated three Plan update meetings with the Committee on March 19th, April 30th, and May 22nd, 2008. On March 19, 2008 the Partnership met with the Committee to discuss the five-year update planning process and requirements. In addition, the Committee identified new hazard history events and mitigation activities that have occurred since 2003 when the plan was first adopted. Minutes to the meeting can be found in the Appendix E Public Process.

On April 30, 2008 the Partnership facilitated the second meeting with the Committee to discuss and revise the goals and objectives of the current plan. The Committee also discussed the status of the 2003 Mitigation Plan action items and identified new action items that addressed hazards and vulnerabilities identified at the March 19, 2008 meeting. The Partnership documented the status of each of the 2003 action item, noting whether it had been completed or not completed, and why. Among the action items that had not been completed, the Committee identified several actions that should be continued in the 2008 Mitigation Plan. The Partnership revised these actions and included them with the new action items developed by the Committee and the Partnership. The new action items, together with the 2003 action items and their status, can be found in Section 5 of this plan. The 2003 actions along with status updates can be found in Appendix D.

The Partnership facilitated a third meeting with the Committee on May 22, 2008 to discuss the changes made to the 2003 Mitigation Plan and present a final draft of the 2008 Salem Natural Hazards Mitigation Plan Update. The Partnership compiled all suggestions made by the Committee and incorporated final edits into the final version of the 2008 Salem Natural Hazards Mitigation Plan Update.

On June 20, 2008, the Committee submitted the plan to the Oregon Office of Emergency Management to in turn submit to FEMA for pre-approval. The City of Salem City Council subsequently adopted the 2008 plan update on August 25, 2008.

1.4.3 Development of the 2013 Salem Natural Hazard Mitigation Plan Update

1.5 Plan Implementation

This section details the formal process that will ensure that the Salem Natural Hazards Mitigation Plan remains an active and relevant document. The plan implementation and maintenance process includes a schedule for monitoring and evaluating the Plan semi-annually, as well as producing an updated plan every five years. This section also describes how the City will integrate public participation throughout the plan maintenance and implementation process.

1.5.1 Convener

The Salem Natural Hazards Mitigation Committee identified Salem Emergency Management as the convener organization for the Salem Natural Hazards Mitigation Plan. Responsibilities of the convener include the following:

- Coordinate Committee meeting dates, times, locations, agendas, and member notification;
- Document outcomes of the Committee meetings;
- Serve as a communication conduit between the Committee and key plan stakeholders; and
- Identify emergency management-related funding sources for natural hazard mitigation projects.

1.5.2 Coordinating Body

The Salem Natural Hazards Mitigation Committee serves as the coordinating body for the Salem Natural Hazards Mitigation Plan. The Committee roles and responsibilities include:

- Coordinate the implementation of the Salem Natural Hazards Mitigation Plan;
- Monitor progress in implementing the Plan;
- Set priorities among competing action items;
- Work with lead agencies/departments identified in action items to develop implementation strategies;
- Coordinate the Plan and its implementation with relevant city, county and state plans;
- Submit an annual progress report to the Planning Commission and City Council;
- Recommend changes needed to this Plan as community needs and experiences dictate;
- Incorporate hazard assessment/mitigation goals into all City activities; and
- Utilize the Risk Assessment as a tool for prioritizing proposed natural hazard risk reduction projects.

1.5.2.1 Coordinating Body Members

The Salem Natural Hazards Mitigation Committee is composed of the following Organizations and City Departments:

- Salem Emergency Management
- Salem Fire Department
- Salem Hospital
- City of Salem Police
- City of Salem Public Works Development Services Division
- City of Salem Public Works Emergency Preparedness Division
- City of Salem Public Works Operations Division
- City of Salem Public Works Administrative Division
- City of Salem Planning Division
- Marion County Emergency Management

To make the coordination and review of the Salem Natural Hazards Mitigation Plan as broad and useful as possible, the Committee will engage additional stakeholders and other relevant hazard mitigation organizations and agencies to implement the identified action items. These organizations may include: City of Salem Public Works – Urban Development Department, City of Salem Public Works – Building and Safety Division, Salem-Keizer Public School District (24J), Polk County Emergency Management and the State Department of Administrative Services. Specific organizations have been identified as either internal or external partners on the individual action item forms found in Section 5.

The process the Committee will use to prioritize mitigation projects is detailed in the section below. The plan's format allows the City to review and update sections when new data becomes available. New data can be easily incorporated, resulting in a Natural Hazards Mitigation Plan that remains current and relevant to the participating jurisdictions.

1.5.3 Project Prioritization Process

The Disaster Mitigation Act of 2000 (via the Pre-Disaster Mitigation Program) requires that jurisdictions identify a process for prioritizing potential actions. Potential mitigation activities often come from a variety of sources; therefore the project prioritization process needs to be flexible. Projects may be identified by the Committee members, local government staff, other planning documents, or the risk assessment.

Depending on the potential project's intent and implementation methods, several funding sources may be appropriate. Examples of mitigation funding sources include, but are not limited to: FEMA's Pre-Disaster Mitigation competitive grant program (PDM), Flood Mitigation Assistance program (FMA), National Fire Plan (NFP), Community Development Block Grants (CDBG), local general or enterprise funds, and private foundations. For projects being funded locally, (i.e., through Capital Improvement Program), the project prioritization process for that particular program will be utilized. Figure 1.1 on the next page illustrates the project development and prioritization process for projects that will utilize non-local and federal funding sources.

Figure 1.1 Action Item and Project Review Process for Federally Funded Projects

Action Item and Project Review Process



Source: Community Service Center's Partnership for Disaster Resilience at the University of Oregon, 2008.

Step 1: Examine funding requirements

The Committee will identify how best to implement individual actions within the appropriate plans, policies, or programs that currently exist within the City of Salem. The Committee will examine the selected funding stream's requirements to ensure that the mitigation activity would be eligible through the funding source. The Committee may consult with the funding entity, Oregon Emergency Management, or other appropriate state or regional organizations about the project's eligibility.

Step 2: Complete risk assessment evaluation

The second step in prioritizing the plan's action items is to examine which hazards they are associated with and where these hazards rank in terms of community risk. The Committee will determine whether or not the plan's risk assessment supports the implementation of the mitigation activity. This determination will be based on the location of the potential activity and the proximity to known hazard areas, historic hazard occurrences, vulnerable community assets at risk, and the probability of future occurrences documented in the plan. Figure 1.2 shows the probability and vulnerability scores for the natural hazards addressed in this plan. Figure 1.2 also shows the City's hazard ranking score for each of the natural hazards. The higher the ranking score, the higher the priority. See Section 4.0, All Hazards Risk Assessment for an explanation of ranking methodology.

Figure 1.2. City of Salem Hazard Probability and Vulnerability Assessments								
Hazard	Probability	Vulnerability	Hazard Ranking					
Flood	High	High	9					
Severe Wind & Ice Storms	High	High	9					
Earthquake	High	High	9					
Landslide	High	Moderate	6					
Hazardous Materials Incidents	Moderate	High	6					
Urban-Wildland Interface Fires	Moderate	Moderate	4					
Drought	Moderate	Moderate	4					
Volcanic Eruptions	Moderate	Low	2					

Figure	e 1.2.	City	v of Salem	Hazard	l P	rob	ability	y and	Vulne	rability	Asses	sments	
					-							1.5	

Step 3: Committee recommendation

Based on steps one and two above, the Committee will recommend whether or not the mitigation activity should move forward. If the Committee decides to move forward with an action, the Lead Agency designated on the action item form will be responsible for taking further action and documenting success upon project completion. The Committee will convene a meeting to review the issues surrounding grant applications and to share knowledge and/or resources. This process will afford greater coordination and less competition for limited funds.

The Committee and the community's leadership have the option to implement any of the action items at any time, (regardless of the prioritized order). This allows the Committee to consider mitigation strategies as new opportunities arise, such as funding for action items that may not be of the highest priority. This methodology is used by the Committee to prioritize the Salem Natural Hazard Mitigation Plan's action items during the annual review and update process.

Step 4: Complete quantitative and qualitative assessment, and economic analysis

The fourth step is to identify the costs and benefits associated with natural hazard mitigation strategies, measures, or projects. Two categories of analysis that are used in this step are: (1) benefit-cost analysis, and (2) cost-effectiveness analysis. Conducting a benefit-cost analysis for a mitigation activity assists in determining whether a project is worth undertaking now in order to avoid disaster-related damages later. A cost-effectiveness analysis evaluates how best to spend a given amount of money to achieve a specific goal. Determining the economic feasibility of mitigating natural hazards provides decision makers with an understanding of the potential benefits and costs of an activity, as well as a basis upon which to compare alternative projects. See Appendix C for more information.

If the activity requires federal funding for a structural project, the Committee will use a Federal Emergency Management Agency-approved benefit-cost analysis tool to evaluate the appropriateness of the activity. A project must have a benefit/cost ratio of greater than one in order to be eligible for FEMA grant funding. For non-federally funded or nonstructural projects, the Committee will defer to the process identified within the appropriate plans, policies, or programs that currently exist within the City of Salem. This way, the Committee can utilize established forms of project development and prioritization processes in order to incorporate natural hazard mitigation activities into the existing framework of the City of Salem.

1.6 Plan Maintenance

Plan maintenance is a critical component of the Salem Natural Hazards Mitigation Plan. Proper maintenance of the Plan ensures that this Plan will maximize the City's efforts to reduce the risks posed by natural hazards. This section includes a process to ensure that a regular review and update of the Plan occurs. The Committee and local staff are responsible for implementing this process, in addition to maintaining and updating the plan through a series of meetings outlined in the maintenance schedule below.

The Committee will meet on a semi-annual basis to complete the following tasks. During the first meeting the Committee will:

- Review existing action items to determine appropriateness for funding;
- Educate and train new members on the plan and mitigation in general;
- Identify issues that may not have been identified when the plan was developed; and
- Prioritize potential mitigation projects using the methodology described below.

During the second meeting of the year the Committee will:

- Review existing and new risk assessment data;
- Discuss methods for continued public involvement; and
- Document successes and lessons learned during the year.

The convener will be responsible for documenting the outcome of the semi-annual meetings and including records in Appendix E Public Process. The process the Committee will use to prioritize mitigation projects is detailed in the section above. The plan's format allows the City to review and update sections when new data becomes available. New data can be easily incorporated, resulting in a Natural Hazards Mitigation Plan that remains current and relevant to the participating jurisdictions.

1.6.1 Continued Public Involvement

The City of Salem is dedicated to involving the public directly in the reshaping and updating of the Salem Natural Hazard Mitigation Plan. Although members of the Salem Natural Hazards Mitigation Committee represent the public to some extent, the public has had and will continue to have the opportunity to provide feedback about the Plan. The City will ensure continued public involvement by posting the Salem Natural Hazards Mitigation Plan on the City's website at www.salem.or.us and www.cityofsalem.net. The plan will also be archived and posted on the University of Oregon Libraries' Scholar's Bank Digital Archive at the following address:

https://scholarsbank.uoregon.edu/dspace/handle/1794/1930/browse-title.

1.6.2 Five-Year Review process

This plan will be updated every five years in accordance with the update schedule outlined in the Disaster Mitigation Act of 2000. During this plan update, the following questions will be asked to determine what actions are necessary to update the plan. The convener will be responsible for convening the Committee to address the questions outlined below.

- Do the plan's priorities align with State priorities?
- Are there new partners that should be brought to the table?
- Are there new local, regional, state or federal policies influencing natural hazards that should be addressed?
- Has the community successfully implemented any mitigation activities since the plan was last updated?
- Have new issues or problems related to hazards been identified in the community?
- Do existing actions need to be reprioritized for implementation?
- Are the actions still appropriate, given current resources, community needs, and priorities?
- Have there been any changes in development patterns that could influence the effects of hazards?
- Are there new studies or data available that would enhance the risk assessment?
- Has the community been affected by any disasters?
- Did the plan accurately address the impacts of this event?

The questions above will help the Committee determine what components of the Mitigation Plan need updating. The Committee will be responsible for updating any deficiencies found in the plan as needed.

1.7 Natural Hazard Mitigation Plan Components

This Mitigation Plan consists of four more sections and supporting appendices. Section 2.0 provides further information about natural hazards of concern to the City of Salem. Section 3.0 defines mitigation goals and objectives. Section 4.0 contains the risk assessment for the Salem Natural Hazards Mitigation Plan. Section 5.0 focuses on which actions should be taken, the rationale for why they should be taken, who the lead agency and partners will be, and a timeline for implementation. The actions in Section 5.0 include some actions found in the 2003 Mitigation Plan and new actions developed during the 2008 update process.

2.0 NATURAL HAZARDS

2.1 Principal Natural Hazard Descriptions

This plan addresses multiple natural hazards that have or could affect portions of the State of Oregon and the City of Salem. The State of Oregon shares multiple hazards with its neighboring states, and has experienced several presidentially declared disasters in recent years. While some hazards present different risks in terms of their frequency and potential impacts, the full range of hazards includes floods, earthquakes, landslides, severe weather hazards, urban-wildland interface fires, volcanic eruptions, drought, and hazardous material releases.

Lost lives, injuries, property damages, service interruptions, and extraordinary costs have been the hallmarks of several recent Oregon disasters. Most of these have resulted in presidential disaster declarations, but other more frequent smaller scale losses occur which might result in local or state declarations of emergencies. While difficult to estimate, total aggregate losses state-wide due to the occurrence of natural hazards have amounted to hundreds of millions of dollars during the last decade. For example, nearly \$222 million was provided to Oregon by several federal programs for three flood disasters in the last decade (1996/7). This plan is designed to help reduce such losses and their costs to the public and private sectors.

The following is a brief summary of the hazards affecting the City of Salem. Several of the natural hazards concerns have been mapped by the City of Salem. These include floodways, floodplains, areas inundated during the 1996 floods, and areas of high landslide hazard. While preparing the Natural Hazards Mitigation Plan, several other map "overlays" of information from State sources were added to the City's map base. Such information includes mapped areas of relative vulnerability to an earthquake (e.g., soil liquefaction, relative earthquake hazard, and earthquake amplification hazard). The maps are included in this section and follow the corresponding hazard.

The maps in this section identify the location and extent of the majority of the hazards. For more detailed information on the history, probability, vulnerability and impacts of these hazards, see Section 4.3 Risk Assessment Table.

2.1.1 Flood Hazard

Flooding results when rain or snowmelt creates water flows that exceed the carrying capacity of rivers, streams, channels, ditches, and other watercourses. In Oregon, flooding is most common from October through April when storms from the Pacific Ocean bring intense rainfall. Flooding can be aggravated when rain is accompanied by snowmelt and frozen ground. In general, Oregon is subject to two principal types of flooding: riverine and local drainage. Riverine flooding on the larger rivers and streams usually results from large storms or prolonged wet periods. Local drainage flooding occurs along the smaller streams, creeks, and drainage ways, and is more likely to result from heavy local storms and debris-clogged storm drainage systems. The areas subject to riverine flooding have been mapped by the Federal Emergency Management Agency (FEMA) for the National Flood Insurance Program (NFIP). The national and state flood mapping standard is the 100-year or base floodplain.

In May of 2008, NFIP's Community Rating System (CRS) improved the City of Salem's rating to a Class 8. The NFIP's CRS is a voluntary incentive program that recognizes and encourages community floodplain management activities that exceed the minimum NFIP requirements. The CRS program looks at these different activities and rates those compared to a national standard. As a result of the City of Salem improving to a Class 8 rating in the CRS, flood insurance premium rates for Salem residents are discounted 10% to reflect the reduced flood risk resulting from the City's flood hazard mitigation actions. For CRS participating communities, flood insurance premium rates are

discounted in increments of 5%; i.e., a Class 1 community would receive a 45% premium discount, while a Class 9 community would receive a 5% discount.

Riverine flooding is the most common type of flooding in Salem. The most significant of the FEMA-determined floodplains and floodways either surround the southern side of the Willamette River west of Salem, or are within the greater Mill Creek/Pringle Creek watershed. The Mill Creek area includes residential and commercial/industrial developments to the southeast of downtown. Please see Map 2: Flood Information on the following page. Localized flooding is also a hazard, especially in Keizer which is located adjacent to Salem on the north and downstream from Salem within the Claggett Creek watershed. The City of Salem has a total of 3 repetitive flood loss properties within City boundaries. The City has also identified the number of structures located in the various floodplains by land use zone, as shown in Figure 2.1.1.

Figure 2.1.1. Number of Structures in Floodplain by Land Use Zone.						
		Count of				
Zone	Zone Name	Structures				
CB	Central Business District	24				
CG	General Commercial	122				
CO	Commercial Office	25				
CR	Retail Commercial	103				
FMU	Fairview Mixed Use	1				
IBC	Industrial Business Campus	3				
IC	Industrial Commercial	201				
IG	General Industrial	351				
IP	Industrial Park	36				
PA	Public Amusement	39				
PE	Public and Private Educational Services	31				
PH	Public and Private Health Services	23				
PM	Capitol Mall	4				
PS	Public Service	111				
RA	Residential Agriculture	39				
RD	Duplex Residential	141				
RH	Multiple Family High-Rise Residential	36				
RM1	Multiple Family Residential 1	3				
RM2	Multiple Family Residential 2	1406				
RS	Single Family Residential	842				
Total		3541				

Source: City of Salem Public Works - GIS, 2008.

CITY OF SALEM NATURAL HAZARD MITIGATION PLAN

MAP 2.1.1: FLOOD INFORMATION



2.1.2 Earthquake Hazard

Oregon sits on the Cascadia Subduction Zone where the Juan de Fuca Plate is sliding under the North American Plate. While earthquakes along this zone occur infrequently, plate movement can produce major earthquakes. In addition, the western part of Oregon is underlain by a large and complex system of faults (e.g., Portland Hills) that can produce damaging earthquakes. There is a direct relationship between a fault's length and its ability to generate damaging ground motions: smaller nearby faults produce lower magnitude events, but their ground shaking can be strong and damage can be high because of the fault's proximity. In contrast, offshore subduction zone events can generate earthquakes with great magnitudes, but because of their distance and depth may result in only moderate shaking in western Oregon. Earthquakes can trigger other geologic and soils failures that contribute to damage.

The Oregon Department of Geology and Mineral Industries (DOGAMI) has mapped earthquake hazards in much of the Salem area. The maps show liquefaction, ground motion amplification, landslide susceptibility, and relative earthquake hazards. As a result of the 1996 study completed by Dames and Moore, the City has completed retrofits to all their pump stations and steel water reservoirs. The City has also installed seismic activated valves on most of the City's water reservoirs which will automatically close in the event of an earthquake.

Within the Salem Urban Growth Boundary (UGB), the area south of the Willamette River and west of River Road has the highest risk of earthquakes. Other small areas with high earthquake risk exist to the east of the city. More information about earthquake risk in Salem can be seen in Map 3A: Earthquake Hazard – Amplification Susceptibility and Map 3B: Earthquake Hazard – Liquefaction Susceptibility on the following pages.

CITY OF SALEM NATURAL HAZARD MITIGATION PLAN MAP 2.1.2A: EARTHQUAKE HAZARD - AMPLIFICATION SUSCEPTIBILITY

[ז Legend Major Streets Salem City Limits Urban Growth Boundary City of Keizer \square Water Amplification Susceptibility Low The hazard locations portrayed on these maps are for City of Salem planning purposes only. The City is not responsible for errors or omissions. Every effort has been made to ensure the accuracy of these maps. However, there may be inaccuracies due to human or Moderate High Very High mechanical error or changes in the hazard maps resulting from updated information.

CITY OF SALEM NATURAL HAZARD MITIGATION PLAN

MAP 2.1.2B: EARTHQUAKE HAZARD - LIQUIFACTION SUSCEPTIBILITY



2.1.3 Landslide Hazard

Many hillsides, especially in Western Oregon, are unstable and vulnerable to landslides, debris, and mudflows. These can result from ground saturation, runoff, improper or poorly designed or maintained drainage systems or earthquakes. Landslides are natural processes that tend to reduce the height and slope of mountains and ridges, and are part of the normal ongoing process of smoothing topographical high points. Slides occur in natural materials and in placed fill materials. The process is simple: a mass of earth slides when the forces from the weight of the slide mass exceed the strength of the material holding it in place. Determining when and where sliding will occur is difficult. Landslides and mudflows are especially prevalent when prolonged heavy rainfall saturates the soil and rock, and when human activities steepen the slopes, remove the toes, or add weight or water to the slopes.

Landslides and mudslides in the City of Salem's drinking water supply watershed have negatively impacted the surface water quality of the North Santiam River, which prevented the use of the City's slow sand filters for drinking water treatment. The City installed a new pre-treatment facility in 1998 to reduce the potential impact of surface water quality, and coordination efforts for activities within the North Santiam watershed have been greatly improved among affected stakeholders (USFS, BLM, Oregon Department of Forestry, North Santiam communities, and private parties).

All areas having a medium to high landslide susceptibility rating are located to the west of downtown Salem in two main locations. The first area is in West Salem and is bounded to the south by the Willamette River, east by Wallace Rd and north and west by the City's Urban Growth Boundary (UGB). The majority of the landslide risks occur around Gibson Creek and Turnage Brook. To the south of the river, the highest landslide risk is around Croisan Creek and westward in the hills south of South River Road. Various areas of low risk are located south of Salem along the southern boundary of the UGB. Please see Map 4: Landslide Susceptibility Study Information on the following page. In 2005, a landslide occurred along South River Road, near South Owen Street, which blocked off traffic to the city, but did not damage any homes. The City has undertaken several projects along South River Road to divert water away from the top of the slope, screen and contain falling debris, reduce vegetation, and remove hazardous trees to decrease the vulnerability of landslide hazards in the area.

CITY OF SALEM NATURAL HAZARD MITIGATION PLAN

MAP 2.1.3: LANDSLIDE SUSCEPTIBILITY STUDY INFORMATION



2.1.4 Severe Weather Hazards

This broadly defined hazard category includes windstorms, severe winter hazards (e.g., snow, ice, extended cold periods), thunderstorms, and tornadoes that disrupt vital systems and threaten lives and property. Wind and ice storms are caused by severe weather conditions. Windstorms can occur at any time of the year while ice storms are confined to the winter months. Wind storms, occasionally associated with Pacific Coast hurricanes migrating north, usually do not exceed 90 miles per hour, but wind speeds of 119 miles per hour were recorded in Portland during the October 1962 Columbus Day storm. Ice storms can be accompanied by high winds, and they have similar impacts, especially to trees, power lines, and residential utility services. These events can affect many areas. Severe ice storms occur more frequently in areas exposed to east wind patterns through the Columbia River Gorge, particularly eastern Multnomah and Clackamas Counties.

The mid-Willamette Valley area has had several wind and winter storm-related events that have affected Salem over the past five years. Funnel clouds occurred in the area in 2001, 2002, 2003, and 2006. Major hail storms occurred in 2003 and 2007. Salem has also been affected by windstorms (2002) and snow and ice storms (2003-2004). Severe weather patterns do not remain stagnant over Salem and are therefore usually short events. The location and extent of severe weather hazards depends on the event. They can impact the entire city, or just portions.

2.1.5 Urban-Wildland Interface Fire Hazard

While more common to the arid areas of central and eastern Oregon, the potential for losses due to urban-wildland interface fires in the urbanized region should not be ignored. Numerous residences are located in the heavily wooded hillsides around Salem and the trend of people locating in or near forested lands continues. Frequently occurring fires in such heavily wooded areas are not a natural occurrence, but the threat increases when subject to more human activity. The State of Oregon has noted that such interfaces really are an intermingling of homes and other structures at various densities and complexity within areas of heavy natural cover or forestlands. When buildings burn in or close to areas of heavy vegetation, especially during the dry months, the risk increases. Areas where structures are built in proximity to dense vegetation may be vulnerable to urban-wildland interface fire. Of particular concern are areas with narrow roadways and few routes of egress and ingress.

In Salem, the areas with the highest risk are characterized by steep slopes, combustible fuels, and values at risk (meaning valuable property adjacent to wildland areas). The majority of the high risk area is along the southern boundary of the UGB, with additional interface risks at the northwestern UGB boundary and in Keizer. Please see Map 5: Wildfire Risk Areas on the following page. Most wildland urban interface areas are adjacent to the Salem Suburban Fire Department's jurisdiction.

CITY OF SALEM NATURAL HAZARD MITIGATION PLAN

MAP 2.1.5: WILDFIRE RISK AREAS



2.1.6 Volcano Eruption Hazard

Evaluation of the volcano eruption hazard is primarily that of predicting lahars (mud flow) paths based on topography and ash fall patterns based on prevailing wind patterns. Prediction of eruptions has only recently become a reality in some cases where a volcano is instrumented and magma movement can be detected by seismic activity.

Salem could be affected by volcanic activity from Mt. St. Helens, Mt. Hood, or Mt. Jefferson. If any of these volcanoes erupted, there would be a possibility of ash that could affect air quality and/or the water quality. Specifically, Salem's North Santiam watershed could be severely impacted by mudflows and volcanic ash falls derived from regional volcanic activity.

2.1.7 Drought

Drought is frequently an "incremental" hazard, the onset and end are often difficult to determine. Also, its effects may accumulate slowly over a considerable period of time and may linger for years after termination of the event. Droughts are not just a summer-time phenomenon in Oregon; winter droughts can have a profound impact on agriculture, particularly east of the Cascade Mountains. Also, below average snowfall in higher elevations has far-reaching affects, especially in terms of hydro-electric power, municipal water supply, irrigation, recreational opportunities and a variety of industrial uses.

Although Salem is spared from most droughts because of its location east of the ocean and west of the Cascades, it has been affected by droughts in the past. Between 1928 and 1941, there was a statewide drought. Low stream flows prevailed in western Oregon during the period from 1976-81, with 1976-77 being the driest year of the century. The 1985-94 drought was not as severe as the 1976-77 drought in any single year, but the cumulative effect of ten consecutive years with mostly dry conditions caused statewide problems. The peak year of the drought was 1992, when a drought emergency was declared for all of Oregon.

The City of Salem Public Works Department has a Water Management and Conservation Plan that addresses drought and sets forth a plan of action to reduce the negative impacts from a drought. The drought hazard's location and extent includes the entire community as drought typically has community wide impacts.

2.1.8 Hazardous Materials Release Hazard

Chemical and other man-made disaster/hazard issues are generally handled through the local Fire Department, and the City of Salem is no exception. For the purposes of the Natural Hazards Mitigation Plan, hazardous materials releases are considered a secondary hazard derived from the impact of a natural hazard event (e.g., flooding in a chemical storage area could results in toxic levels of chemicals in water or air).

The Office of State Fire Marshal utilizes a hazardous materials database and providing it to local fire departments. This database includes information on chemicals stored by address with name/phone of a contact person. The sites shown on Map 1 are the more intense uses hazardous materials facilities. The City of Salem Environmental Services Section also maintains a vast database (e.g., underground fuel tanks, waste generators, contaminated properties, etc.). These and other databases are linked to addresses of sites that use/generate hazardous materials/waste.

In Salem, specific places have higher than average risks for hazardous material releases. In particular, trucking routes along I-5 and Highway 22 that run through Salem are vulnerable because of the quantity of materials transported along these routes. Also, the railroad lines that run through downtown Salem near the Capital area are a concern because they carry significant quantities of hazardous materials transported through Salem each year.

The City of Salem Fire Department and Public Works have utilized the information in these databases and have a fullcapacity hazmat response team to respond to hazardous materials incidents.

CITY OF SALEM NATURAL HAZARD MITIGATION PLAN MAP 2.1.8: IMPORTANT FACILITIES AND HAZARDOUS MATERIALS LOCATIONS CLASSIFIED EHS



3.0 GOALS AND OBJECTIVES

3.0 GOALS

The City of Salem is vulnerable to a variety of hazards, including both chronic and catastrophic hazards. This section sets forth goals for subsequent activities to mitigate the threats from those hazards. The City of Salem Natural Hazard Mitigation Plan goals are as follows:

Goal 1. Develop and implement mitigation activities to protect human life.

Goal 2. Protect existing buildings and infrastructure as well as future development from the impacts of natural hazards.

Goal 3. Strengthen communication and coordination of public and private partnerships and emergency services among local, county and regional governments and the private sector.

Goal 4. Enhance economic resilience to reduce the impact on the local economy.

Goal 5. Preserve and rehabilitate natural systems to serve natural hazard mitigation functions and protect natural resources.

The Salem Natural Hazard Mitigation Committee recognizes that the Mitigation Plan's goals represent what the City of Salem hopes to accomplish, while its actions define how the goals will be met. There are a number of methods available in a community's "mitigation toolbox" for reducing risk. The Salem Natural Hazard Mitigation Committee has identified a comprehensive range of actions that can be utilized to reduce risk, including the following:

- Research & development;
- Education & outreach;
- Partnership & coordination;
- Risk assessment activities;
- Emergency services;
- Policy development;
- Property protection & structural projects; and
- Professional development & training.

4.0 ALL HAZARDS RISK ASSESSMENT

4.1 Purpose

The foundation of the Salem Natural Hazards Mitigation Plan is the risk assessment. Risk assessments provide information about the areas where the hazards may occur, the value of existing land and property in those areas, and an analysis of the potential risk to life, property, and the environment that may result from natural hazard events.

This section identifies and profiles the previous occurrences, and future probability of natural hazards that can impact the community. The following natural hazards are addressed in the risk assessment:

- Flood
- Landslide
- Earthquake
- Volcanic Eruptions
- Severe Wind & Ice Storms
- Urban Wildland Interface Fires
- Drought
- Hazardous Materials

Section 4.3 includes risk assessment information from the 2003 Natural Hazard Mitigation Plan and new information gathered from the March 19, 2008 Salem Natural Hazards Mitigation Committee meeting. More detailed information about new hazard history and probability and vulnerability assessments can be found in the minutes to the March 19 meeting in Appendix E Public Process.

4.2 Hazard Assessments and Rankings

All of the hazards included in this plan are of concern to the City of Salem. Some hazards, however, are of significantly lower concern than others. During the 2003 plan-development process, the Technical Advisory Committee (TAC) ranked all seven primary hazards in terms of overall vulnerability and then assigned a weighted vulnerability (low, medium, or high) to each of the hazards. In the 2008 Plan update, the Committee reviewed and updated the 2003 TAC rankings with additional information provided by the Oregon Partnership for Disaster Resilience. As shown in the Risk Assessment table in Section 4.3, hazards are given probability and vulnerability ratings.

Probability scores address the likelihood of a future major emergency or disaster within a specific period of time, as follows:

High = One incident likely within a 10 to 35 year period.

Moderate = One incident likely within a 35 to 75 year period.

Low = One incident likely within a 75 to 100 year period.

The vulnerability scores address the percentage of population or regional assets likely to be affected by a major emergency or disaster, as follows:

High = More than 10% affected Moderate = 1-10% affected Low = Less than 1% affected

During the March 19, 2008 Hazard Identification meeting, the Committee identified probability and vulnerability scores, using the definitions described above, to each of the natural hazards addressed in this plan. The Committee members based their scores on historical information, experience, and probability and vulnerability scores found in the

Marion and Polk County Natural Hazards Mitigation Plans. Once Committee members identified the probability and vulnerability scores, the scores were assigned a numerical value, with a high score having a value of 3, a moderate score a value of 2, and a low score a value of 1. The probability and vulnerability scores were multiplied together to determine the natural hazard ranking, which can be found in the last row of the Risk Assessment in Section 4.3. The higher the score, the higher the hazard priority. Floods, earthquakes, and severe wind and ice storms are of primary concern to the City of Salem. Landslides and hazardous materials are of secondary concern, and wildfire and drought are of lesser concern. Lastly, volcanic eruptions are of little concern to the city.

NATURAL HAZARD MITIGATION PLAN City of Salem

4.3 Risk Assessment Table

Salem Natural Hazard Mitigation	n Plan		AL	L-HAZARD ASSESSME	NT					
	HAZARD									
QUESTION	Floods	Landslides	Earthquakes	Volcanic Eruptions	Severe Wind & Ice Storms	Urban-Wildland Interface Fires	Drought	Hazardous Materials Incidents		
What is the continuing threat?	Buildings, fill, and other restrictions in the floodplain of Mill, Pringle and Claggett Creek, Willamette River, and other urban streams.	Existing development on historic/ancient slides, development exacerbating marginally stable slopes	The threat of the "big one" (maximum credible event)	Minimal	Overhead utilities, tree limbs, and falling trees.	Northwest & southwest areas, field and forest fires adjacent to the UGB (especially to the east and west, see Map 2.1.5) Areas with the highest risk are on steep slopes, have combustible fuels, and are near valuable property	Dry periods in the summer time occasionally reduce drinking water levels in rivers.	Releases especially during floods, earthquakes, and storms; accidents/spills, resulting fish kills, stream pollution, groundwater impacts, and public exposure.		
What triggered concern about this hazard originally?	1996 flood damage and evacuations	1996 slide damage and lawsuits	1990's quakes	Mt. St. Helens eruption in 1980 impacted the City's water treatment plan	Seasonal events, especially December 1995	Previous drought conditions and short-term hot/dry weather	Previous drought conditions, especially in 1976-1981 and 2005	Willamette River basin studies of fish, local spills and accidents		
What are the previous occurrences for this hazard? (Any commonly known history occurrences, and all disasters that resulted in a local, state or federal disaster declaration.) (continued on next page)	Dec. 29, 1937 Mill Creek Flood - 2 nd largest peak flow on Mill Creek on record. Dec. 24, 1964 Christmas Flood, Rated "approximately a 100 yr. Flood" by FEMA and probably the most damaging in Oregon history. Over \$157 million in losses, seven lives lost and thousands homeless. An unusually early and heavy snowpack in the lower elevations followed by warm weather and heavy rains combined to crest the Willamette River above flood levels. Most of the streams in	Feb. 1996 South Salem, Damage to five homes near Heath Street and Aldous Avenue in Dorchester Heights. 1997 West Salem, Damage to several homes on Gibsonwoods Court. 2005 South River Road landslide near S. Owen Street, blocked	July 19, 1930: magnitude 5.0, 14.5 km NW of Salem. Aug. 19, 1961: magnitude 4.1, 50.2 km SE of Salem, caused minor damage at Albany and Lebanon.	1980 Mt. St. Helens. 2004 small eruption on Mt. St. Helens.	1962 Columbus Day Storm, Effects included more than 38 deaths, power outages for 2-3 weeks, damage to over 50,000 dwellings and \$170-200 million (\$800 million today) in damages. Salem sustained winds of 58 mph and a peak gust of 90mph. Jan. 1950 Salem recorded 32.8 in of snowfall. Freezing	Recent acreage fires that threatened homes. 1989 South River Road 1991Skyline Chinook Estates 1993 Eola Hills 1996 Viewcrest Fire 2000 Battlecreek Fire 2001 Landau Fire 2002 Minto Brown	1928-1941 All of Oregon experienced drought. 1976-1981 Very dry period with low stream flows 1985-1944 Statewide problems due to drought	1967 Berheimer Fire with pesticides. 1968 Star Ice, ammonia leak and explosion. 1975 Ammonia railcar leak at Boise Cascade near City Hall. 1975 Major oil spill at Boise Cascade into Willamette R.		

August 2008

NATURAL HAZARD MITIGATION PLAN City of Salem

Salem Natural Hazard Mitigatior	AL	L-HAZARD ASSESSME	NT					
	HAZARD							
QUESTION	Floods	Landslides	Earthquakes	Volcanic Eruptions	Severe Wind & Ice Storms	Urban-Wildland Interface Fires	Drought	Hazardous Materials Incidents
(continued from previous page) What are the previous occurrences for this hazard? (Any commonly known history occurrences, and all disasters that resulted in a local, state or federal disaster declaration.)	Salem overflowed their banks. Jan. 11-16, 1974 Mill Creek flooded some residences and businesses. Demonstrated that the City of Salem is susceptible to_significant flooding during a flood with a return frequency of less than 20yrs. 1992 Summer "Flash" Flood on Clark Creek Feb. 1996, The combination of record-breaking rain, warm temperatures, and a deep snowpack led to the most severe flooding along the Salem area streams ever recorded. Federal and State disaster relief distributed \$1.7 million in grants, loans, and insurance claims to almost 2,000 applicants. Total damages from the flood have not been calculated. Nov. 1996/Jan. 1997 - Localized flooding on Mill Creek and in Keizer	traffic.	March 25, 1993: magnitude 5.6, Scotts Mills, 21 miles NE of Salem. It was Marion County's largest earthquake in recent recorded history and cause \$25-30 million in damages. No serious injuries were reported. February 8, 1995: magnitude 3.6, 12 km E of Woodburn November 25, 1999: magnitude 3.5, 7 km SE of Woodburn, OR		rain downed many trees and power lines and caused thousands of dollars in damage. 1968 Large snow event in Salem Dec. 12, 1995 Gov. declared State of Emergency for Western OR. Severe winds, over 2ft. of rain and mudslides caused damages including loss of life, lost roofs, widespread power outages (est. 250,000 customers) and many downed trees blocking roadways and taking down utilities lines. December 2007, strong windstorm along the coast and to a lesser extent in the Willamette Valley, but no significant damage in Salem. Snow and ice storms are annual occasions, recently in Jan. 2008	2006 Fire started by fireworks burned 14 acres of a field in South Salem by Battle Creek Road.	2000-2001 general statewide and national drought 2005 Driest period on record for western Oregon since 1977	 1979 Chlorine railcar at Salem Industrial Dr. 1983 PCB transformers at 14th and Mill. 1986 Gasoline spill, 41,700 gallons in Mill Creek, prevented spill from entering the Willamette River. 1988 State Pen. Fire with hazmat 1995 I-5 & Mill Creek vinyl acetate truck accident. 1992 Butadiene adhesive truck accident. 2002 Gasoline tanker truck overturned on I-5, 11,000 gallons contained, 300 spilled 2004 gas spill occurred near Detroit Reservoir
Salem Natural Hazard Mitigation Plan ALL-HAZARD ASSESSMENT								
--	---	---	---	--	---	--	---	--
	HAZARD			-		-	-	
QUESTION	Floods	Landslides	Earthquakes	Volcanic Eruptions	Severe Wind & Ice Storms	Urban-Wildland Interface Fires	Drought	Hazardous Materials Incidents
What happened as a result of previous events?	Section 205 study and Stormwater Master Plan (2000), Land use restrictions, and update of FEMA FIRM in Mill Cr./Shelton Ditch/Lower Pringle system. Digitizing of FIRM maps. Streambank Restoration and Flood Mitigation projects under various Federal programs. Installation of flood alarm systems. Revised City Emergency Response Plan and EOC Training. Inclusion into the Community Rating System (CRS) in May 2008 as an 8. ODOT raised I-5 at Mill Creek to avoid flooding. Bridge replacements at Mill Creek and Center Street, Shelton at 25 th . Annual stream cleanups and Public Service Announcements to help with education and outreach. Upsizing culverts in West Salem. Developed an evacuation plan for mobile home park.	Slide mapping and landslide ordinance regulating development Inclinometers are used to measure movements of hillsides, such as along Heath St. Reservoirs are being relocated away from landslide hazards. Erosion control program (SRC 75) developed in 2001.	DOGAMI & USGS area mapping. Mid-1990's seismic risk assessment for City of Salem water and sanitary sewer system. Salem Fire Station upgraded to meet seismic codes. CERT provides earthquake information to the public. Operations Section has a new Emergency Operations Center (EOC). 911 and EOC centers are out of City Hall to a seismically safe location.	State studies Pre-treatment facility at the water treatment plant to prevent ash from infiltrating water.	Revised draft of Emergency Response Plan (severe weather annex) New radio tower on Skyline Road designed to withstand high winds.	No significant events Improved water access to West Salem by taking over Illahe and Orchard Heights private water for fire fighting.	Low stream flows during drought conditions	Better understanding of hazardous materials storage sites at risk. Improved hazardous materials emergency response capability in January 2005

Salem Natural Hazard Mitigation	Ilem Natural Hazard Mitigation Plan					ALL-HAZARD ASSESSMENT				
	HAZARD									
QUESTION	Floods	Landslides	Earthquakes	Volcanic Eruptions	Severe Wind & Ice Storms	Urban-Wildland Interface Fires	Drought	Hazardous Materials Incidents		
(continued on next page)										
(continued from previous page) What happened as a result of previous events?	Mutual aid agreements between other cities, Marion County, and the State. Experimenting with porous pavement in the Pringle Creek Community to decrease stormwater runoff. Construction of a new hospital farther from flood hazards and with parking on ground floor. Construction of a retention wall along Pringle Creek. Completion of a June 2004 Regional Detention Feasibility Study for three sites on West Pringle, Croisan, and Glenn Creeks (1 each)		Most unreinforced masonry buildings are brought up to seismic codes when redeveloped. Bridge upgrades. New hospital built to seismic standards. School district has a construction plan for retrofit. City-owned Bush Barn Art Center is being seismically upgraded.							
Could losses from this hazard increase in the future? Why?	Less likely due to land use restrictions in flood prone areas. Also increased community awareness	Less likely with increasing controls on land use and application of the new landslide ordinance	As development continues in high risk areas, many structures are not up to code	Yes, as area grows	Yes, as area grows	Yes, as development encroaches into wildland areas, risk will become greater.	Yes, as area grows, more pressure is put on local water resources.	Yes, as industrial development continues in high hazard areas and as population grows near hazardous materials storage		

Salem Natural Hazard Mitigatior	em Natural Hazard Mitigation Plan					ALL-HAZARD ASSESSMENT				
	HAZARD									
QUESTION	Floods	Landslides	Earthquakes	Volcanic Eruptions	Severe Wind & Ice Storms	Urban-Wildland Interface Fires	Drought	Hazardous Materials Incidents		
								sites.		
What has been done to prevent losses to future development? What has been done to prevent losses to future development? (continued from previous page)	Guide future development during permit review process. Pursuant to adoption of the Stormwater Master Plan, a Stormwater Management Agreement has been agreed to by Salem, Keizer, and Marion County; and a staff- level working group meets as needed to address issues of mutual concern, including flooding.	Landslide ordinance and use of DOGAMI landslide hazard maps during permit application review.	Possible indirect benefit through application of landslide hazard maps during the permitting process.	Nothing	Undergrounding utilities and limiting use of certain tree species Downtown utility master plan being developed. Cleaning of hazardous trees in Salem creeks and parks.	Building ordinances (fire resistance roofing materials) Public education in wildland fire risk mitigation	Improved standards for streets such as porous streets in the Pringle Creek subdivision.	Fire Dept uses the State Fire Marshal's database and deploys hazmat response teams Monitor materials to prevent accidents and develop an effective response system.		
What is being done to better protect existing development?	Planning for flood detention, stream monitoring, stream/habitat improvement, conveyance, and prioritization of same. Adoption of Erosion Prevention and Sediment Control ordinance to keep soils on-site and out of the drainage system, which preserves conveyance capacity and reduces threat of localized flooding.	See above	On-going initial inventory of unreinforced masonry buildings in downtown area.	Nothing	Tree maintenance/removal Backup generators installed at critical facilities including EOC and pump stations.	Grass mowing, nuisance ordinances Public education	The City is currently doing water conservation outreach within the community through the City's Environmental Protection Assistance Program	Fire Dept regularly monitors hazardous materials in businesses Developed GIS mapping systems to predict which areas need to be notified at a certain flood stage		
Probability Rating	High	High	High	Moderate	High	Moderate	Moderate	Moderate		
Vulnerability Rating	High	Moderate	High	Low	High	Moderate	Moderate	High		

Page 34

Salem Natural Hazard Mitigation Plan				AL	LL-HAZARD ASSESSMENT				
	HAZARD								
QUESTION	Floods	Landslides	Earthquakes	Volcanic Eruptions	Severe Wind & Ice Storms	Urban-Wildland Interface Fires	Drought	Hazardous Materials Incidents	
Hazard Ranking	9	6	9	2	9	4	4	6	

5.0 MITIGATION ACTION ITEMS

5.1 Updated Mitigation Action Items

This section describes the components that guide implementation of the identified mitigation strategies and is based on strategic planning principles. Short and long-term action items identified through the planning process are an important part of the mitigation plan. Action items are detailed recommendations for activities that local departments, citizens and others could engage in to reduce risk. They both address multi-hazard (MH) and hazard-specific issues.

Action items included in this plan address the following natural and man made hazards that can affect the City of Salem:

- Flood
- Landslide
- Earthquake
- Volcanic Eruptions
- Severe Wind & Ice Storms
- Urban Wildland Interface Fires
- Drought
- Hazardous Materials

Given the relative low probability of occurrence and cost effectiveness of mitigation activities, the drought and volcanic eruptions hazards do not have any hazard specific action items, but are addressed within the multi-hazard action items.

5.1.1 2008 Action Item Update

The Salem Natural Hazards Mitigation Committee, together with the Partnership for Disaster Resilience, developed the action items presented in this plan. The action items are a combination of revised action items from the 2003 Mitigation Plan, and new action items that address hazards and vulnerabilities identified during the 2008 Salem Natural Hazard Mitigation Plan update process. Action items that are taken from the 2003 Mitigation Plan reference the 2003 action items, which are found in Appendix D. The 2003 action items in Appendix D include documentation of the status of each action and indicate whether or not the action has been completed or if it has been revised into a new action item.

5.1.2 Coordination with Existing Plans and Programs

The Salem Natural Hazards Mitigation Plan includes a range of action items that, when implemented, will reduce loss from hazard events in the City. Within the plan, FEMA requires the identification of existing programs that might be used to implement these action items. The City of Salem currently addresses statewide planning goals and legislative requirements through its comprehensive land use plan, various master plans, capital improvements plan, mandated standards and land use development, and building codes. To the extent possible, the City of Salem will work to incorporate the recommended mitigation action items into existing programs and procedures.

Where possible, the City of Salem should implement the Natural Hazard Mitigation Plan's recommended actions through existing plans and policies. Plans and policies already in existence have support from local residents, businesses, and policy makers. Many land-use, comprehensive, and strategic plans get updated regularly, and can adapt easily to changing conditions and needs. Implementing the Natural Hazard Mitigation Plan's action items through such plans and policies increases their likelihood of being supported and implemented.

Plans that can incorporate mitigation action items include the City of Salem Comprehensive Plan, master plans, the City of Salem Subdivision Ordinance, the Community Wildfire Protection Plan (CWPP), and the City of Salem Emergency Operations Plan. Because these plans are used on a regular basis, incorporating mitigation actions into these plans will likewise facilitate their implementation. In addition, when local funds are used to implement mitigation projects related to existing programs, like the Capital Improvement Program (CIP), the regular CIP funding process will be utilized instead of the process highlighted in Section 1.5.3.

5.2 Action Item Worksheets

Each action item has a corresponding action item worksheet describing the activity, identifying the rationale for the project, identifying potential ideas for implementation, and assigning coordinating and partner organizations. The worksheet also has a status section that will allow the Salem Natural Hazards Mitigation Committee to track progress of an action item and identify whether or not it has been completed. The action item worksheets can assist the community in pre-packaging potential projects for grant funding. The worksheet components are described below, followed by a matrix of the action items and the individually listed action items. While the action items are numbered, the numbering is for identification purposes only and does not prioritize the action items. Preceding the worksheets on the following page is an Action Item Matrix that summarizes all the plan's actions.

5.2.1 Rationale or Key Issues Addressed:

Action items should be fact-based and tied directly to issues or needs identified throughout the planning process. Action items can be developed at any time during the planning process and can come from a number of sources, including participants in the planning process, noted deficiencies in local capability, or issues identified through the risk assessment.

5.2.2 Ideas for Implementation:

The ideas for implementation offer a transition from theory to practice and serve as a starting point for this plan. This component of the action item is dynamic, since some ideas may prove infeasible, and new ideas may be added during the plan maintenance process. Ideas for implementation include such things as collaboration with relevant organizations, grant programs, tax incentives, human resources, education and outreach, research, and physical manipulation of buildings and infrastructure.

5.2.3 Lead Agency:

The lead agency is the public agency with the regulatory responsibility to address natural hazards, or that is willing and able to organize resources, find appropriate funding, or oversee activity implementation, monitoring and evaluation.

5.2.4 Internal and External Partners:

The internal and external partner organizations listed in the Action Item Worksheets are potential partners recommended by the Committee but not necessarily contacted during the development of the plan. The lead agency should contact the identified partner organizations to see if they are capable of and interested in participation. This initial contact is also to gain a commitment of time and/or resources toward completion of the action items.

Internal partner organizations are departments/divisions/sectors within the City that may be able to assist in the implementation of action items by providing relevant resources to the coordinating organization.

External partner organizations can assist the coordinating organization in implementing the action items in various functions and may include local, regional, state, or federal agencies, as well as local and regional public and private sector organizations.

5.2.5 Plan Goals Addressed:

The plan goals addressed by each action item are identified as a means for monitoring and evaluating how well the mitigation plan is achieving its goals following implementation.

5.2.6 Timeline:

Action items include both short and long-term activities. Each action item includes an estimate of the timeline for implementation. Short-term action items are activities that may be implemented with existing resources and authorities in one to two years. Long-term action items may require new or additional resources and/or authorities, and may take from one to five years to implement.

5.2.7 Action Item Status:

As the Salem Natural Hazards Mitigation Committee implements action items, they can note progress made on each action in the Action Item Status section. Information that should be noted includes when an action is implemented, who implemented it, and how. Noting the status of each action will also facilitate determining the status of the action for the next update cycle. Please note that Action Item Status is only completed for the 2003 actions located in Appendix D.

				Alignment with Plan Goals					
Action Item	Proposed Action Title	Lead Agency	Partner Organizations	Timeline	Protect lives.	Protect existing and future development.	Increase cooperation and coordination among stakeholders.	Reduce Economic Loss.	Protect the natural environment.
PLAN IMPLEMENTATION #1	Request FEMA approval of the Natural Hazards Mitigation Plan Update.	City of Salem Emergency Management	Salem Natural Hazard Mitigation Committee, Salem City Council, Oregon Office of Emergency Management, FEMA	Short-Term			х		
PLAN IMPLEMENTATION #2	Salem Emergency Management will take on the role of convener to coordinate hazard mitigation meetings and implementation of mitigation action items.	City of Salem Emergency Management	FEMA Region X, Oregon Office of Emergency Management (OEM)	Ongoing			х		
PLAN IMPLEMENTATION #3	The Salem Natural Hazard Mitigation Committee will be the coordinating body responsible for implementing the Salem Natural Hazards Mitigation Plan.	City of Salem Emergency Management	Salem Natural Hazard Mitigation Committee Oregon Office of Emergency Management	Ongoing			х		
MULTI-HAZARD # 1	Coordinate with the Capital Projects Advisory Board to integrate natural hazard mitigation into State and City respective capital improvements.	Salem Community Development Department	Salem Natural Hazard Mitigation Committee, FEMA	Ongoing		Х	Х		
MULTI-HAZARD # 2	Develop an inventory of the number and type of structures within the community that are at risk for each hazard type.	Salem Community Development Department	Salem Natural Hazard Mitigation Committee, FEMA, OEM, Capital Projects Advisory Board	Short-Term		Х			
MULTI-HAZARD # 3	Develop and annually update contacts and telephone numbers of personnel that would be involved in emergency preparedness/response to coordinate emergency response actions.	Salem Emergency Management	Salem Natural Hazard Mitigation Committee, FEMA, Oregon State Police, Oregon Office of Emergency Management, Marion and Polk Counties Emergency Management	Short-Term	X		Х		
MULTI-HAZARD # 4	Develop and annually update an electronic map/database containing critical facilities at risk so that the data is readily accessible to all City personnel responding to a disaster.	Salem Fire Department	Salem Community Development Department, Public Works, Administrative Services, FEMA, Oregon State Police, Oregon Office of Emergency Management	Short-Term	X		Х		
MULTI-HAZARD # 5	Develop public outreach materials for all natural hazard risks addressed in the Salem Natural Hazards Mitigation Plan. Materials should include mitigation actions residents and businesses can implement to reduce their risk to natural hazards, and where they can obtain more detailed natural hazard information.	Salem Emergency Management	Salem Community Development Department, , Community Services, Public Works, FEMA, Oregon State Police, Oregon Office of Emergency Management	Ongoing		х	х	х	
MULTI-HAZARD # 6	Develop a database of names, phone numbers, etc. of all stakeholders (cities, counties, citizens, businesses, and others) involved in planning for and responding to natural disasters.	Salem Emergency Management	Community Services, Public Works, Community Development, Administrative Services, FEMA, Chamber of Commerce, CERT	Short-Term	Х		х		
MULTI-HAZARD # 7	Include a post-disaster recovery and mitigation annex/appendix in the Salem Emergency Operations Plan that encourages property owners to incorporate retrofitting and mitigation measures in recovery efforts.	Salem Emergency Management	Salem Natural Hazard Mitigation Committee, FEMA, Oregon State Police, Oregon Office of Emergency Management	Short-Term		х	х		

I age to

					Ali	gnme G	nt wit Joals	h Pla	an
Action Item	Proposed Action Title	Lead Agency	Partner Organizations	Timeline	Protect lives.	Protect existing and future development.	Increase cooperation and coordination among stakeholders.	Reduce Economic Loss.	Protect the natural environment.
MULTI-HAZARD # 8	Maintain a liaison with the Office of Emergency Management to remain informed of state laws and regulations, and potential grant opportunities, governing all aspects of post-disaster recovery and mitigation.	Salem Emergency Management	Salem Police Department, FEMA, Oregon Office of Emergency Management	Ongoing			х		
MULTI-HAZARD # 9	Consider measures when updating the Salem code for mitigating the location of future development in identified/mapped high hazard areas.	Salem Community Development Department	Salem Natural Hazard Mitigation Committee, DLCD, FEMA	Ongoing		Х			х
MULTI-HAZARD # 10	Develop and improve natural hazard map overlays, including a composite hazard map, to guide the planning and permit review process.	Salem Public Works Department	Salem Natural Hazard Mitigation Committee, Salem Community Development Department, DLCD, DOGAMI, FEMA	Short-Term		х			
MULTI-HAZARD # 11	Strengthen or replace unsafe public structures (especially facilities critical to disaster and post-disaster planning/response).	Salem Public Works Department	Salem Public Works, Fire Department, Police Department, Community Development, Urban Development, Administrative Services, FEMA, ODOT	Long Term	х	Х			
MULTI-HAZARD # 12	Prescribe standards for the design and construction of new public facilities in high hazard areas.	Salem Emergency Management	Salem Public Works, Community Development, Urban Development, ODOT, DOGAMI, DLCD	Long Term		Х			х
MULTI-HAZARD # 13	Continue developing alert and warning systems to notify residents of incidents involving natural hazards and hazardous materials.	Salem Fire Department	Public Works, Police Department, GIS and Mapping Departments, ODOT, FEMA, OSHA	Long Term	х		Х		
LANDSLIDE # 1	Map areas of landslide risk adjacent to the North Santiam River (upstream of the Geren Island water intake structures) and areas impacted by a catastrophic failure of the Detroit or Big Cliff Dams.	Salem Public Works Department	Salem Community Development, US Army Corps, DLCD, FEMA, BLM, USFS	Long Term	x	X			
LANDSLIDE # 2	Develop a general landslide awareness program based on monitoring soil saturation and rain forecasts (e.g. press release associated with extended period of rain).	Salem Public Works	Salem Community Development Department, Community Services, Marion County, Polk County, FEMA, DOGAMI, ODOT, Neighborhood Associations	Long Term	X	Х			х
LANDSLIDE # 3	Educate and inform residents in designated landslide areas about mitigation activities that will reduce the impact of landslides on their properties.	City of Salem Community Development Department	Salem Emergency Management, FEMA, DLCD, DOGAMI, City of Salem Neighborhood & Community Services Division	Ongoing			х	х	
LANDSLIDE # 4	Improve the existing Erosion Prevention and Sediment Control (EPSC) program and regulations established in SRC 65 and 69 to help control erosion.	Salem Public Works Department	Salem Natural Hazards Mitigation Committee, Community Development FEMA, DLCD, ODEO, ODOT	Ongoing					

					Alignment with Plan Goals				
Action Item	Proposed Action Title	Lead Agency	Partner Organizations	Timeline	Protect lives.	Protect existing and future development.	Increase cooperation and coordination among stakeholders.	Reduce Economic Loss.	Protect the natural environment.
LANDSLIDE # 5	Update landslide overlay maps using Light Detection and Ranging (LIDAR) data.	Salem Public Works Department	Salem Natural Hazards Mitigation Committee, City GIS technicians, FEMA, NOAA, DLCD, DOGAMI, Keizer, Turner, Marion County, Polk County	Long Term		Х	х		
EARTHQUAKE # 1	Develop an inventory of un-reinforced masonry structures and develop appropriate mitigation action items to reduce the impacts of seismic events.	Salem Community Development Department	Salem Urban Development, Public Works, Fire, FEMA, DOGAMI	Long Term	х	Х			
EARTHQUAKE # 2	Identify and inventory critical facilities that require seismic retrofit.	City of Salem Emergency Management	Salem Natural Hazards Mitigation Committee, Salem Community Development Department, Salem Public Works FEMA, OEM, DOGAMI, Local School Districts	Long Term		Х	Х		
EARTHQUAKE # 3	Partner with the school districts to help identify and prioritize seismic retrofits to school district facilities.	City of Salem Emergency Management	Salem Natural Hazards Mitigation Committee, Salem Community Development Department, FEMA, OEM, DOGAMI, Salem-Keizer School District, private schools, Chemeketa C.C., Willamette, Corban College	Long Term	Х	Х			
FLOOD # 1	Develop a flood warning system on local creeks that is consistent with the 2000 Stormwater Master Plan and the Corps of Engineers (COE) Section 205 Flood Damage Reduction Study for Mill Creek.	City of Salem Public Works Department	Fire Department, Salem Community Development FEMA, DLCD, Army Corps of Engineers, Marion and Polk Counties, MCSWCD, Santiam Water Control District	Long Term	Х				
FLOOD # 2	Encourage landowners in the UGB to provide set-asides for open space for less intensive development.	Salem Community Development Department	Salem Public Works Department DLCD, FEMA, Marion and Polk Counties	Ongoing			х		Х
FLOOD # 3	Implement recommendations presented in the 2000 Stormwater Master Plan (SWMP) to reduce flood hazards in the city.	Salem Public Works Department	Salem Community Development Dept. Marion County, Polk County, FEMA	Short-Term		х			х
FLOOD # 4	Revise SRC 140 Floodplain Overlay Zones to increase flood freeboard at 2 feet above the 100-year elevation in areas where past floods (including the 1996 floods) have been higher than the FIRM-predicted 100-year elevation.	Salem Public Works Department	Salem Community Development, Legal, Public Works FEMA, Neighborhood Associations, Marion-Polk Building Industry Association (MPBIA)	Long Term		х		х	
FLOOD # 5	Improve the City of Salem's National Flood Insurance Program (NFIP) Community Rating System (CRS) to reduce NFIP premiums.	Salem Public Works Department	Salem Community Development DLCD, National Flood Insurance Program, FEMA, Marion and Polk Counties	Short-Term			х	X	
WINDSTORM/WINTER STORM #1	Partner with public and private utilities to educate the public about hazardous trees and the damage they can cause in the event of a wind or winter storm.	Salem Public Works Department	Salem Community Services Parks Operations , Salem Fire Department ODOT Portland General Electric, Salem Electric	Long Term		Х		х	

					Alignment with Plan Goals				
Action Item	Proposed Action Title	Lead Agency	Partner Organizations	Timeline	Protect lives.	Protect existing and future development.	Increase cooperation and coordination among stakeholders.	Reduce Economic Loss.	Protect the natural environment.
WILDFIRE #1	Conduct wildfire prevention outreach, as outlined in the Marion County Community Wildfire Protection Plan (CWPP), to residents near the wildland-urban interface.	Salem Fire Department	Salem Public Works and Community Development Departments, Police Department, Community Services, Oregon Department of Forestry, Marion County Fire District #1, South Suburban Fire District, Neighborhood Associations	Long Term	X		Х		
HAZARDOUS MATERIALS # 1	Map facilities that handle or contain hazardous materials, rank them based on their level of risk, and refine response strategies for each situation in the event of an accident.	Salem Fire Department	Salem Emergency Management, Public Works OSHA, Salem Chamber of Commerce, Neighborhood Associations, ODOT, OEM, State Police, State Fire Marshall	Long Term	X		х		x

PLAN IMPLEMENTATION ACTION #1

Proposed Action Item:		Alignment with Plan Goals:				
Request FEMA approval of the Salem Natural Mitigation Plan Update.	Hazards	Increase cooperation and coordination among stakeholders.				
Rationale for Proposed Action Item:						
 FEMA approval of the Salem Natural Hazards Mitigation Plan update is required to maintain eligibility for the Pre-Disaster Mitigation Grant Program, the Hazard Mitigation Grant Program funds, and Flood Mitigation Assistance program funds The Disaster Mitigation Act of 2000 requires plans to include an action item that ensures the coordination of all relevant jurisdictions to request FEMA approval of the plan update [§201.6(c)(3)(iv)]. The Mitigation Plan should be coordinated with, and ideally developed in cooperation with, all of the local jurisdictions within the geographical area. [FMA FEMA 299 Guidance] 						
Ideas for Implementation:						
Plan Approval Process:						
1. After updates have been completed and incorporated into the plan, the plan will be approved by the Salem Natural Hazards Mitigation Committee.						
2. The plan will be submitted to the State Management for their approval.	Hazard Mitigation C	Officer at Oregon Emergency				
 Oregon Emergency Management will s Agency (FEMARegion X) for review 	submit the plan to the and approval.	e Federal Emergency Management				
4. Upon acceptance by FEMA, the City o	f Salem's City Coun	cil will adopt the plan via resolution.				
Lead Agency: City of Salem Emergency Ma	anagement					
Internal Partners:	External Partners	:				
Salem Natural Hazards Mitigation	Salem City Council Management FFM	l, Oregon Office of Emergency				
Timeline:	If available, estimation	ated cost:				
Short Term (0-2 years) Long Term (2-4 or more years)	II uvullubic, estilli					
Short Term						
Form Submitted by: Salem Natural Haza	ards Mitigation Com	mittee				
Action Item Status:						

PLAN IMPLEMENTATION ACTION #2

Proposed Action Item:		Alignment with Plan Goals:			
Salem Emergency Management will take on the to coordinate hazard mitigation meetings and in mitigation action items.	e role of convener mplementation of	Increase cooperation and coordination among stakeholders.			
Rationale for Proposed Action Item:					
The convener identified in the 2003 Hazard Department, which focused on the creation regarding the Salem Natural Hazards Mitig Emergency Management would be a more implement and maintain the plan together y	d Mitigation Plan is t of the Hazard Mitig gation Plan and discu appropriate convene with the coordinating	the Community Development gation Plan. Review of past meetings assions with staff indicate that Salem r. The convener's role would be to g body identified in the mitigation plan.			
Ideas for Implementation:					
The roles and responsibilities of the convener include the following:					
1. Coordinate Salem Natural Hazards Mitigation Committee meeting dates, times, locations, agendas, and member notification;					
2. Document outcomes of Committee meetings;					
3. Serve as a communication conduit between	3. Serve as a communication conduit between the Committee and key plan stakeholders;				
4. Identify emergency management-related fu	unding sources for na	tural hazard mitigation projects;			
5. Incorporate, maintain, and update the City	's natural hazard risk	GIS data elements; and			
6. Utilize the Risk Assessment as a tool for pa	rioritizing proposed i	natural hazard risk reduction projects.			
Contact: Roger Stevenson, Emergency Manage	er				
City of Salem Emergency Management					
595 Cottage St NE Salem OR 97301					
Lead Agency: City of Salem Emergency M	anagement				
Internal Destroyes	Extornal Dartnar				
Salem Natural Hazards Mitigation	FEMA Region X.	Dregon Office of Emergency			
Committee	Management (OEN	(1)			
Timeline:	If available, estim	ated cost:			
<u>Short Term (0-2 years)</u> <u>Long Term (2-4 or more years)</u>	-				
Form Submitted by: Salem Natural Haz	Form Submitted by: Salem Natural Hazards Mitigation Committee				
Action Item Status:					

PLAN IMPLEMENTATION ACTION #3

Proposed Action Item:			Alignment with Plan Goals:			
The Salem Natural Hazard coordinating body respons Natural Hazards Mitigatio	s Mitigation Commi ible for implementir n Plan.	ittee will be the ag the Salem	Increase cooperation and coordination among stakeholders.			
Rationale for Proposed A	ction Item:					
 The Salem Natural Ha members, to be the ma The Disaster Mitigatic describing the method within a five-year cycl coordinating body resp it remains relevant to \$ 	zards Mitigation Co in body to implement on Act of 2000 require and schedule of mor- e [201.6(c)(4)(i)]. A ponsible for both the Salem's needs.	mmittee identified its nt the Salem Natural res Mitigation Plans nitoring, evaluating, A key component to e maintenance and im	self, with the inclusion of other Hazards Mitigation Plan. to include a maintenance section and updating the Mitigation Plan effective maintenance is to have a plementation of the plan to ensure that			
Ideas for Implementation:						
 Convene the Salem Natural Hazards Mitigation Committee on a semi-annual basis to discuss Plan action items and methods for their implementation. After natural hazard events occur, convene the coordinating body to discuss action items for implementation or strategies for amending the plan to incorporate new action items. 						
Lead Agency:	City of Saler	n Emergency Manag	ement			
Internal Partners		Extornal Dartner	•			
Salem Natural Hazards Mi Committee	tigation	Oregon Office of E	Emergency Management			
Timeline:		If available, estim	ated cost:			
<u>Short Term (0-2 years)</u> <u>Long</u>	<u>[erm (</u> 2-4 or more years)	-				
Ongoi						
Form Submitted by:	Salem Natural Haz	ards Mitigation Com	mittee			
Action Item Status:	Action Item Status:					

Proposed Action It	em: (from action # 3)		Alignment with Plan Goals:
Coordinate with the natural hazard mitig improvements.	Capital Projects Advisory ation into State and City re	Increase cooperation and coordination among stakeholders. Protect existing and future development.	
Rationale for Proposed Action Item:			
 The Capital Projects Advisory Board is identified as the main body to implement the State of Oregon capital improvement projects within the greater Salem area. A similar responsibility rests with the Salem Public Works Department in the development and implementation of the City's Capital Improvements Program (CIP). It is important that natural hazard mitigation be integrated into both the State's and Salem's Capital Improvement Program so that critical public facilities, including government buildings, are constructed to function during and after natural disasters. Local units of government want to ensure continuous service by strengthening essential facilities. Ensuring continuous service will assist residents in recovering from a natural disaster as well as make the process easier. The Disaster Mitigation Act of 2000 requires communities to maintain the Hazard Mitigation Plan by having local governments incorporate the requirements of the mitigation plan into other planning mechanisms [201.6(c)(4)(ii)]. Coordinating mitigation activities with other planning activities will help local governments incorporate mitigation into other plans and policies currently being developed Coordination will also reduce duplication of planning efforts, strengthening the overall mitigation planning the overall mitigation 			oody to implement the State of Oregon similar responsibility rests with the mentation of the City's Capital both the State's and Salem's Capital g government buildings, are l units of government want to ensure ng continuous service will assist he process easier. maintain the Hazard Mitigation Plan by mitigation plan into other planning es with other planning activities will and policies currently being developed. trengthening the overall mitigation
Ideas for Implement	Ideas for Implementation:		
 Determine what roles the Capital Projects Board plays in mitigating natural hazards, especially for State of Oregon properties or others in Salem for which it has jurisdiction. Review action items and discuss which ones can be integrated into Capital Improvement Program fo the City of Salem. Inventory critical facilities that may be potentially vulnerable to a natural disaster and present these t the Capital Projects Advisory Board for their review. Include members of the Capital Projects Advisory Board in the Natural Hazard Mitigation Committee meetings as needed. Realign or replace roads and utilities when feasible in the course of regularly scheduled replacement reduce the impact of natural hazard events on new development. Explore the possibility of under grounding utilities that are vulnerable to windstorms and winter 			ating natural hazards, especially for urisdiction. nto Capital Improvement Program for o a natural disaster and present these to Natural Hazard Mitigation Committee e of regularly scheduled replacement to t. nerable to windstorms and winter
Lead Agency: Sa	lem Community Developm	ent Department	
Internal Partners:	enting Committe	External Partners	
Timeline	gation Committee	FEMA, OEM, Cap	nai Projects Advisory Board
Short Tarm (0.2 years) Long Tarm (2.4 or more years)		n avanable, estim	
<u>Short rerni (0-2 years)</u>	<u>Ongoing</u>		
Form Submitted by	y: Salem Natural Haz	ards Mitigation Com	mittee
Action Item Status	•		

Proposed Action Item: (from Action # 5, # 16)		Alignment with Plan Goals:		
Develop an inventory of the number and type the community that are at risk for each hazard	of structures within type.	Protect existing and future development.		
Rationale for Proposed Action Item:				
 Many older commercial buildings in the City of Salem are vulnerable to damage in the event of a natural disaster. This could have significant impacts on the City of Salem's economy. Identifying and retrofitting buildings that are susceptible to a natural disaster will reduce the vulnerability of the buildings in the event of a natural disaster and improve the resiliency of Salem's local economy. OEM's checklist for local mitigation plans includes the need to estimate the type and number of structures within the community at risk for each hazard type, including residences, businesses, critical facilities (hospitals, fire stations, and storage sites for hazardous materials), and infrastructure (e.g., roads and utilities). There also needs to be a map of repetitive flood loss properties (extent of flooding, no evaluation of cost of property damage) and discussion of potential mitigation activities for these properties. The Disaster Mitigation Act of 2000 requires communities to identify actions and projects that reduce the effects of hazards on the community, particularly to buildings and infrastructure [201.6(c)(3)(ii)]. Inventorying important historic and cultural resources and identifying their vulnerability to natural hazards will help to develop mitigation actions that reduce the City of Salem's overall vulnerability to natural hazards. 				
Ideas for Implementation:				
 Determine vulnerabilities of community structures to natural hazards Identify appropriate mitigation measures to help preserve structures within the community that are at risk for each hazard type. Create an electronic data base which illustrates an inventory of the number and type of structures within the community that are at risk for each hazard type. Identify significant cultural and historic resources, whether on the national register or not, that are worthy of additional protection 				
Lead Agency: Salem Community Developm	nent Department			
Internal Partners:	External Partners	:		
Salem Natural Hazards Mitigation Committee	FEMA			
Timeline:	If available, estimation	ated cost:		
Short Term (0-2 years) Long Term (2-4 or more years)	-			
Form Submitted by: Salem Natural Haz	zards Mitigation Com	mittee		
Action Item Status:				

Proposed Action Item: (from Action Item # 6)		Alignment with Plan Goals:	
Develop and annually update contacts and telep personnel that would be involved in emergency preparedness/response to coordinate emergency	phone numbers of y y response actions.	Protect lives. Increase cooperation and coordination among stakeholders.	
Rationale for Proposed Action Item:			
 Emergency response personnel are vital to an effective and efficient response to a natural disaster. Developing and continuously updating a contact list of City of Salem personnel to be involved in emergency preparedness/response ensures a quick and adequate response in the event of a natural disaster. Maintaining an updated list of emergency preparedness/response personnel increases the ability of the City of Salem to respond to a natural disaster, therefore making the City more resilient. The Disaster Mitigation Act of 2000 requires communities to identify comprehensive actions that will reduce the impact of a natural hazard [201.6(c)(3)(ii)]. Developing a contact list of personnel to be involved in the coordination of emergency response actions within will help reduce response time in the event of a natural disaster, which can help reduce the overall impact of an incident on the community. 			
Ideas for Implementation:			
 Solicit City personnel who may be interested in participating in the emergency preparedness/response process. Offer trainings to City personnel that focus on how to prepare and respond to a natural disaster. Offer incentives to those interested in participating in the emergency preparedness/response process. Once formed, regularly convene with City personnel dedicated to emergency preparedness/response to discuss operational processes in the event of a natural disaster. 			
Lead Agency: Salem Emergency Manageme	ent		
Internal Partners:	External Partners	:	
Salem Natural Hazards Mitigation Committee	FEMA, Oregon State Police, Oregon Office of Emergency Management, Marion and Polk Counties Emergency Management		
Timeline:	If available, estimation	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years) Short Term			
Form Submitted by: Salem Natural Haz	ards Mitigation Com	mittee	
Action Item Status:			

Proposed Action Ite	e m: (fr	rom action # 6)		Alignment with Plan Goals:
Develop and annually update an electronic map/databas containing critical facilities at risk so that the data is rea accessible to all City personnel responding to a disaster			p/database lata is readily a disaster.	Protect lives. Increase cooperation and coordination among stakeholders.
Rationale for Proposed Action Item:				
 Developing an electronic map/database noting the location of critical facilities vulnerable to a natural disaster increases the City of Salem's ability quickly and adequately respond to damaged critical facilities in the even of a natural disaster. In addition, restoring utility services is an essential prerequisite for beginning other recovery efforts (i.e. economic recovery, hospital services, and public facilities) that put a community back online. Completing an inventory of critical facilities vulnerable to damage resulting from a natural disaster will assist the City of Salem in prioritizing buildings for seismic retrofit. The Disaster Mitigation Act of 2000 requires communities to identify comprehensive actions that reduce the effects of hazards on the community, particularly to new and existing buildings and infrastructure [201.6(c)(3)(ii)]. 				
Ideas for Implemen	tatior	1:		
 Combine the database of critical facilities with known mapped hazard areas to understand risk and develop appropriate response measures. Evaluate a building's survivability following a single and compound disaster. Develop mitigation funding strategies. Utilize information attained from the Statewide Seismic Needs Assessment resulting from Oregon 2005 Senate Bill 2 Relating to Public Safety, Earthquakes, and Seismic Rehabilitation of Public Buildings to develop or update the City of Salem's electronic map/data base containing critical facilities vulnerable to natural disasters. Utilize grant funds appropriated from Oregon 2005 Senate Bill 2-5 for rehabilitation of critical facilities. Note: grant funding for seismic rehabilitation is directly related to seismic needs assessmen 			hazard areas to understand risk and bound disaster. Assessment resulting from Oregon Seismic Rehabilitation of Public hap/data base containing critical 2-5 for rehabilitation of critical tly related to seismic needs assessment.	
Internal Partners:			External Partners	
Salem Community D Department, Public V Services	Salem Community Development Department, Public Works, Administrative Services		FEMA, Oregon Sta Management	ate Police, Oregon Office of Emergency
Timeline:			If available, estim	ated cost:
Short Term (0-2 years) Long Term (2-4 or more years) short term				
Form Submitted by	:	Salem Natural Haz	ards Mitigation Com	mittee
Action Item Status:				

Proposed Action Item: (fro	om action # 7)		Alignment with Plan Goals:
Develop public outreach ma	aterials for all natur	al hazard risks	Increase cooperation and
addressed in the Salem Natu	ural Hazards Mitiga	tion Plan. Materials	coordination among stakeholders.
should include mitigation ad	ctions residents and	businesses can	Reduce economic loss.
implement to reduce their ri	ls, and where they	Protect existing and future	
can obtain more detailed na	tural hazard inform	ation.	development.
Rationale for Proposed Ac	ction Item:		
 Conducting public outreach campaigns raises awareness about natural hazards and helps illustrate what residents and businesses can do to reduce the impact of a natural disaster on their properties, thereby significantly reducing the impact of a natural disaster on the City of Salem. Several natural hazards, such as severe weather, earthquakes, and floods, have the potential for disrupting transportation services and isolating rural residents from basic services and needs. The City of Salem has a large number of residents, and they need to be educated about the dangers that natural hazards pose and what actions they can take to mitigate the impact hazards on the community. The Disaster Mitigation Act of 2000 requires communities to identify comprehensive actions and 			
projects that reduce the	effects of a hazard	on the community [20]	
Ideas for Implementation:	:		
 Conduct public outreach campaigns, such as articles in the newspaper or through brochures instructing residents and businesses about the risks natural hazards pose and mitigation actions they can implement. Coordinate with other groups conducting other emergency management activities to assist in conducting public outreach campaigns, developing emergency kits, and educating residents and businesses about other mitigation activities Develop handouts that inform residents and businesses about natural hazard risk, appropriate mitigation actions that can be implemented, and where citizens can obtain further information. Create an online informational website where residents and businesses can be educated about appropriate mitigation actions residents and businesses can implement to reduce the impact of natural hazards Work with local real estate trade associations to prepare informational handouts advising property owners of natural hazard risks in their area and measures they can implement to reduce their risk of exposure 			
Lead Agency: Salem Em	ergency Manageme	ent	
Internal Partners:		External Partners:	
Salem Community Develop Department, Community Se Works	oment ervices, Public	FEMA, Oregon State Emergency Managem	Police, Oregon Office of ent
Timeline:		If available, estimate	ed cost:
Short Term (0-2 years) Long Term (2-4 or more years)			
Form Submitted by	salem Natural Haz	ards Mitigation Comm	ittee
Action Item Status:			

Proposed Action Item: (from action # 7)		Alignment with Plan Goals:	
Develop a database of names, phone numbers,	etc. of all	Protect lives.	
stakeholders (cities, counties, citizens, business	ses, and others)	Increase cooperation and	
involved in planning for and responding to nat	ural disasters.	coordination among stakeholders.	
Rationale for Proposed Action Item:			
 Stakeholders within community organizations and programs that provide social and community-based services are important resilience factors because of their existing connections to the public. Communication within the community is a key part of many mitigation strategies. On a daily basis, social service providers work and communicate directly with the public on a number of issues, one of which could be natural hazard preparedness and mitigation. In addition, organizations found within the inventory may also provide grant search resources and temporary working office space for NGOs. The Disaster Mitigation Act of 2000 requires communities to develop actions that reduce the impact of a natural hazard [201.6(c)(3)(ii)]. Developing a database of stakeholders involved in responding to natural disasters will diminish the effects of a natural disaster by providing the City of Salem with a framework for responding to community needs in a potentially chaotic situation. 			
Ideas for Implementation:			
 Develop an electronic database that includes: names, phone numbers, address, and other contact information for of all stakeholders involved in planning for and responding to natural disasters. Distribute the database to all of the stakeholders and relevant public/private agencies Update the database periodically to maintain an effective database of stakeholders 			
Lead Agency: Salem Emergency Manageme	ent		
Internal Partners:	External Partners	:	
Community Services, Public Works, Community Development, Administrative Services			
Timeline:	If available, estim	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years)	Minimal.		
Short term (Within 1			
year)			
Form Submitted by: Salem Natural Haz	ards Mitigation Com	mittee	
Action Item Status:			

Proposed Action Item: (from Action # 9)		Alignment with Plan Goals:	
Include a post-disaster recovery and mitigation the Salem Emergency Operations Plan that enco owners to incorporate retrofitting and mitigation recovery efforts. Rationale for Proposed Action Item:	Protect existing and future development. Increase cooperation and coordination among stakeholders.		
 Disaster response is an important component to natural hazards planning that can save lives and property during a natural disaster. Coordinating disaster response efforts with the mitigation plan will ensure that the plan remains relevant to the larger community. Resources that may not be available on a routine basis for certain improvements may become available through various disaster relief sources, particularly where careful planning has allowed the community to identify certain needs in advance, saving critical time in the aftermath of a disaster. The Disaster Mitigation Act of 2000 requires communities to develop actions that reduce the impact of a natural hazard [201.6(c)(3)(ii)]. Incorporating information about mitigation and retrofitting will increase the City of Salem's ability to recover from a natural disaster. 			
Ideas for Implementation:			
 Seek guidance from FEMA and the Oregon Office of Emergency Management on how to incorporate recovery and mitigation measures into the Salem Emergency Operations Plan. Periodically update the recovery and mitigation measures that have been incorporated into the Salem Emergency Operations Plan 			
Lead Agency: Salem Emergency Management			
Internal Partners:	External Partners	:	
Salem Natural Hazards Mitigation Committee	FEMA, Oregon Sta Emergency Manage	te Police, Oregon Office of ement	
Timeline:	If available, estim	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years)	Minimal.		
Short term			
Form Submitted by: Salem Natural Hazards Mitigation Committee			
Action Item Status:			

Proposed Action Item: (from	m Action # 9)		Alignment with Plan Goals:
Maintain a liaison with the Office of Emergency Management to remain informed of state laws and regulations, and potential grant opportunities, governing all aspects of post-disaster recovery and mitigation.			Increase cooperation and coordination among stakeholders.
Rationale for Proposed Act	tion Item:		
 Disaster response is an important component to natural hazards planning that can save lives and property during a natural disaster. Understanding of the laws and regulations and coordinating disaster response/mitigation efforts with other public agencies will increase the resiliency of the City of Salem to Natural Hazards. The Disaster Mitigation Act of 2000 requires communities to develop actions that reduce the impact o a natural hazard [201.6(c)(3)(ii)]. Communicating with the Office of Emergency Management will increase the City of Salem's resiliency and ability to recover from a natural disaster. 			
Ideas for Implementation:			
 Identify a contact at both the Office of Emergency Management Schedule regular meetings to gather up-to-date understanding of the rules and regulations, and potential grant opportunities, governing aspects of post-disaster recovery and mitigation. 			t f the rules and regulations, and recovery and mitigation.
Lead Agency: Salem Emergency Management		ent	
Internal Partners:		External Partners	:
Salem Police Department		FEMA, Oregon Of	fice of Emergency Management
Timeline:		If available, estimation	ated cost:
Short Term (0-2 years) Long Ter Ongoing Ongoing	m_(2-4 or more years)	Minimal.	
Form Submitted by:	Form Submitted by: Salem Natural Hazar		mittee
Action Item Status:			

Proposed Action Ite	em: (fi	rom Action # 10)		Alignment with Plan Goals:
Consider measures when updating the Salem code for mitigating the location of future development in identified/mapped high hazard areas.			ode for mitigating d/mapped high	Protect existing and future development. Protect the natural environment.
Rationale for Proposed Action Item:				
 Goal 7 of Oregon's Land Use Planning Goals requires that local governments "adopt or amend, as necessary, based on the evaluation of risk, plan policies and implementing measures[that prohibit] the siting of essential facilities, major structures, hazardous facilities and special occupancy structures, as defined in the state building code (ORS 455.447(1) (a)(b)(c) and (e)), in identified hazard areas" The Disaster Mitigation Act of 2000 requires communities to identify actions and projects that reduce the effects of hazards on the community [201.6(c)(3)(ii)]. Adjusting the Salem code to move future development from identified/mapped hazards areas will reduce the vulnerability of new development to natural hazards. 				
Ideas for Implemen	itatior	n:		
 Consider transferring development rights from high hazard areas to safer areas, especially in those areas where the risk to people and property cannot be mitigated. Address high hazard areas and consider measures for mitigating the location of future development these areas during the update of the Salem code. 			as to safer areas, especially in those l. g the location of future development in	
Lead Agency: Salem Community Development Department				
Internal Partners:			External Partners	:
Salem Natural Hazar Committee	rds Mi	tigation	DLCD, FEMA	
Timeline:			If available, estim	ated cost:
Short Term (0-2 years)	Short Term (0-2 years) Long Term (2-4 or more years) Ongoing		_	
Form Submitted by: Salem Natural Haza		ards Mitigation Com	mittee	
Action Item Status:				

Г

Proposed Action	Item: (f	rom Action # 10)		Alignment with Plan Goals:	
Develop and improve natural hazard map overlays, including a composite hazard map, to guide the planning and permit review process.			Protect existing and future development.		
Rationale for Pro	posed A	Action Item:		<u> </u>	
 Overlay districonventional to designations a serve to conformeded in dea Overlay zones standards to a overlay zoning comprehensiv time and experiment provide clear the use of the The Disaster I the effects of I prone areas the development of the server the s	cts are u use distri- ind whose ound zon ling with s are zon ddress sp g helps t e plan m inse of re- informatizoned pa Mitigatic hazards to natura	sed to solve problem icts. Generally they se inclusion would re- ing enforcement effor- nenvironmental cons- es imposed over exis- pecial land use needs o: (1) Eliminate any hap; (2) Ensure consi- e-interpreting the cor- tion, to all current an arcel. on Act of 2000 requin- on the community [2 ases restrictions on de- l hazards.	as in zoning codes that aim to address specific esult in a level of deli- ports. They also allow straints, with floodpla- sting zones and "prov- s"1 such as floodplain confusion created by stent administration inprehensive plan may d prospective landow res communities to ic 201.6(c)(3)(ii)]. Development will redu	at are not adequately addressed in fic needs that cut across other district ineation in normal districts that would a degree of flexibility that is often ains being a common example. vide an additional layer of development as or other natural hazard areas. Using the broadly defined boundaries on the of all hazards ordinances; (3) Avoid the p for each development request; (4) vners, of the regulations which affect dentify actions and projects that reduce eloping an overlay zone in hazard- ice the vulnerability of new	
 Coordinate wi and display ha Periodically u 	th other azard ove pdate th	public agencies to g erlays in maps. is information to reta	ather the necessary d	ata to create effective overlay districts	
Lead Agency:	Salem Pı	ublic Works Departn	nent		
Internal Partners	s:		External Partners	:	
Salem Natural Hazards Mitigation Committee, Salem Community Development		DLCD, DOGAMI, FEMA			
Timeline:			If available, estim	ated cost:	
<u>Short Term (</u> 0-2 years) <u>Short Term</u>	Long	Term (2-4 or more years)	Medium.		
Form Submitted	by:	Salem Natural Haz	ards Mitigation Com	mittee	
Action Item State	us:	·			

Proposed Action Item: (from Action # 11)		Alignment with Plan Goals:	
Strengthen or replace unsafe public structures (facilities critical to disaster and post-disaster pla	Protect lives. Protect existing and future development.		
Rationale for Proposed Action Item:			
 The Disaster Mitigation Act of 2000 requires communities to assess their vulnerability to natural hazards, particularly by identifying the types and number of buildings, infrastructure, and critical facilities that could be affected. It is important that critical facilities function during and after disasters. Strengthening all essential facilities will improve recovery capacity and reduce risk and loss of life. Retrofitting of vital infrastructure, such as schools and community buildings, provides important improvements that reduce hazard exposure and the cost and time associated with recovery. 			
Ideas for Implementation:			
 Develop formal agreements with internal and external partners who could assist the partners in collaborating and sharing the responsibility of natural hazard mitigation. Such actions to form collaborative partnerships and commitments to mitigation can assist the City in reducing its risk to the natural hazards addressed by the Natural Hazards Mitigation Plan. Conduct structural and non-structural retrofits of critical facilities to reduce the impacts of a natural hazard. Conduct a cost-benefit analysis to assess whether the cost of mitigation improvements to critical facilities balance with the benefits to be gained. Create proposals to reinforce buildings so they can withstand an earthquake and thereby reduce vulnerability risks; ORS 455.447 regulates vulnerable building retrofits. Lead Agency: Salem Public Works Department 			
Internal Partners:	External Partners	:	
Salem Public Works, Fire Department, Police Department, Community Development, Urban Development, Administrative Services	FEMA, ODOT		
Timeline:	If available, estimation	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years) Long Term		mittee	
Form Submitted by: Salem Natural Hazards Mitigation Committee Action Item Status:			

Proposed Action Item: (from Action # 11)		Alignment with Plan Goals:		
Prescribe standards for the design and construct facilities in high hazard areas.	Protect existing and future development. Protect the natural environment.			
Rationale for Proposed Action Item:				
 The Disaster Mitigation Act of 2000 requires communities to assess their vulnerability to natural hazards, particularly by identifying the types and number of buildings, infrastructure, and critical facilities that could be affected. It is important that critical facilities function during and after disasters. Strengthening all essential facilities will improve recovery capacity and reduce risk and loss of life. Retrofitting of vital infrastructure, such as schools and community buildings, provides important improvements that reduce hazard exposure and the cost and time associated with recovery. 				
Ideas for Implementation:				
 Developing formal agreements with internal and external partners could assist the partners in collaborating and sharing the responsibility of natural hazard mitigation. Such actions to form collaborative partnerships and commitments to mitigation can assist the City in reducing its risk to the natural hazards addressed by the Natural Hazards Mitigation Plan. Develop a funding matrix that provides a list of potential funding mechanisms for disaster mitigation activities Coordinate mitigation planning activities with existing land use and master planning activities to incorporate mitigation actions and avoid duplicating efforts. 				
Lead Agency: Salem Emergency Manageme	ent			
Internal Partners:	External Partners	:		
Salem Public Works, Community Development, Urban Development	ODOT, DOGAMI,	DLCD		
Timeline:	If available, estimation	ated cost:		
Short Term (0-2 years) Long Term (2-4 or more years)				
Short Term				
Form Submitted by: Salem Natural Haz	ards Mitigation Com	mittee		
Action Item Status:				

Proposed Action Item	1:		Alignment with Plan Goals:	
Continue developing alert and warning systems to notify			Protect Lives.	
materials			Increase cooperation and coordination among stakeholders.	
Rationale for Propose	ed Action Item:			
 Alert and warning systems can provide a life-saying service to residents in the event of a natural or 				
manmade disaster. Natural and manmade disasters can occur at any time, often unannounced, putting				
people at risk. Dev	people at risk. Developing alert and warning systems can reduce the risk of exposure to natural hazard			
• Alert and warning	system have significant	relevance to hazardo	nd property.	
materials are locate	ed near businesses and r	esidences in Salem as	s well as along major transportation	
routes. Trucking r	outes along the I-5 corri	dor and Highway 22	may also contain hazardous materials	
because there are n	10 restrictions on the typ	e of cargo that travel	s over these routes which run through	
residential and con	nmercial areas in the city	y. In addition, the he	avily-traveled railroad line near the	
Accidents in busir	proximately 12,000 cars	bove routes can have	an adverse impact on the quality of	
life and economy of	of the city and the state;	significant events ha	ve already occurred in Salem in 1976	
and along the I-5 c	orridor. Alert and warn	ing systems can help	to prevent larger accidents from	
occurring and help	to save lives and proper	rty.		
Ideas for Implementa	ition:			
• Continue to enforc	• Continue to enforce the Salem Fire Prevention Code to regulate hazardous materials.			
• Develop strategies	• Develop strategies in local building codes and zoning ordinances to reduce the impact of natural			
hazard and manma	ide hazard events on buil	ldings and infrastruct	ure.	
Continue to develo	p a reverse 9-11 system	to alert nearby resid	ents and businesses of natural hazard	
Develop improved	is materials accident.	Inerable to natural he	azard events and hazardous materials	
• Develop imploved	maps to locate areas vu		zard events and nazardous materials.	
Lead Agency: Saler	n Fire Department			
Internal Partners:		External Partners	:	
Public Works, Police I	Department, GIS and	ODOT, FEMA, OS	HA	
Mapping Departments				
Timeline:		If available, estimation	ated cost:	
Short Term (0-2 years) L	ong Term (2-4 or more years)			
Long Term				
Form Submitted by:	Salem Natural Haz	ards Mitigation Com	mittee	
Action Item Status:				

Proposed Action Item: (from action # 5)		Alignment with Plan Goals:			
Map areas of landslide risk adjacent to the North	h Santiam River	Protect existing and future development			
impacted by a catastrophic failure of the Detroit	or Big Cliff	uevelopmeni. Protect lives			
Dams	of Dig Cilli	Protect natural environment			
Rationale for Proposed Action Item:					
 The current landslide hazard maps are a compilation of existing maps. These maps are a "work in progress" and have been compiled at widely varying scales and sometimes only depict risk for certain types of landslides. These various scales and levels of detail may lead to people to believe that some areas have no slope hazard, when the case is that those areas just have not yet been evaluated. Systematic upgrading of these maps will lead to greater understanding of hazard locales. Focusing on areas that will be developed and will affect people and critical infrastructure will improve land use planning and provide for more efficient and cost effective development. Better data provides for better decisions to minimize loss. Incorporating indirect economic loss better depicts the cost from natural hazard events. Ideas for Implementation: Improve knowledge of debris flow (rapid moving) landslide hazard areas. Improve landslide hazard area maps for a variety of types of landslides that focus on areas that will affect people and critical infrastructure and facilities. Educate identified vulnerable residential and commercial building owners, occupants, and developers of their wuherability to risk. 					
Lead Agency: Salem Public Works Departme	ent				
Internal Partners:	External Partners	:			
Salem Community Development	US Army Corps, D	LCD, FEMA, BLM, USFS			
Timeline:	If available, estimation	ated cost:			
Short Term (0-2 years) Long Term (2-4 or more years)					
Long Term					
Form Submitted by: Salem Natural Hazards Mitigation Committee					
Action Item Status:	Action Item Status:				

Proposed Action Item: (from Action # 8)		Alignment with Plan Goals:		
Develop a general landslide awareness program	n based on	Protect lives.		
monitoring soil saturation and rain forecasts (e.	g. press release	Protect existing and future		
associated with extended period of rain).		development.		
		Protect the natural environment.		
Rationale for Proposed Action Item:				
 The City of Salem is particularly vulnerable to landslide events. In February 1996, residents in the southwest area around Heath Street, Aldous Avenue, and Dorchester Heights suffered from a landslide event. More recently in 2005, a landslide near Owen Street blocked South River Road. The City has a landslide ordinance, passed in 2000 and revised in 2003, that provides development standards in landslide hazard areas and incorporates mitigation activities for new and existing development. Developing a landslide awareness program that monitors soil saturation and advises residents during high rain periods would significantly help in reducing the impact of a landslide in a neighborhood, and assist in providing warning before a landslide occurs. The Salem Natural Hazards Mitigation Committee identified landslides as a high probability and moderate vulnerability. While an awareness program would not reduce the probability of a hazard recurring, it would help reduce the vulnerability to Salem residents to natural hazard events. The Disaster Mitigation Act of 2000 requires communities to identify actions and projects that reduce the effects of hazards on the community, particularly to new and existing buildings and infrastructure [201.6(c)(3)(ii)]. Developing a general landslide awareness program based on monitoring soil saturation and rain forecasts can reduce the impacts of landslides on new and existing development conduction and rain forecasts can reduce the impact of a landslide so new and existing development. 				
Ideas for Implementation:				
 Develop appropriate public outreach materials, such as press releases, handouts, radio advertisements, and mailings to inform residents of the landslide risk present in their area. Expand and continue monitoring of landslide-prone areas to provide an overall assessment of the landslide risk in an area. Coordinate efforts with the Marion County Public Works Department to address landslide risks near, but outside, the City's jurisdiction. 				
Lead Agency: Salem Public Works Departm	ient			
Internal Partners:	External Partners	:		
Salem Community Development Department, Community Services	Marion County, Po Neighborhood Asso	lk County, FEMA, DOGAMI, ODOT, ociations		
Timeline:	If available, estimation	ated cost:		
<u>Short Term (0-2 years)</u> <u>Long Term (</u> 2-4 or more years)				
Short Term				
Form Submitted by: Salem Natural Haza	ards Mitigation Com	mittee		
Action Item Status:				

Proposed Action It	em:			Alignment with Plan Goals:
Educate and inform residents in designated landslide areas about mitigation activities that residents can take to reduce the impact of landslides on their properties.			Increase cooperation and coordination among stakeholders. Reduce economic loss.	
Rationale for Prop	osed Acti	on Item:		
 The City of Salem has identified several areas as susceptible to landslide events. Furthermore, the Salem Natural Hazards Mitigation Committee identified the Heath Street and South River Road areas as particularly susceptible to landslide events. Educating the residents in designated landslide areas about the impacts from a landslide, and providing mitigation actions they can take to reduce the impact, will help to stabilize these areas to prevent future landslides from occurring. The Disaster Mitigation Act of 2000 requires communities to identify mitigation actions property owners might take that address new and existing buildings and infrastructure [201.6(c)(3)(ii)]. Educating residents in landslide prone areas about the effects of landslides and mitigation activities they can implement can significantly reduce the impact of future landslide events. 				
Ideas for Implement	Ideas for Implementation:			
 Identify specific Provide informa educate them ab Conduct public through informa 	 Identify specific landslide issues for areas designated as landslide hazards. Provide informational brochures to current, new, and prospective residents in landslide prone areas to educate them about landslide risks in Salem. Conduct public outreach to residents in other areas that are prone to landslide hazards, possibly through information posted online and communication with relevant community associations. 			
Lead Agency:		City of Salen	n Community Develo	opment Department
Internal Partners:			External Partners	:
Salem Emergency ManagementFEMA, DLCD, DOGAMI, City of Salem Neighb & Community Services Division			GAMI, City of Salem Neighborhood vices Division	
Timeline:			If available, estimation	ated cost:
Short Term (0-2 years)	Long Term Ongoing	(2-4 or more years)		
Form Submitted by	y: Sa	alem Natural Haz	ards Mitigation Com	mittee
Action Item Status	:			

Proposed Action Iter	m:			Alignment with Plan Goals:
Improve the existing Erosion Prevention and Sediment Co (EPSC) program and regulations established in SRC 65 and help control erosion.			ediment Control SRC 65 and 69 to	<i>Reduce Economic Loss.</i> <i>Protect the natural environment.</i>
Rationale for Propos	sed Action	Item:		
 Each year tons of sediment nationally are washed and blown from construction sites into municipal storm drainage systems and local streams, rivers, wetlands, and lakes. It is a major source of pollution to these water bodies. Eroded materials also clog streets, storm drains, culverts, and stream channels and cause private property damage. The degradation of fish and wildlife habitat and water quality, plu the burden placed on ratepayers for cleanup, could be largely avoided through implementation of adequate erosion prevention and sediment control practices. (EPSC Handbook) The Salem Natural Hazards Mitigation Committee identified SRC 75 and 69 as ordinances that need further improvement to help control sediment erosion on construction sites, especially those with excavation activity. The Disaster Mitigation Act of 2000 requires communities to identify mitigation actions that address new and existing buildings and infrastructure [201.6(c)(3)(ii)]. Developing mitigation actions for erosion control can significantly reduce the impact of future landslide events and help maintain environmental quality in streams around the City of Salem. Ideas for Implementation: Maintain plan submittal requirements and recommended measures to prevent erosion and control sediments on construction sites and other properties as set forth in the EPSC Handbook. Restrict construction activity during rainy times of the year to control erosion on construction sites. Identify areas in SRC 75 and 69 that could use further improvement to control sediment erosion. 				 bm construction sites into municipal lakes. It is a major source of pollution drains, culverts, and stream channels I wildlife habitat and water quality, plus voided through implementation of PSC Handbook) RC 75 and 69 as ordinances that need uction sites, especially those with lentify mitigation actions that address Developing mitigation actions for idslide events and help maintain in the EPSC Handbook. and updating of the EPSC handbook. control erosion on construction sites. ment to control sediment erosion.
Lead Agency:		Salem Public	e Works Department	
Internal Partners:			External Partners	:
Salem Natural Hazards MitigationFEMA, DLCD, ODEQ, ODOTCommittee, Community DevelopmentFEMA, DLCD, ODEQ, ODOT)EQ, ODOT	
Timeline:			If available, estimation	ated cost:
<u>Short Term (</u> 0-2 years)	Long Term (2-4	4 or more years)		
Form Submitted by:	Saler	n Natural Haz	ards Mitigation Com	mittee
Action Item Status:				

Proposed Action Item	:			Alignment with Plan Goals:
Update landslide overlay maps using Light Dete Ranging (LIDAR) data.			tection and	Increase cooperation and coordination among stakeholders. Protect existing and future development.
Rationale for Propose	d Action	Item:		
 LIDAR (Light Detection and Ranging) is a new tool that can provide very precise, accurate, and high-resolution images of the surface of the earth, vegetation, and the built environment. The data are collected with aircraft-mounted lasers capable of recording elevation measurements at a rate of 2,000 to 5,000 pulses per second and have a vertical precision of 15 centimeters (6 inches). LIDAR mapping increases the ability to identify areas that are prone to landslides. In 2007 the Oregon Legislature Assembly directed DOGAMI to extend LIDAR collection efforts throughout the state. The ultimate goal is to provide high-quality LIDAR coverage for the entire state. 				
Ideas for Implementat	tion:			
 Contact DOGAMI : Seek funding oppor Once mapping is co Explore potential co mapping of the entitient 	 Contact DOGAMI and provide a map of Salem along with an estimate of available funding. Seek funding opportunities with DOGAMI to conduct LIDAR mapping for the City of Salem. Once mapping is complete assess the need to update landslide ordinances. Explore potential cost-sharing agreements with Keizer, Turner, Marion and Polk Counties for LIDAR mapping of the entire Salem-Keizer urbanized area. 			
Lead Agency:		Salem Public	Works Department	
Internal Partners:			External Partners	:
Salem Natural Hazards	Mitigatic	n	FEMA, NOAA, DLCD, DOGAMI, Keizer, Turner,	
Committee, City GIS te	chnicians	5	Marion County, Po	Jlk County
Timeline:			If available, estim	ated cost:
Short Term (0-2 years) Lo Lo Lo	Short Term (0-2 years) Long Term (2-4 or more years) Long term		Potentially very ex	pensive
Form Submitted by:	Saler	n Natural Haz	ards Mitigation Com	imittee
Action Item Status:	I			

EARTHQUAKE ACTION #1

Proposed Action Item: (from Action # 16)		Alignment with Plan Goals:		
Develop an inventory of un-reinforced masonry develop appropriate mitigation action items to of seismic events.	Protect lives. Protect existing and future development.			
Rationale for Proposed Action Item:				
• The City of Salem has numerous un-reinforced masonry structures in their downtown. Un-reinforced masonry structures are particularly susceptible to earthquakes, and if damaged, can disrupt businesses located in historic downtown buildings. Inventorying un-reinforced masonry structures and developing action items to address these buildings will help reduce the vulnerability to seismic events.				
 The Salem Natural Hazards Mitigation Conprobability of recurrence and a high vulner vulnerable buildings first, those made of un seismic events. The Disaster Mitigation Act of 2000 required to the set of t	 The Salem Natural Hazards Mitigation Committee identified seismic events as having a high probability of recurrence and a high vulnerability in the City of Salem. Addressing the most vulnerable buildings first, those made of un-reinforced masonry, will reduce the city's vulnerability to seismic events. The Disaster Mitigation Act of 2000 requires communities to identify actions and projects that reduce 			
the effects of hazards on the community, particularly to buildings and infrastructure $[201.6(c)(3)(ii)]$. Inventorying un-reinforced masonry structures will identify the major issues surrounding these buildings and what appropriate mitigation measures should be used to address these issues. In addition, protecting existing buildings and infrastructure will help reduce the negative impact of a spismic quant on the community.				
Ideas for Implementation:				
 Identify critical facilities constructed of un-reinforced masonry and develop appropriate mitigation action items or consider relocating the facility to a new building. Seek funding to develop programs to retrofit un-reinforced masonry buildings and provide outreach on seismic hazards. 				
Lead Agency: Salem Community Developm	nent Department			
Internal Partners:	External Partners	:		
Salem Urban Development, Public Works, Fire	FEMA, DOGAMI			
Timeline:	If available, estimation	ated cost:		
Short Term (0-2 years) Long Term (2-4 or more years) Long Term Long Term				
Form Submitted by: Salem Natural Haz	ards Mitigation Com	mittee		
Action Item Status:				

EARTHQUAKE ACTION #2

Proposed Action Item:		Alignment with Plan Goals:		
Identify and inventory critical faciliti retrofit.	es that require seismic	Protect existing and future development.		
		Increase cooperation and		
		coordination among stakeholders.		
Rationale for Proposed Action Iten	n:			
• The Salem Natural Hazards Mitigation Committee noted that certain critical facilities have a high vulnerability for seismic events. Seismically retrofitting these facilities will significantly reduce their vulnerability in the event of an earthquake.				
• Oregon Senate Bill 3 (2005) enables the Oregon Office of Emergency Management to develop a grant program to seismically rehabilitate critical public facilities. While the grant program is still being developed, conducting an inventory of critical facilities early will assist communities in obtaining funding once the grant program is in place.				
• The Disaster Mitigation Act of 2000 requires communities to identify comprehensive actions that protect new and existing buildings [201.6(c)(3)(ii)]. Seismically retrofitting existing critical facilities, including reservoirs and pump stations, will help Salem reduce their vulnerability to seismic events.				
• The Department of Geology and completed in 2007 of educationa school structures with a high or v	• The Department of Geology and Mineral Industries (DOGAMI) Statewide Seismic Needs Assessment completed in 2007 of educational and emergency service facilities in the state of Oregon identified 48 school structures with a high or very high likelihood of collapse in the event of a major earthquake. In			
addition, three fire structures and an earthquake. These facilities sl	five police structures had a hould be retrofitted according	high likelihood of collapse in the event of ngly to reduce the likelihood of collapse		
should an earthquake occur.				
Ideas for Implementation:	Ideas for Implementation:			
 Use DOGAMI's Seismic Needs Assessment of buildings in Salem to identify and prioritize building vulnerable to seismic events. Seek additional information from DOGAMI, if vulnerable reservoirs pump stations are not included in the Seismic Needs Assessment. Coordinate with OEM and FEMA to determine funding for conducting seismic retrofit of buildings 				
Lead Agency: Cit	y of Salem Emergency Mar	agement		
Internal Partners:	External Partn	ers:		
Salem Natural Hazards Mitigation	FEMA, OEM, D	OGAMI, Local School Districts		
Committee, Salem Community Deve	lopment			
Department, Salem Public Works				
Timeline:	If available, est	imated cost:		

Internal Partners:			External Partners:
Salem Natural Haza	rds Mitiga	tion	FEMA, OEM, DOGAMI, Local School Districts
Committee, Salem C	Communit	Development	
Department, Salem	Public Wo	rks	
Timeline:			If available, estimated cost:
Short Term (0-2 years)	Long Term (2-4 or more years)		Potentially very expensive, but costs can be offset by
	Long Term		grant programs.
Form Submitted by: Salem Natural Haz		em Natural Haz	ards Mitigation Committee
Action Item Status:			

EARTHQUAKE ACTION #3

Proposed Action Iten	n:		Alignment with Plan Goals:
Partner with the school districts to help identify and prioritize seismic retrofits to school district facilities.			Protect lives. Protect existing and future development.
Rationale for Propos	ed Action Item:		
 Due to the high concentration of students and the relative vulnerability of that population, schools have large negative impacts from seismic events. Seismically retrofitting these facilities will significantly reduce their vulnerability in the event of an earthquake. Oregon Senate Bill 3 (2005) enables the Oregon Office of Emergency Management to develop a grap program to seismically rehabilitate critical public facilities. While the grant program is still being developed, conducting an inventory of critical facilities early will assist communities in obtaining funding once the grant program is in place. The Department of Geology and Mineral Industries (DOGAMI) Statewide Seismic Needs Assessmet completed in 2007 of educational facilities in the state of Oregon identified 48 school structures with high or very high likelihood of collapse in the event of a major earthquake. These facilities should b retrofitted accordingly to reduce the likelihood of collapse in the event of an earthquake. The Disaster Mitigation Act of 2000 requires communities to identify comprehensive actions that protect new and existing buildings [201.6(c)(3)(ii)]. Seismically retrofitting existing critical facilitie including reservoirs and pump stations and especially schools, will help Salem reduce their vulnerability to seismic events. Ideas for Implementation: Use DOGAMI's Seismic Needs Assessment of Salem school facilities to identify and prioritize school district facilities that are vulnerable to seismic events. Educate school district officials about the effectiveness of natural hazard mitigation actions. Coordinate with OEM and FEMA to seek funding for conducting seismic retrofit of buildings. 			erability of that population, schools etrofitting these facilities will uake. rgency Management to develop a grant nile the grant program is still being ill assist communities in obtaining) Statewide Seismic Needs Assessment on identified 48 school structures with a earthquake. These facilities should be e event of an earthquake. lentify comprehensive actions that y retrofitting existing critical facilities, will help Salem reduce their icilities to identify and prioritize school ral hazard mitigation actions. ng seismic retrofit of buildings. ral Hazards Mitigation Committee.
Lead Agency: City	of Salem Emergency Ma	anagement	
Internal Partners:		External Partners	:
Salem Natural Hazards Mitigation Committee, Salem Community Development Department, Timeline:		FEMA, OEM, DOO private schools, Ch College	GAMI, Salem-Keizer School District, emeketa C.C., Willamette, Corban
Short Term (0-2 years)	Long Term (2-4 or more years)	Potentially very ext	pensive, but costs can be offset by
Long Term		grant programs.	pensite, our costs can be offset by
Form Submitted by:	Salem Natural Haz	ards Mitigation Com	mittee
Action Item Status:			

FLOOD ACTION #1

Proposed Action Item: (from Action # 8)		Alignment with Plan Goals:	
Develop a flood warning system on local creek with the 2000 Stormwater Master Plan and the Engineers (COE) Section 205 Flood Damage F Mill Creek.	Protect lives.		
Rationale for Proposed Action Item:			
• The Disaster Mitigation Act of 2000 requires that communities identify actions and projects that reduce the impact of a natural hazard on the community, particularly to new and existing buildings and infrastructure [201.6(c)(3)(ii)]. Developing a flood warning system on local creeks will help monitor flood levels and provide adequate warning to residents and businesses near flood prone areas to evacuate			
Ideas for Implementation:			
 Refine the existing continuous stream monitoring stations to include rating curves for improved flow measuring accuracy, and install monitors on currently un-monitored streams as funding allows. Perform a cost-benefit analysis of a local flood warning system on local creeks depending on the level of risk for each creek. Conduct a public awareness campaign targeting residents in the floodplain to educate them about the potential flood hazard their property faces. Make floodplain information available at the Marion County building permit counter and at the city of Salem. Lead Agency: Salem Public Works Department 			
Internal Partners:	External Partners	:	
Fire Department, Salem Community Development	FEMA, DLCD, Ar Polk Counties, MC District	my Corps of Engineers, Marion and SWCD, Santiam Water Control	
Timeline:	If available, estimation	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years) Long Term Long Term	-		
Form Submitted by: Salem Natural Haz	ards Mitigation Com	mittee	
Action Item Status:			
Proposed Action Item: (from action # 10)		Alignment with Plan Goals:	
---	--------------------------	--	--
Encourage landowners in the UGB to provide set-asides for open space for less intensive development in floodplains.		Increase cooperation and coordination among stakeholders. Protect the natural environment.	
Rationale for Proposed Action Item:			
 Often, homeowners and recent purchasers of property are unaware of the potential flooding hazards found on their property. Educating current property owners, as well as recent arrivals and prospective buyers, about the floodplain hazard on their property will enable property owners to be better prepared in the event of a flooding event. In addition, providing mitigation actions property owners can take will significantly reduce the impact of a flooding event. The Disaster Mitigation Act of 2000 requires that communities identify actions and projects that reduce the impact of a natural hazard on the community, particularly to new and existing buildings and infrastructure [201.6(c)(3)(ii)]. Encouraging landowners to provide set asides for open space on developments in the floodplain will reduce the impact of flooding events. 			
Ideas for implementation:			
 Conduct a public awareness campaign targeting residents in the floodplain to educate them about the potential flood hazard their property faces. Make floodplain information available at the Marion building permit counter and at the city of Salem. Educate current homeowners and prospective buyers of property within Salem's UGB about potential floodplain issues on their property and actions they can implement to mitigate the impacts of a flood. 			
Lead Agency: Salem Community Developm	ent Department		
Internal Partners:	External Partners	•	
Salem Public Works Department	DLCD, FEMA, Ma	rion and Polk Counties	
Timeline:	If available, estimation	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years) Ongoing Ongoing			
Form Submitted by: Salem Natural Haza	ards Mitigation Com	mittee	
Action Item Status:			

Proposed Action Item: (from Action # 12)Alignment with Plan Goals:				
Implement recommendations presented in the	2000 Stormwater	Protect existing and future		
Master Plan (SWMP) to reduce flood hazards	in the city.	development.		
	_	Protect the natural environment.		
Rationale for Proposed Action Item:				
• The City of Salem completed a Stormwate	er Master Plan in 200	0 whose overall goal is to cost		
effectively balance reductions in flood damages with improvements in stream water quality, reflecting				
the community's financial resources to sup	port such a broadene	ed program. The plan lists policies and		
projects. Stormwater management plays a	prioritized actions for the stormwater program and provides recommendations and costs for specific projects. Stormwater management plays an important role in flood mitigation, and the action items			
included in the stormwater plan will help r	educe the impact of f	flooding events in the city of Salem.		
• Stormwater management is a key element	in maintaining and er	nhancing a community's livability.		
There is a direct link between stormwater	and a community's su	irface and ground waters. As a		
community develops, the impervious surfa	ices that are created in	ncrease the amount of runoff during		
channels and prevent groundwater recharge	ologic cycle. without	t control, these conditions erode stream		
 The Disaster Mitigation Act of 2000 requi 	es that communities	identify actions and projects that		
reduce the impact of natural hazards on the	e community, particu	larly to new and existing buildings and		
infrastructure [201.6(c)(3)(ii)]. Implement	ting recommendation	s presented in the Stormwater Master		
Plan will help to reduce the impact of floo	ds in the City of Sale	m while improving water quality for		
surrounding rivers and streams.				
• The Disaster Mitigation Act of 2000 requi	• The Disaster Mitigation Act of 2000 requires communities to maintain the Hazard Mitigation Plan by			
having local governments incorporate the figure $(201.6(c)(4)(ii))$. Implementing the Storm	requirements of the p	an into other planning mechanisms		
hazards will coordinate existing plans and	policies with the mit	igation plan actions and simplify the		
implementation process by using existing	plans and agencies.	igation plan actions and simplify the		
Ideas for Implementation:				
Include City of Salem staff responsible for	implementing the St	tormwater Master Plan in Salem		
Natural Hazards Mitigation Committee me	etings.			
• Identify and prioritize stormwater action items based on the benefits they will have in reducing the				
impact of future flooding events.				
• Seek reactal funding unough the pre-disaster initigation (PDW) program and unough the Flood Mitigation Assistance (FMA) program to assist in implementing the Stormwater Master Plan				
Whitgation Assistance (1997) program to assist in implementing the Stormwater Waster Flair.				
Lead Agency: Salem Public Works Department				
Internal Partners:	External Partners			
Salem Community Development Dept.	Marion County, Po	olk County, FEMA		
Limenne:	If available, estimate	ated cost:		
Short Term (0-2 years) Long Term (2-4 or more years)	-			
Form Submitted by: Salem Natural Haz	zards Mitigation Com	mittee		
Action Item Status:				

Proposed Action Item: ()	from Action # 13)		Alignment with Plan Goals:	
Revise SRC 140 Floodplain Overlay Zones to increase flood			Protect existing and future	
freeboard at 2 feet above the 100-year elevation in areas whe			development.	
past floods (including the 1996 floods) have been higher than			Reduce economic loss.	
FIRM-predicted 100-year elevation.				
Rationale for Proposed A	Action Item:			
 The 2003 Salem Natural Hazards Mitigation Plan identified increasing flood freeboard to two? one foot above the 100-year elevation as an action item (see Action # 13, Appendix D. 2003 Salem Natural Hazards Mitigation Plan Action Items). Conversations with the Salem Natural Hazards Mitigation Committee identified the desire to increase flood freeboard to two feet in height, which would require revision of Salem Revised Codes (SRC) 140, but assist in reducing the impact of flooding events. The City of Salem has three repetitive flood loss properties. Increasing the flood freeboard to two feet may significantly reduce the impact of flooding of these properties if they were to be raised in the future. The City of Salem has a total of 3541 structures located in the 100-year floodplain. Increasing the flood freeboard to two feet above the 100-year elevation could significantly reduce the impact of flooding events or these structures should a future flooding event occur. The Disaster Mitigation Act of 2000 requires communities to identify actions and projects that reduce the effects of hazards on the community, particularly to current and future buildings and infrastructure [201.6(c)(3)(ii)]. Increasing the flood freeboard level to two feet will significantly protect structures in the floodplain during a 100-year flood event, and provide further protection during a 500-year flood 				
Ideas for Implementation	Ideas for Implementation:			
 Work with community members and neighborhood groups to discuss the benefits of raising the flood freeboard to two feet. Coordinate revisions to SRC 140 with National Flood Insurance Protection and Community Rating Service requirements to ensure compliance. 				
Lead Agency: Salem P	ublic Works Departn	nent		
Internal Partners:		External Partners	:	
Salem Community Develo	opment, Legal,	FEMA, Neighborh	ood Associations, Marion-Polk	
Public Works		Building Industry A	Association (MPBIA)	
Timeline:		If available, estimation	ated cost:	
<u>Short Term (</u> 0-2 years) <u>Long Term (</u> 2-4 or more years)				
Short Term				
Form Submitted by:	Salem Natural Haz	ards Mitigation Com	mittee	
Action Item Status:				

Proposed Action Item:		Alignment with Plan Goals:		
Improve the City of Salem's National Floo Program (NFIP) Community Rating Syster NFIP premiums.	d Insurance m (CRS) to reduce	Increase cooperation and coordination among stakeholders. Reduce economic loss.		
Rationale for Proposed Action Item:				
 The National Flood Insurance Program's (NFIP) Community Rating System (CRS) is a voluntary incentive program that recognizes and encourages community floodplain management activities that exceed the minimum NFIP requirements. As a result, insurance premiums under the NFIP are discounted to reflect the reduced flood risk resulting from the community actions meeting the three goals of the CRS: (1) reduce flood losses; (2) facilitate accurate insurance rating; and (3) promote the awareness of flood insurance. The City of Salem has entered the CRS program with a rating of eight. Implementing action items to improve the CRS rating will significantly reduce NFIP premiums on structures located within the floodplain. The Disaster Mitigation Act of 2000 requires communities to identify mitigation actions that address existing buildings and infrastructure [201.6(c)(3)(ii)]. Maintaining the status of the Community Rating System program can help the community to enhance mitigation efforts and decrease the vulnerability to floods. In addition, the Flood Mitigation Assistance Program requires that 				
Ideas for Implementation:	Ideas for Implementation:			
 Coordinate with the Department of Land Conservation and Development (DLCD) and FEMA to maintain the Community Rating System. Educate businesses and homeowners currently under the NFIP program about the CRS program and any mitigation actions they can implement to reduce their insurance premiums. Identify homes not in the NFIP that should have flood insurance. Develop mitigation activities to address repetitive and single loss flood properties in Salem, particularly in the area of McGilchrist Avenue and Pringle Road SE, adjacent to West Pringle Creek. 				
Lead Agency: Salem Public Works Department				
Internal Partners:	External Partners:			
Salem Community Development	DLCD, National Flood Insurance Program, FEMA, Marion and Polk Counties			
Timeline:	If available, estim	ated cost:		
Short Term (0-2 years) Long Term (2-4 or more years) Short Term				
Form Submitted by: Salem Natural Haz	ards Mitigation Com	mittee		
Action Item Status:				

WINDSTORM/WINTER STORM ACTION #1

Proposed Action Iter	m:			Alignment with Plan Goals:
Partner with publ about hazardous t event of a wind o	Partner with public and private utilities to educate the public about hazardous trees and the damage they can cause in the event of a wind or winter storm.			Reduce economic loss. Protect existing and future development.
Rationale for Propos	sed Action	Item:		
 Overhead electrical lines and other above ground utilities are subject to damage from nearby trees in high winds and winter storm damage. Post-disaster, it is difficult to remove debris from the downed utility lines and this difficulty delays the time for restoration of power to the community. Partnering with utility companies to maintain and remove hazardous trees, in addition to educating the public about the damage hazardous trees can cause, will help reduce risk of damage from severe wind and winter storms. The Disaster Mitigation Act of 2000 requires communities to develop comprehensive actions to reduce the impacts of natural hazards, with an emphasis on new and existing buildings and infrastructure [201.6(c)(3)(ii)]. Maintenance and removal of hazardous trees will reduce the impact of severe weather, and will continue power service to rural customers as well as ODOT, State Police, county sheriff, emergency services, telephone utilities, and cell phone companies. 				
Tueas for Implement	ideas for implementation:			
 Coordinate with t maintenance and Work with the co to damage from n Create a hazardou 	 Coordinate with the City of Salem Public Works Department to gather information about the maintenance and removal of hazardous trees. Work with the community and City of Salem Public Works Department to identify areas that are prone to damage from nearby trees and perform the necessary maintenance or removal of those trees. Create a hazardous tree inventory. 			
Lead Agency:		Salem Public	Works Department	
Internal Partners:			External Partners	:
Salem Community Se , Salem Fire Departm	Salem Community Services Parks OperationsODOT Portland General Electric, Salem Electric, Salem Fire DepartmentODOT Portland General Electric, Salem Electric			eneral Electric, Salem Electric
Timeline:	Timeline:			ated cost:
Short Term (0-2 years)	Short Term (0-2 years) Long Term (2-4 or more years) Long Term Long Term			
Form Submitted by:	: Saler	n Natural Haz	ards Mitigation Com	mittee
Action Item Status:	I			

WILDFIRE ACTION #1

Proposed Action Item:			Alignment with Plan Goals:
Conduct wildfire prevention outreach, as outlined in the Marion County Community Wildfire Protection Plan (CWPP), to residents near the wildland-urban interface.			Increase cooperation and coordination among stakeholders. Protect lives.
Rationale for Proposed Action	Item:		
 The Oregon Department of Forestry (ODF) recently completed a Communities at Risk Assessment for the City of Salem that shows areas in northwest and south Salem that are at high risk to wildfire events (See Map 2.1.5: Fire Hazard Areas). Although these areas are just outside of the Salem city limits, they are vulnerable to wildfire events that could impact residents within the city limits. Conducting wildfire prevention outreach to residents near these areas can significantly reduce the vulnerability of the neighborhoods to wildfire events. Interviews with Salem Fire Department staff indicate that the areas with the highest risk have the steepest slopes, the right fuels, and high valued property. The areas outlined by the ODF Communities at Risk Assessment show that many of the areas at risk are near steep slopes and have combustible fuels. Conducting wildfire prevention outreach can help to reduce vulnerability of residents to wildfire events. The Marion County Community Wildfire Protection Plan (2008) outlines strategies for conducting wildfire prevention outreach to residents living in the wildland-urban interface. Conducting wildfire prevention outreach using the CWPP will help to integrate mitigation into existing plans and policies as required by the Disaster Mitigation Act of 2000 [201.6(c)(4)(ii)]. The Disaster Mitigation Act of 2000 requires communities to identify mitigation actions that address new and existing buildings and infrastructure [201.6(c)(3)(ii)]. Conducting wildfire prevention outreach measures as outlined in the Marion County CWPP will help to protect new and existing 			
Ideas for Implementation:			
 The Marion County Community Wildfire Protection Plan contains several action items for reducing the impacts of wildfire on communities throughout the county, including actions to conduct public outreach about fuels reduction and defensible space (see Chapter 6: Action Plan, Marion County CWPP). Using these action items can assist in reducing the impact of wildfire on the City of Salem. Coordinate with responsible agencies listed in the Marion County CWPP to implement action items. 			
Lead Agency:	Salem Fire Departr	nent	
Internal Partners:		External Par	rtners:
Salem Public Works and Community Development Departments, Police Department, CommunityOregon Department of Forestry, Marion County Fire District #1, South Suburban Fire District, Neighborhood Associations			rtment of Forestry, Marion County #1, South Suburban Fire District, d Associations
Timeline:		If available,	estimated cost:
Short Term (0-2 years) Long Term (2-	-4 or more years)		
Short Term			
Form Submitted by: Sale	m Natural Hazards M	Intigation Com	mittee
Action Item Status:			

HAZARDOUS MATERIALS ACTION #1

Proposed Action It	em: (f	rom Action # 5)		Alignment with Plan Goals:
Map facilities that handle or contain hazardous materials, rank them based on their level of risk, and refine response strategies for each situation in the event of an accident.			Increase cooperation and coordination among stakeholders. Protect lives. Protect the natural environment.	
Rationale for Propos	ed Act	ion Item:		
 The City of Salem has identified and mapped hazardous materials located in the city. These maps need to be updated to determine the number and types of natural hazards present, and their level of risk. The Salem Natural Hazards Mitigation Committee indicated how the railroad running near the Capital Mall area in Salem is an area for potential concern because of the significant amount of hazardous materials that run through the area each year. Accidents with people and automobiles could derail cars and have the potential to spill hazardous materials in the Capital Mall area, affecting City and State operations. Refining response strategies for accidents on the railroad line would reduce the vulnerability of the City of Salem to hazardous materials incidents. The Disaster Mitigation Act of 2000 requires communities to identify actions and projects that reduce the effects of hazards on the community, particularly to buildings and infrastructure [201.6(c)(3)(ii)]. Identifying facilities that handle or contain hazardous materials, ranking them based on their level of risk, and developing appropriate response strategies will help reduce the negative impact of hazardous materials on the population in Salem and improve disaster response efforts. Ideas for Implementation: Contact businesses and property owners with hazardous materials about strategies they can implement to reduce the impacts of hazardous materials in their immediate area. Coordinate response strategies with alert warning systems to minimize potential exposure to hazardous materials 			ated in the city. These maps need to be , and their level of risk. railroad running near the Capital Mall area nount of hazardous materials that run ould derail cars and have the potential to State operations. Refining response ility of the City of Salem to hazardous actions and projects that reduce the effects ure [201.6(c)(3)(ii)]. Identifying facilities ir level of risk, and developing appropriate materials on the population in Salem and ut strategies they can implement to reduce e potential exposure to hazardous materials. and neighbors to reduce exposure to	
• Identify vulnerabl	e areas	along the Union	Pacific and Burlington Nort	hern Santa Fe railroad lines and coordinate
Lead Agency: Sal	em Fir	e Department	gies for reducing accidents a	iong the famoat mies.
Internal Partners:			External Partners:	
Salem Emergency Mar Works	nagem	ent, Public	OSHA, Salem Chamber of ODOT, OEM, State Police	Commerce, Neighborhood Associations, , State Fire Marshall
Timeline:			If available, estimated cos	st:
Short Term (0-2 years)	Short Term (0-2 years)Long Term (2-4 or more years)			
		0.1		
Form Submitted by:		Salem Natural	Hazards Mitigation Commit	ee
Action Item Status:				

6.0 **REFERENCES**

Specific to City Of Salem Area

City of Salem, Salem Revised Code (SRC)

City of Salem Landslide Ordinance (SRC 69)

City of Salem, 1992, Salem Comprehensive Plan (revised September 1999)

City of Salem, 1997, Post Flood Report - Floods of February 7-9, 1996. Public Works Department

City of Salem, 1990, Salem/Mill Creek, Oregon; Draft Feasibility Report on Flood Damage Reduction

City of Salem, 1999, Emergency Management Plan

City of Salem, 1999, Request for Proposals - Development of a Natural Hazard Mitigation Plan

Marion County, 1999 (update), Manual for Completion of Hydrogeology Reviews and Studies in Compliance with

Marion County Sensitive Groundwater Overlay Zone.

Marion County Natural Hazards Mitigation Plan

Montgomery Watson, City of Salem, Stormwater Master Plan, September 2000.

Montgomery Watson, City of Salem, Drainage System Improvement Plan: A Technical Supplement to the Stormwater Master Plan (September 2000)

Montgomery Watson, City of Salem, Oregon Stormwater Management Program Plan: A Technical Supplement to the Stormwater Master Plan, September 2000

National Oceanic and Atmospheric Administration, Climactic Data Center, Storm Events Database

Oregon Dept. of Land Conservation & Development, 1998, Oregon Model Flood Damage Prevention Ordinance. Oregon Dept. of Land Conservation & Development, 1999, Local Planning for Natural Hazards - Comprehensive Plan Evaluation Questionnaire (October draft)

Oregon DOGAMI, 1996, Relative Earthquake Hazard Maps of the Salem East and Salem West Quadrangles, Marion and Polk Counties, Oregon (Dept. of Geology and Mineral Industries) GMS-105

Oregon DOGAMI, 1999, Water-Induced Landslide Hazards, Western Portion of the Salem Hills, Marion County, Oregon, IMS-6

Oregon DOGAMI, 1999, Water-Induced Landslide Hazards, Eastern Portion of the Eola Hills, Polk County, Oregon, IMS-7

Oregon Revised Statutes (*see Appendix A. of this Natural Hazard Mitigation Plan for details*) Polk County Natural Hazards Mitigation Plan

US Army Corps of Engineers, 1998, Hydrologic Engineering Management Plan (HEMP) For Salem/Mill Creek, Section 205 Study, CENWP-PE-HY(Draft, part of City of Salem Quality Plan for Salem, Mill Creek, Oregon).

US Army Corps of Engineers, 1999, Salem, Oregon (Mill Creek), Section 205 Study: Hydrologic and Hydraulic Analysis (Draft Interim Report), DCAW57-96-0011 (Task 005).

US Army Corps of Engineers, 2002, Salem, Oregon, Mill Creek Watershed Section 205 Flood Control Feasibility Study.

US Geological Survey, 1989, Geologic map, Salem 1° x 2° quadrangle, I-1893.

US Geological Survey,

6.1 Principal References

(Unknown), <u>Annex to State of Oregon Natural Hazards Mitigation Plan Pursuant to Disaster No. FEMA-1061-DR-OR</u> (July 8 and 9, 1995 Wasco County Flash Flood), undated.

Beaulieu, John, and Olmstead, Dennis, <u>Risk Reduction for Communities with Geologic Hazards, A Guide: Salem</u> <u>Landslide Study as a Pilot Effort,</u> State of Oregon, Oregon Department of Geology and Mineral Industries, Portland, OR, Draft of 3/18/99.

Beaulieu, John, <u>Risk Reduction from Geologic Hazards in Oregon</u>, presented at Oregon Emergency Managers Annual Meeting, Eugene, OR, and Oregon Dept. of Geology and Mineral Industries, Portland, OR, 8/26/97.

Biennial Report for 1997 – 1999, from Oregon's Department of Land Conservation and Development to the Seventieth Legislative Assembly, *Reducing Risks and Costs of Natural Hazards*, February, 1999, P. 26.

Booth, Derek B., <u>Urbanization and Natural Drainage System--Impacts</u>, Solutions, and Prognoses, *The Northwest Environmental Journal*, Vol. 7, No. 1, Spring/Summer 1991.

City of Vernonia, Hazard Mitigation Committee, Vernonia Natural Hazard Mitigation Plan, September 1996.

Federal Register, Federal Emergency Management Agency (FEMA), 44 CFR Parts 201 and 206, *Hazard Mitigation Planning and Hazard Mitigation Grant Programs; Interim Final Rule*, February 26, 2002.

FEMA, <u>How to Determine Cost-Effectiveness</u> of Hazard Mitigation Projects, Interim Edition, December, 1996, <u>Mitigation Directorate</u>

FEMA and Oregon State Police (Office of Emergency Management), <u>Interagency Hazard Mitigation Team Report for</u> the Crook County Flooding Disaster, FEMA DR-1221-OREGON, June 1998.

FEMA, Mitigation Directorate, <u>Technical Information on Elevating and Relocating Residential Buildings in the State of</u> <u>Oregon</u>, undated.

FEMA, Multi-Hazard Identification and Risk Assessment: A Cornerstone of the National Mitigation Strategy, 1997.

FEMA, <u>National Flood Insurance Program Community Rating System Example Plans</u> (Washington, DC: FEMA, July 1996).

FEMA, The Project Impact Hazard Mitigation Guidebook for Northwest Communities, Washington, DC, 1998.

FEMA Region IV (with assistance from the Center for Urban and Regional Studies at the University of North Carolina-Chapel Hill), Local Natural Hazard Mitigation Planning Manual, Atlanta, GA: draft, undated.

FEMA Region X Interagency Hazard Mitigation Team, <u>Interagency Flood Hazard Mitigation Report</u>, <u>In Response to</u> <u>the January 18, 1990 Disaster Declaration</u>, <u>State of Oregon</u>, <u>Tillamook and Clatsop Counties</u></u>, FEMA-853-DR-OR, undated.

Goettel & Associates Inc., <u>Regional All Hazard Mitigation Master Plan for Clackamas County</u>, Draft (Partial), August 11, 1998.

Hofmeister, J., <u>Slope Failure in Oregon During 1996/97 Storm Events</u>, Department of Geology and Mineral Industries, 1999.

Interagency Hazard Mitigation Team Report, <u>February 1996 Landslides & Stream Erosion (draft)</u>, FEMA 1099-DR-OR, undated.

Marion County, <u>Marion County Natural Hazard Mitigation Plan</u>, Mid-Willamette Valley Council of Governments, 2006.

Metro, Natural Hazard Mitigation Workshop Proceedings, March 13-14, 1997 (Portland, OR: Metro, undated).

Metro, Other Natural Hazards: Information in the Metro Region (Portland, OR: Metro, March 10, 1997.

National Research Council, <u>Reducing Losses From Landsliding in the United States</u>, National Academy Press, Washington, DC, 1985.

Oregon Department of Forestry, Landslides and Public Safety, Issue Paper, April 10, 1998.

Oregon Department of Forestry, 1999, Storm Impacts and Landslides of 1996 - Forest Practices Technical Report No. 4 (authored by E.R. Robinson, K.A. Mills, J. Paul, and L. Dent with the Oregon Department of Forestry and A. Skaugest, Oregon State University).

Oregon State Department of Geology and Mineral Industries, 1999, <u>Geologic Hazards: Reducing Oregon's Losses</u> (Special Paper 32 by John Beaulieu and Dennis Olmstead).

Oregon State Police, Office of Emergency Management, 1991, Action Plan to Mitigate Flood and Wind Hazards in Clatsop and Tillamook Counties, June 1991.

Oregon State Police (Oregon State Police, Office of Emergency Management)) and Federal Emergency Management Agency, Region 10, <u>Hazard Mitigation: Early Implementation Strategy for the 1996 Oregon State Flood, DR-1099-OR</u> (Salem, OR: OEM, March 15, 1996).

Oregon State Police (Oregon State Police (Office of Emergency Management)) and Federal Emergency Management Agency, Region 10, <u>Homeowner's Landslide Guide</u> (Salem, OR: OEM, undated).

Oregon State Police (Office of Emergency Management) and Federal Emergency Management Agency, <u>Interagency</u> <u>Hazard Mitigation Team Report, FEMA-DR-1149-OR and FEMA-DR-1160-OR</u> (Salem, OR: OEM, March 1997).

Oregon State Police (Office of Emergency Management) and Federal Emergency Management Agency, <u>Reducing</u> <u>Windstorm Damage to Electric Utilities</u>, Interagency Hazard Mitigation Team Report for the Western Oregon Windstorms of December 10-12, 1995 (FEMA-1107-DR-OR), undated.

Oregon State Police (Office of Emergency Management) and Federal Emergency Management Agency, <u>Interagency</u> <u>Hazard Mitigation Team Report, FEMA-DR-1099-OR (February 1996 Flooding, Landslides, and Stream Erosion in the State of Oregon),</u> (Salem, OR: OEM, 1996).

Oregon State Police (Office of Emergency Management), <u>Elevation Guidelines for Oregon Flood Disasters DR-1099/1149/1160</u>, and Flood Mitigation Assistance (FMA) Program, 3/19/98 revision.

Oregon State Police (Office of Emergency Management), <u>Oregon Revised Statutes, Chapters 401, 445 and Others</u>, 1993.

Oregon State Police (Office of Emergency Management), <u>State of Oregon Natural Hazards Mitigation Plan</u> (Salem, OR: OEM, 10/97).

Oregon State Police (Office of Emergency Management), 1996 (and revisions), Hazard Analysis (guidance and worksheets).

Oregon Natural Hazards Workgroup, 2002, <u>Local Natural Hazard Mitigation Plans: An Evaluation Process</u>, Oregon State Police (Office of Emergency Management) and Department of Land Conservation and Development (DLCD).

Oregon's Statewide Planning Goals & Guidelines, 1995 Edition, Goal 7: Areas Subject to Natural Disasters and Hazards, Department of Land Conservation and Development (DLCD), Salem OR, P. 11.

<u>Oregon's Statewide Planning Goals & Guidelines, February 1, 2001 DRAFT.</u> *Goal 7: Areas Subject to Natural Disasters and Hazards,* Department of Land Conservation and Development (DLCD), Salem OR, P. 3

Oregon's Statewide Planning Goals & Guidelines, September 28, 2001. Goal 7: Areas Subject to Natural Disasters and Hazards, Department of Land Conservation and Development (DLCD), Salem OR.

Portland State University, Department of Geology, <u>Landslides in the Portland Area Resulting from the Storm of February</u>, 1996: Inventory Map, Database and Evaluation, Draft (Portland, OR: PSU, April 6, 1998).

Protecting Our Region's Rivers, Floodplains and Wetlands, METRO Regional Services, Portland, OR, June, 1998.

Scofield, David and Jochim, Candace, <u>Environmental, Groundwater and Engineering Geology: Applications from</u> <u>Oregon, Edited by Scott Burns</u>, Portland State University, <u>Engineering Geology of Transportation Routes in Oregon</u>, Star Publishing Co., Belmont, CA, 1998, P. 193.

State of Oregon, Department of Land Conservation & Development, <u>Report to the 70th Legislative Assembly</u>, August 1999.

State of Oregon, Joint Interim Task Force on Landslides and Public Safety, <u>Report to the 70th Legislative Assembly</u>, October 7, 1998.

State of Oregon, Oregon Progress Board, <u>Oregon Shines II: Updating Oregon's Strategic Plan</u> (Salem, OR: OPB, January 21, 1997).

State of Oregon, State Flood Control Plan Task Force, Report to the 70th Legislative Assembly (undated).

Taylor, G.H. and Hatton, R.R., 1999, The Oregon Weather Book - A State of Extremes (Oregon State University Press).

University of Oregon, Community Planning Workshop, <u>Implementation of Goal 7: An Evaluation and Discussion of Hazards Planning in Oregon</u>, prepared for the Oregon Department of Land Conservation and Development, July 1998.

Wallin, Phil, Restoring the Floodplain of Oregon's Willamette River, River Voices, Summer 1997: 17-19.

Williams, Philip and Coulton, Kevin, <u>Reducing Flood Risk and Restoring Salmon: A Blueprint for Oregon's</u> <u>Integrated River Management Strategy</u>, Prepared for The U.S. Fish and Wildlife Service, Oregon State Office, Portland, OR.

Wold, Robert and Jochim, Candace, <u>Landslide Loss Reduction</u>: A Guide for State and Local Government Planning, Colorado Geological Survey, Dept. of Natural Resources, Denver, CO, 1989.

Appendix A. Additional Federal and State Government Mitigation Information

The Federal *Robert T. Stafford Relief and Emergency Assistance Act* provides a specific focus on loss prevention. Section 409 of the act and its implementing regulations (44 Code of Federal Regulations, Part 206, Subpart M, Hazard Mitigation Planning) requires states, as a condition for receiving any Federal disaster grant or loan, to evaluate the impacts of natural hazards within designated disaster areas, to identify actions that will reduce the effects of such hazards, and to prepare and implement hazard mitigation plans to fulfill the act's intent.

Further information about this act and implementing regulations can be obtained from the Federal Emergency Management Agency's web site: <u>http://www.fema.gov</u> The regulations for implementing Section 409 of the Stafford Act can be found in 44 CFR Section 206, Subpart M.

- Within 15 days of any presidential major disaster declaration, a Federal Interagency Hazard Mitigation Team (IHMT) is required to submit a hazard mitigation report for the counties included in the declaration.
- Within 90 days, and based on the IHMT report, the State Hazard Mitigation Officer (SHMO) must then revise the state's natural hazards mitigation plan taking into account lessons learned from the disaster.
- In many cases, the SHMO will coordinate the preparation of separate event-specific mitigation plans, which become references to this plan so a connection and continuity is maintained throughout the program over time.
- In addition to the above reference, Federal regulations governing floodplain management and the protection of wetlands are contained in 44 CFR Part 9.

Two teams are prescribed in the Federal regulations: the Hazard Mitigation Survey Team (HMST) and the Interagency Hazard Mitigation Team (IHMT). These are different from Oregon's Governor's Interagency Hazard Mitigation Team (GIHMT), but they do involve some of the same state agencies.

• According to 44 CFR Section 206.401, the HMST means "the FEMA/State/Local survey team that is activated following disasters to identify immediate mitigation opportunities and issues to be addressed in the section 409 hazard mitigation plan. The team may include representatives of other Federal agencies, as appropriate."

- In contrast, the IHMT focuses primarily on floods. The same regulation notes that this is "the mitigation team that is activated following flood related disasters pursuant to the July 10, 1980 Office of Management and Budget Directive on Nonstructural Flood Protection Measures and Flood Disaster Recovery, and the subsequent December 15, 1980 Interagency Agreement for Nonstructural Damage Reduction." While this team's report is essentially a Federal report, the team is a joint intergovernmental team that generally seeks to accomplish the following:
- Identify areas of significant hazards.
- Evaluate sites of significant hazards and determine impacts.
- Identify areas of damage that would require reconstruction to the standards of the National Flood Insurance Program (NFIP) or to state regulation.
- Review and evaluate any applicable land use regulations, construction standards, or other hazard mitigation measures.
- Review and evaluate existing emergency plans, including warning and evacuation plans.
- Review other pertinent information, such as urban renewal, rehabilitation or master plans.
- Identify and evaluate measures to mitigate the disaster impacts.
- Recommend appropriate hazard mitigation measures.
- Coordinate and take action necessary to implement the recommendations.

While several Federal agencies have special disaster related authorities, such as the U.S. Army Corps of Engineers, Small Business Administration and the Department of Transportation, for purposes of this city-level hazard mitigation planning the most important legislative authority is Public Law 93-288, as amended, commonly known as "The Robert T. Stafford Disaster Relief and Emergency Assistance Act." Now known as the "Disaster Relief Act," it has three pertinent sections related to hazard mitigation:

- Section 404: Defines the Hazard Mitigation Grant Program (HMGP). A matching fund program (currently 75% Federal and 25% non-Federal), it funds projects which are considered cost-effective and which substantially reduce the risk of future damage, hardship, loss, or suffering in any area affected by a major disaster. If no unmet needs exist in the disaster area, remaining funds can be applied to projects anywhere in the state.
- Section 406: Defines the Public Assistance Program, which provides funding to eligible government agencies for rebuilding destroyed or damaged public facilities. Funds for hazard mitigation can be included in approved projects during the rebuilding process.
- Section 409: Requires that as a condition of receiving Federal disaster assistance, states must develop or revise hazard mitigation plans which evaluate natural hazards, and make recommendations for mitigating these hazards in the geographic areas receiving such assistance. Statewide plans that address multiple hazards and entire states also are covered under Section 409 which has been renamed to Section 322 of the amended Stafford Act.
- The Stafford Act was amended by the Disaster Mitigation Act of 2000 (PL 106-390). FEMA released a new interim final regulation, dated February 26, 2002, on mitigation planning for state and local governments that implements the local mitigation plan requirements section of the amended Stafford Act. Note that per the new FEMA requirements, local jurisdictions must have plans approved by FEMA by November 1, 2004, to be eligible for Hazard Mitigation Grant Program project grants for Presidentially-declared disasters after this date.

Selected mitigation-related federal authorities that address actions to help minimize future losses are noted below. Some can be taken during the "build-up" phase of a potential disaster, others have longer term pre-disaster mitigation possibilities, and some, while available only in Presidentially-declared situations, have provisions that if applied have the potential for reducing future losses (e.g., a requirement to follow current building codes). Further information can be obtained from the *Catalog of Federal Disaster Assistance*, published by the General Services Administration (GSA); *Disaster Assistance: A Guide to Recovery Programs*, published by the FEMA (#229-4, November 1995), and the *Federal Response Plan*, also published by FEMA (April 1999). The selections include:

- Small Business Act of 1953, as amended and others: Provides low-interest, long-term loans to repair or replace damaged personal and real property, but within specified limits. The "loans may be increased up to 20 percent for mitigating devices to protect real property from possible future disasters of the same kind." In addition, the Small Business Administration outlined rules for hazard mitigation loans. In July of 1999, Congress enacted Public Law 106-24, which appropriated \$15 million to the Small Business Administration (SBA) for a pilot program to provide disaster mitigation loans to small businesses in support of FEMA's community based initiative, previously referred to as, Project Impact. This rule amends SBA's regulations allowing pre-disaster mitigation loans so that small businesses may install mitigation devices to prevent future damage. It states that mitigation can include such activities as elevating flood-prone structures; constructing retaining and sea walls; grading and contouring land; relocating utilities; and retrofitting and strengthening structures to protect them against high winds, earthquakes, floods, wildfires, and other natural hazards. Loans are available only to those small businesses located in pre-disaster mitigation communities, and applicants may borrow up to \$50,000 per year at four percent interest per year or less. The SBA amended its regulations again in October of 2002 to further implement the Pre-Disaster Mitigation Program. The new regulations require mitigation measure to conform with the priorities and goals of the State or local government's mitigation plan. To demonstrate this, the applicant must obtain a written statement from the State or local Emergency Management Director confirming this fact. Applicants may also submit 4 years worth of projects within FY2003 since funding was delayed for FY2000, 2001, and 2002. Responsible agency: Small Business Administration.
- Flood Control Act of 1941, as amended and others: While there must be an "immediate threat of unusual flooding," authorized mitigation assistance of a temporary nature includes removal of waterway obstructions, actions to prevent the failure of dams, and "work necessary to prepare for abnormal snowmelt." Responsible agency: Department of the Army, Office of the Chief of Engineers.
- Fishery Conservation and Management Act of 1996: Project grants to eligible recipients can be used for various purposes, including "restoring fisheries or preventing future failures," as long as the project meets specified criteria. Responsible agency: National Marine Fisheries Service.
- Housing Act of 1949, as amended and others: Provides for direct loans to qualified lower income rural families "to meet emergency assistance needs resulting from natural disasters ... or improve dwellings in rural areas." Responsible agency: Department of Agriculture, Rural Housing Service.
- Federal-Aid Highway Program, 23 United States Code, as amended: Among a wide array of transportation-related activities, includes formula and project grants intended "to foster safe highway design" and "to replace or rehabilitate deficient or obsolete bridges," including activities related to capital improvement projects for "safety reasons." Responsible agency: Department of Transportation, Federal Highway Administration.

NATURAL HAZARD MITIGATION PLAN City of Salem

State of Oregon

Chaired by the Director of Oregon Military Department (Office of Emergency Management) (OEM) or the Director's designee, the Hazard Mitigation Grant Review Board, a sub-committee of the State's Interagency Hazard Mitigation Team, is an intergovernmental body which reviews, ranks, and selects projects for funding under Section 404 of the Federal Stafford Act. It uses the hazard mitigation plans for the geographic area in question to evaluate proposed projects for funding. The criteria used are those of the Stafford Act and several specific Oregon criteria. The board may also be assembled to evaluate proposed hazard mitigation projects, for which no Federal grant funding is available, if the state has an interest in funding such projects.

The City of Salem, in the Salem Area Comprehensive Plan (SACP), recognizes that the State of Oregon Land Use Goals "are the final standard whereby provisions of the comprehensive plan are to be interpreted". The comprehensive plan is described in Oregon Revised Statute (ORS) 197.015(5), conforming to Statewide Planning Goals promulgated by the Oregon Land Conservation and Development Commission (LCDC.). The Commission's Goal 7 requires every community in an area subject to one or more natural hazards to have a comprehensive plan element that addresses that or those natural hazards.

During the development of community plans, there is a need to consider the primary natural hazards including floods (coastal and riverine), landslides, earthquakes and related hazards, tsunamis, coastal erosion, and wildfires.

LCDC adopted amendments to Statewide Planning Goal 7 on September 28, 2001, with an effective date of June 1, 2002. The amendments to Goal 7 establish a timeline and procedure for local governments to respond to certain new information on natural hazards. Goal 7 now requires local governments to evaluate the risk to people and property from the new hazard information and to allow an opportunity for citizens to review and comment on the information. Based on the evaluation and citizen comments, local governments will incorporate the new hazard information into their comprehensive land use plans as necessary.

When local governments respond to new hazards information through amendments to their comprehensive land use plans and implementing measures, the new Goal 7 language encourages them to:

a. Avoid development in areas of natural hazards where the risk to people and property cannot be mitigated; and b. Prohibit siting of "critical facilities" (e.g., fire stations, hospitals) in hazard areas unless the facility is needed in the hazard area to provide emergency response services in a timely manner.

The new guidelines highlight the relationship of natural hazards to protection of natural resources, discuss the importance of coordinating land use decisions with emergency management programs, and encourage local governments to adopt measures that are more protective than the minimum National Flood Insurance Program changes.

Also of significance on a State level is preparation of the Oregon Military Department (Office of Emergency Management) Plan (OEMP). This plan includes three volumes: Part I - Preparedness and Mitigation of the OEMP, which includes the plans and guidance necessary for the state to prepare to reduce the effects of a disaster, Part II - Emergency Operations Plan, and Part III, Relief and Recovery.

Important state legislation, pertinent to hazard mitigation, has been adopted over many years. Some of the key chapters of the Oregon Revised Statutes (ORS) are summarized below:

- ORS, Chapter 196: Deals with wetlands, and ORS 196.800 to 196.095 are concerned with filling and removing of materials from the "the beds and banks of the waters of this state." Many of these provisions have a tangential effect on floodplain management and flood hazard mitigation.
- ORS, Chapter 197: Provides the basis for comprehensive land use planning in the State of Oregon, including provisions governing development in floodplains and in areas of geological hazard (Goal 7) which are intended to mitigate the effects of such hazards.
- ORS, Chapter 401: Includes many of the state's emergency management statutes, one section of which states that the general purpose of the law is to "reduce the vulnerability of the State of Oregon to loss of life, injury to persons or property, human suffering, and financial loss resulting from emergencies."
- ORS, Chapter 477: Addresses the fire protection of forests and vegetation, including sections on urban interface fire protection, hazard abatement, fire abatement, fire prevention, and related sections.
- ORS, Chapter 516: Creates and defines the duties of the Department of Geology and Mineral Industries (DOGAMI). Section 516.030(3) directs DOGAMI to administer, on a cooperative basis, studies and programs that will "reduce the loss of life and property by understanding and mitigating geological hazards."
- ORS Chapter 455: Provides legal authority for the Building Codes Division's (BCD) natural hazard mitigation activities in ORS 455.020 (code adoption), .220 (training), .440 (site soil analysis), .446 (construction in tsunami zones), .447 (seismic site hazard analysis), and .448-.449 (entry and inspection of earthquake damaged buildings).
- ORS Chapter 527: Authorizes the statewide regulation of forest practices. The DOF provides leadership in forest policy and programs including forest fires and management of forest practices in order to mitigate hazards.
- ORS Chapter 1103: Establishes policy for protection of public from rapidly moving landslides. Provides for landslide mapping by the state (DOGAMI and DOF). The law provides DLCD with grants in order to fund development of statewide model ordinances and model Transfer of Development Rights (TDR) measures necessary for local government to regulate in landslide areas. Local governments must regulate in mapped areas, but must adopt a TDR program if regulations result in landowner costs that would exceed certain amounts set in the ORS.
- ORS Chapters 98.805-992, 368-039, 453, 476, 477: Authorizes a wide variety of education and training programs, inspections, investigative and information reports and other activities related to fire and hazardous materials incident prevention, safety, and management.
- ORS Chapters 196 and 390: Defines the Division of State Land's (DSL) roles related to issue of removal and fill permits or enforcement actions on public and private waterways, wetlands, the Pacific Ocean, and other waters of the state.
- ORS Chapter 527: Provides authority to the Department of Forestry (DOF) for insect and disease control for the purposes of protecting forested land resources.
- ORS Chapter 540: Provides the Water Resources Department (WRD) statutory authorities for dam safety and a statewide hydrographic program for measuring river and stream flows.
- ORS Chapter 569: Provides authority to the Department of Agriculture (DOA) for its water quality and soil conservation measures.
- ORS Chapter 570: Provides authority to the Department of Agriculture for its pest and disease control programs.

• ORS Chapter 810: Designates the Department of Transportation as the "road authority for all state highways" and specifies a wide range of maintenance, operations, and analysis activities related to hazard mitigation (e.g., drainage maintenance, culvert inventory, and bridge seismic retrofit program).

Local governments are represented by several associations that play important roles in hazard mitigation. These include the Association of Oregon Counties (AOC), League of Oregon Cities (LOC), Oregon Special Districts Association, and the Oregon State Police, Office of Emergency Management Association (OEMA). For example, the AOC, LOC, and OEMA sit on the Hazard Mitigation Grant Review Board, all advocate legislation and/or regulatory changes, and all provide contacts with their members as the need arises.

Appendix B. ORS 455.477 (excerpt) - Essential Facility, Major Structure, Hazardous Facility, and Special Occupancy Structures

From 455.447 - Regulation of certain structures vulnerable to earthquakes and tsunamis [note: ORS 455.446, ORS 455.447, and OAR Chapter 632 Div.5 apply to constructing such facilities in tsunami inundation zones (such zones do not exist within the Salem UGB)].

- (1) As used in this section, unless the context requires otherwise:
 - (a) "Essential facility" means:
 - (A) Hospitals and other medical facilities having surgery and emergency treatment areas;
 - (B) Fire and police stations;
 - (C) Tanks or other structures containing, housing or supporting water or fire-suppression materials or equipment required for the protection of essential or hazardous facilities or special occupancy structures;
 - (D) Emergency vehicle shelters and garages;
 - (E) Structures and equipment in emergency-preparedness centers;
 - (F) Standby power generating equipment for essential facilities; and
 - (G) Structures and equipment in government communication centers and other facilities required for emergency response.
 - (b) "Hazardous facility" means structures housing, supporting or containing sufficient quantities of toxic or explosive substances to be of danger to the safety of the public if released.
 - (c) "Major structure" means a building over six stories in height with an aggregate floor area of 60,000 square feet or more, every building over 10 stories in height and parking structures as determined by Department of Consumer and Business Services rule.
 - (d) "Seismic hazard" means a geologic condition that is a potential danger to life and property which includes but is not limited to earthquake, landslide, liquefaction, tsunami inundation, fault displacement, and subsidence.
 - (e) "Special occupancy structure" means:

(A) Covered structures whose primary occupancy is public assembly with a capacity greater than 300 persons;

(B) Buildings with a capacity greater than 250 individuals for every public, private or parochial school through secondary level or child care centers;

(C) Buildings for colleges or adult education schools with a capacity greater than 500 persons;

(D) Medical facilities with 50 or more resident, incapacitated patients not included in subparagraphs (A) to (C) of this paragraph;

(E) Jails and detention facilities.

Appendix C. Benefit-Cost Methodologies For Use in Evaluating the Cost Effectiveness of Proposed Hazard Mitigation Measures

1.0 Introduction

This appendix outlines the basic criteria and approach to determining the economic feasibility of natural hazard mitigation measures. Determining the economic feasibility of mitigating natural hazards can be classified according to the type of decision being made and the perspectives of those involved. Examples are included at the end of this appendix.

Developing and evaluating a policy on mandating mitigation of natural hazards is a difficult process. After determining that a sufficient risk exists and that effective mitigation alternatives are possible, knowing whether mitigation is economically feasible is useful in selecting a mitigation plan or strategy. If a public decision is being made, economic feasibility takes on a special definition that differs from economically feasible decisions made in the private sector.

The economic question in the public sector is usually complicated because it involves estimating all of the economic benefits and costs, regardless of to whom they shall accrue. Economic benefits and costs are defined as true changes in economic efficiency. Economists have developed methods to evaluate the economic feasibility of public decisions. One such benefit-cost analysis procedure was prepared for and published by the Federal Emergency Management Agency (FEMA). These models were developed in conjunction with industry economists, engineers and public officials. They are generally accepted when making decisions regarding mitigating natural hazards.

The selection of mitigation projects to be funded can be made using three criteria: 1) maximum present value, 2) benefit-cost ratio, and 3) internal rate of return. Information on using the first two criteria result from a properly structured benefit-cost analysis.

The Maximum Present Value Criterion

The maximum present value criterion states that the optimal investment strategy is to select the set of projects that maximizes present value of future expenditures subject to the available budget. In benefit-cost analysis, those projects with the greatest benefits minus costs calculates this value. All projects or public investment alternatives must be evaluated simultaneously in this procedure.

Benefit/Cost Ratio

Selecting projects for public investment using the benefit-cost ratio criteria is similar to the maximum present value criterion if unlimited funds are available. The set of projects where benefits exceed costs would be the same as the projects selected by using the maximizing present value criterion.

Internal Rate of Return

The internal rate of return for a project or set of projects is that rate of discount that yields a present value of zero. This criterion yields the same results as the maximum present value criterion if capital is unlimited. If the supply of capital is limited, optimal project selection is attained by selecting those projects with the highest internal rate of return.

2.0 Benefit-Cost Models

Mandating specific standards to mitigate natural hazards in new construction represents a cost to building/land owners with a substantial portion of the benefits to be realized by the public. The decision to adopt new regulations and standards requires a formal benefit-cost analysis. The FEMA models were developed to provide this analysis.

Mandating the Mitigation of Privately Owned Buildings/Land

The FEMA benefit-cost model was to aid local, county and state planners in determining the economic feasibility of seismic rehabilitation programs. The model estimates the expected net present value of benefits of seismic rehabilitation derived from the following parameters:

- Building damages prevented
- Rental losses avoided
- Relocation expenses avoided
- Personal and proprietors' income losses avoided
- Business inventory damages prevented
- Personal property losses prevented.
- Value of casualties avoided

Procedures were developed to analyze a single building or a building inventory. The model was the result of an extensive two year research and development project. The model was also extensively tested. Nine seismic rehabilitation projects located in different cities throughout the country were analyzed using the single building model. An inventory of 67 Seattle-area buildings was analyzed.

Mandating the Mitigation of Publicly Owned Buildings

It may seem appropriate for public agencies to use full benefit-cost analysis to make decisions regarding rehabilitation, but that is usually not the case. Agencies commonly include only those benefits and costs that the agency is responsible for, and they tend to exclude those that are the responsibilities of other agencies or the private sector. When an agency is directed to perform a full benefit-cost calculation, the full range of each is considered. With this objective in mind, FEMA developed a benefit-cost model for publicly-owned buildings based on the earlier benefit-cost model for privately-owned buildings. The publicly-owned buildings model includes the value of avoiding lost public services.

The model provides a procedure of evaluating lost public services based on the quasi-willingness to pay principal. Data on the cost of service, payroll, and a post earthquake continuity premium are used to estimate the value of lost public services.

Public Cost Sharing of Mitigating Natural Hazards

The type of decision dictates the nature of the decision process. The first case is when the agency is faced with meeting established mitigation or construction standards. In those cases, the appropriate decision is a cost effectiveness analysis. The decision process objective cannot vary; the total benefits of the action, both public and private, have been implicitly embodied into the regulation or standard. The only issue for the agency to address is to select the best alternative to comply with the process goals. The decision process involves determining and estimating the costs to the agency of those alternatives. The alternative with the least cost to the agency is usually selected. The agency must decide whether the benefits of mitigation exceed the costs if no regulation or standard is requiring the

construction. The benefits in this case are the present value of the reduction in expected future property losses and downtime and disruption due to natural hazards and any other collateral benefits that may be gained from the construction, such as improved public services. The costs are the present value of all additional capital, operating and maintenance costs of the proposed project.

3.0 Mitigation Decisions

Investing by the private sector in a mitigation measure may occur on the basis of one or two reasons. First, it may be mandated by a regulation or standard, or second, it may be economically justified on its own merits.

Conforming to a Mitigation Standard

A building owner, be it a private entity or public agency, having to conform to a mandated standard may consider the following options:

- 1. Employ resources to change the standard as it relates to the owner's situation,
- 2. Request cost-sharing from public agencies,
- 3. Dispose of the building either by sale or demolition,
- 4. Change the designated use of the building to change the mitigation compliance requirement, or
- 5. Evaluate the feasible alternatives to meet the standard and initiate the least-cost mitigation scheme.

The optimal decision will be made on the basis of selecting the most "cost effective" alternative that often can also be termed the "least-cost alternative".

<u>Example</u>: California's real estate disclosure laws (Civil Code, Section 1102 *et seq.*) require sellers of real property to disclose known defects and deficiencies in the property, including earthquake weaknesses and hazards, to perspective purchasers. The seller must also disclose whether the property is located in an earthquake fault rupture zone. To comply with this law, you must disclose if your building has any of the design or construction flaws that will increase the risk of seismic damages.

Correcting some of these deficiencies is expensive and time consuming, but their existence can prevent the sale of the building. The law does not require either the seller or the buyer to correct the deficiencies. Therefore, the conditions of the sale regarding the deficiencies and the price of the building can be negotiated between the buyer and the seller. The economic decision to rehabilitate the structure would be based on an accurate estimate of the risk and the effectiveness of the project.

Mitigating Natural Hazards: The Private Decision

A building owner may also decide to reduce the risk of natural hazard based only on economic criteria. This decision is usually based on the results of an investment or capital budgeting analysis.

Owners deciding the economic feasibility of mitigation should consider the reductions in physical damages and financial losses. A partial list of avoided future losses follows:

- 1. Building damages
- 2. Building contents damage
- 3. Business inventory damages
- 4. Rental income losses
- 5. Relocation and disruption expenses
- 6. Proprietor's income losses

First, these can be estimated using observed prices, costs and engineering data. The difficult part is to correctly determine the effectiveness of the mitigation and the resulting reduction in future damages and losses.

Second, the damages and losses should include those that will be borne only by the owner. Also, the salvage value of the investment can be important in determining economic feasibility. Salvage value becomes more important as the owner's time horizon declines.

4.0 The Costs of Mitigation

Mitigation projects have initial investment costs and recurring costs over the period of the investment. The project may also deteriorate (or be subject to destruction) over the relevant time horizon. The expected loss of the investment is approximated by multiplying the annual probability of destroying the investment times the value of the investment. Estimating deterioration can be captured by using normal depreciation schedules.

The following example benefit-cost analyses have been taken from *How to Determine Cost-Effectiveness of Hazard Mitigation Projects*, Interim Edition, December 1996, Mitigation Directorate, Federal Emergency Management Agency, Washington, DC.

The following example, *Table 11. Limited Data Module for Riverine Flooding*, shows an example benefit-cost analysis for a road culvert replacement project.

The following example, *Table 13. Full Data Module Printout for Riverine Flooding*, is for a flood hazard mitigation project to acquire and demolish structures subject to repetitive flood losses.

The following example, *Table 19. The Full Data Module Printout—Structural*, is for an earthquake hazard mitigation project to improve the resistance of an existing building's structural element.

Recognizing that some projects nearly always are cost-effective, the Federal Emergency Management Agency (FEMA) issued guidance to eliminate the need for conducting benefit-cost analyses for the types of mitigation projects listed.

Appendix D. 2003 Salem Natural Hazards Mitigation Plan Action Items

This appendix includes all the action items found in the 2003 Salem Natural Hazard Mitigation Plan and describes their level of completion, or action item status. The 2003 Mitigation Plan included 17 action items at various levels of completion, and many of the action items had several smaller actions included in their framework. To organize the different action item components for the 2008 update process, the Partnership incorporated the 17 action items into action item worksheets. The "Proposed Action Item" field includes the main title of the action item, and those actions with smaller action items in their framework are included in the "Ideas for Implementation" field. The action item status field details the level of completion for each action item, and if included in the form, for each idea for implementation. Also, since the worksheets were developed separately from the original 17 action items, there are blank fields in most of the forms.

For every mitigation plan five-year plan update, FEMA requires that communities identify which action items have been completed or not completed. If not completed, communities must explain why and whether or not the action will be continued, deferred, or deleted. The Salem Natural Hazards Mitigation Committee, together with the Partnership, reviewed each 2003 action item and discussed their status. For those actions that have not been completed, the Committee discussed whether the action should be kept, deferred, or deleted. The final status of each action item is included in the "Action Item Status" field of the worksheet, and those that are being kept include a reference to a new action in Section 5 of the Salem Natural Hazards Mitigation Plan.

Proposed Action Ite	em:			Alignment with Plan Goals
Accept the Natural H	lazards Mit	igation Plan.		
Rationale for Propo	osed Action	Item:		
Ideas for Implemen	tation:			
Lead Agency:		Salem City C	Council	
Internal Partners:			Exter	nal Partners:
City of Salem depar	rtments			
Timeline:			If ava	ilable, estimated cost:
<u>Short Term (</u> 0-2 years)	Long Term (2-	4 or more years)		
Summer of 2002				
Form Submitted by	Form Submitted by: Technical Advisory Committee			nittee
Action Item Status:	l			
Completed and addressed in the 2008 NHMP Plan Update.				
Salem City Council adopted the plan in 2002.				
See PLAN IMPLEMENTATION ACTION #1 in 2008 NHMP Plan Update.				

CITY POLICY ACTION #1

CITY POLICY ACTION #2

Proposed Action Item:		Alignment with Plan Goals	
Designate a standing Mitigation Coo	ordinating		
Committee (MCC).			
Rationale for Proposed Action Ite	m:		
Ideas for Implementation:			
 The MCC will be a technical committee made up of staff from city departments, other agencies and private sector experts (such as businesses, academia, and other private and non-profit interests). 			
The Committee should be charged w	vith:		
2) Coordinating the implement	ation of this Natu	Iral Hazard Mitigation Plan,	
3) Monitoring progress in impl	lementing the Pla	n,	
4) Setting priorities among cor 5) Working with each departm	npeting action ite	ms,	
action items as prioritized b	v the Committee	phene of a work plan and budgeting of each of the	
6) Coordinating the Plan and it	s implementation	with relevant city county and state plans and city	
state and federal regulations	state and federal regulations		
7) Submitting an annual progra	7) Submitting an annual progress report to the Environmental Commission Planning Commission		
and City Council,			
8) Recommending changes needed to this Plan as community needs evolve, and Incorporating hazard			
assessment/mitigation goals into all City activities.			
Lead Agency: Community Development Department in partnership with other			
stakeholders represented on the Technical Advisory Committee.		ented on the Technical Advisory Committee.	
Internal Partners:	Exte	rnal Partners:	
Timeline:	If av	ailable, estimated cost:	
<u>Short Term (0-2 years)</u> <u>Long Term (2-4 or n</u>	nore years)		
Winter of 2003			
Form Submitted by: Technic	al Advisory Com	mittee	
Action Item Status:			
Completed Action Item, new committee established in the 2008 NHMP Plan Update.			
Since the adoption of the NHMP, the Mitigation Coordinating Committee has met twice; September 9, 2005 and April 4, 2006. Responsibilities of Mitigation Coordinating Committee are ongoing.			
Changes to the Mitigation Coordinating Committee members and responsibilities are included in the 2008 NHMP Plan Update. See section 1.5 PLAN IMPLEMENTATION and PLAN			

IMPLEMENTATION ACTION #3.

Proposed Action Item:			Alignment with Plan Goals:
Meet with the Capitol Planning Commission.			
Rationale for Proposed Action	Item:		
Determine what roles the Capitol Planning Commission plays in mitigating natural hazards, especially for State of Oregon properties or others in Salem for which it has jurisdiction.			
Ideas for Implementation:			
Lead Agency: Community Development Department			ment
Internal Partners: External Partners:		:	
Timeline:		If available, estimation	ated cost:
Short Term (0-2 years)Long Term (2Within 6 months of accepting plan	-4 or more years)		
Form Submitted by: Tecl	nnical Advisory	v Committee	
Action Item Status:			
Not completed and addressed in the 2008 NHMP Plan Update.			
Board.			
See MULTI-HAZARD ACTIO	ON # 1 in 2008	NHMP Plan Updat	te.

Proposed Action Item:			Alignment with Plan Goals:
Update "All-Hazards" Ranking.			
Rationale for Proposed A	ction Item:		
Better identify the primary	Better identify the primary hazards of concern to the citizens of the Salem area		
Ideas for Implementation	:		
 Obtain additional input from stakeholders (cities, counties, citizens, businesses, and others) on the "All-Hazards" ranking (see Section 4.3). Input can be solicited through ongoing meetings of the Natural Hazards Mitigation Coordinating Committee, the Salem Futures process, and other venues. For comparison purposes, consider conducting a hazard ranking issue using the "Hazard Analysis" system used by the Office of Emergency Management (Oregon State Police) in conjunction with Oregon counties (see Section 6.0 of the NHMP for a reference). 			
Lead Agency:	Lead Agency: Community Development Department and Fire Department		ment and Fire Department
Internal Partners:		External Partners	:
Timeline:		If available, estimation	ated cost:
Short Term (0-2 years) Long Term (2-4 or more years) TBD			
Form Submitted by: Technical Advisory Committee			
Action Item Status:			
Not completed and addressed in the 2008 NHMP Plan Update. 1) Not completed. The "All-Hazards" Ranking was updated by the Salem Natural Hazards			

1) Not completed. The "All-Hazards" Ranking was updated by the Salem Natural Hazards Mitigation Committee in the 2008 NHMP Update process. See 1.4 PLANNING PROCESS SUMMARIZED and Section 4.0 ALL-HAZARDS RISK ASSESSMENT of the 2008 NHMP Update.

2) Not completed. In the future, the "All-Hazards" Ranking will be updated annually by the Salem Natural Hazards Mitigation Committee. See Section 1.6 PLAN MAINTENANCE and the PLAN IMPLEMENTATION ACTION #3.

٦

Proposed Action Item:	Alignment with Plan Goals:			
Refine Existing Hazards Mapping Consider additional research and mapping to be and property at risk.	etter define hazards			
Rationale for Proposed Action Item:				
Any mapping of critical facilities and infrastructure should give due consideration to potential security ramifications. Consider aggregating data as appropriate to protect critical facilities and/or site-specific property data.				
Ideas for Implementation:				
 Include mapping that addresses pathways for volcanic mudflows, areas impacted by a catastrophic failure of the Detroit Dam and other dams or large above-ground reservoirs, areas of landslide risk adjacent to the North Santiam River (upstream of the Geren Island water intake structures), and urban-wildland interface high fire risk areas. Pertinent to hazardous materials concerns, develop a City-wide map of facilities manufacturing, storing, using, transporting, or otherwise handling significant amounts of hazardous materials as a basis for requiring or encouraging owners to prevent spills and other releases. Rank the facilities in terms of the risk of impact following a natural disaster and develop several disaster response scenarios. Incorporate this inventory and ranking into a database for plotting on City of Salem computer generated maps. Incorporate the response scenarios into the current Fire Department training program. OEM's checklist for local mitigation plans includes the need to estimate the type and number of structures within the community at risk for each hazard type, including residences, businesses, critical facilities (hospitals, fire stations, and storage sites for hazardous materials), and infrastructure (e.g., roads and utilities). There also needs to be a map of repetitive flood loss property demage) and discussion of 				
Lead Agency: Community Development Dep	partment and Fire Department			
Internal Partners:	External Partners:			
Timolino	If available estimated east.			
Short Term (0-2 years) Long Term (2-4 or more years)	ii avanable, estimated cost:			
TBD				
Form Submitted by: Technical Advisory Committee				
Action Item Status:				
Partially completed and portions addressed in the 2008 NHMP Plan Update 1) Partially completed See MIII TLHAZARD ACTION #2 and I ANDSI IDF ACTION #1				
2) Completed through training and enforcement of regulations surrounding the hazardous materials				
storage and use provisions of the Fire Code. See SECTION 2.1 PRINCIPAL NATURAL HAZARD				
DESCRIPTIONS and HAZARDOUS MATERIALS ACTION #1.				
3) Partially completed. See Maps in SECTION 2.1 PRINCIPAL NATURAL HAZARD				
DESUKIPTIONS AND MULTI-HAZAKD ACTION #2 and #10.				

Proposed Action Item:	Alignment with Plan Goals:		
Develop mitigation plans for facilities at risk & emergency preparedness/response.			
Rationale for Proposed Action Item:			
 Critical facilities play an important role in emergency response and preparedness and should be structurally retrofitted and located in non-hazardous areas to prevent damage from natural hazard events. The Disaster Mitigation Act of 2000 requires communities to identify mitigation actions that address existing critical facilities and infrastructure [201.6(c)(3)(ii)]. Developing mitigation actions for critical facilities will protect them from future natural hazard events. 			
Ideas for Implementation:			
 The City should continue to contact each of the identified facilities to: Develop a list of contacts and telephone numbers, determine if they have their own emergency response plans, determine any special coordination that will be needed during a disaster, and reminder contacts should be made and the list should be updated at least annually The City should develop an electronic map-database containing the Fire Department's list of critical and other facilities at risk so the data is readily accessible to all City and other personnel responding to a disaster. This information should be combined with mapped hazards. Each of these facilities should be evaluated for its' survivability following a single and compound disaster. The City should consolidate and refine existing lists of facilities subject to a single or compound natural hazard event into a single electronic map-database so that the data is readily accessible to all City and other personnel responding to a disaster. This information should be evaluated for its' survivability following a single or compound natural hazard event into a single electronic map-database so that the data is readily accessible to all City and other personnel responding to a disaster. This information should be combined with mapped hazards. Each critical facility should be evaluated for its' survivability following a single and compound disaster. Public Works department notes that completing this action would be a multi-year work item subject to funding in the annual budget. 			
Lead Agency: Fire Department and Public Work	as Department		
Internal Partners: External Partners	:		
Timeline: If available, estimation of the statement of th	ated cost:		
Short Term (0-2 years) Long Term (2-4 or more years)			
TBD Technical Advisory Committee			
Action Item Status:			
 Partially completed and portions addressed in the 2008 NHMP Plan Update. The City of Salem received a Pre-Disaster Mitigation Grant for Fire Station Seismic Retrofits in 2007 totaling \$1.03 million. 1) Partially completed. MULTI-HAZARD ACTION #3. 2) Not completed. See MULTI-HAZARD ACTION #4. 3) Partially completed. MULTI-HAZARD ACTION #4. Portions not completed due to data and resource limitations. 			

Proposed Action Item:		Alignment with Plan Goals:		
Public Outreach				
Inform the public about the benefits of hazar	d mitigation			
Rationale for Proposed Action Item:	Rationale for Proposed Action Item:			
• Informing the public about hazard mitigation will help enable the public to implement mitigation measures and protect lives.				
Ideas for Implementation:				
 Use city programs providing loans for housing rehabilitation (e.g., Homeowner Rehabilitation Loan Program) as a means to provide public outreach on appropriate hazard protection and retrofitting measures. The City could support this by, for example, holding seminars and providing 				
2) Present the Natural Hazard Mitigation	on Plan at neighborhoo	d association meetings.		
3) Develop an updateable database of r	ames, phone numbers,	etc. of all stakeholders (cities,		
counties, citizens, businesses, and others) involved in planning for and responding to natural disasters.				
4) Expand publicity on the availability of hazard data for any address. Develop generic handouts that explain the data, its shortcomings, and how to obtain more accurate data on any particular site. The handouts should also describe appropriate protection and retrofitting measures and where more information on them can be obtained.				
5) Inventory properties located within high hazard areas and inform property owners of the limited ability to obtain natural disaster insurance.				
6) The City should work with and encourage local real estate trade associations to prepare a handout advising property buyers about the variety of possible hazards and where to get locally available information about a property's exposure to these hazards.				
7) Consider revising the Salem Revised Code (SRC) to include the need for owner disclosure and recording of known natural hazards in a fashion similar to that required in the landslide ordinance.				
8) Include a copy of the NHMP and any future updates as an additional Annex to the City's				
Emergency Management Plan (EMP).				
Lead Agency: Communit	y Development Depart	ment and Fire Department		
Internal Partners:	External Partners	:		
Timeline:	If available, estim	ated cost:		
<u>Short Term (0-2 years)</u> <u>Long Term (2-4 or more years)</u>				
TBD				
Form Submitted by: Technical Advisory Committee				

MITIGATION PLAN IMPLEMENTATION ACTION #7 (CON'T.)

Action Item Status:

Partially completed and portions are addressed in the 2008 NHMP Update.

1) Not completed due to resource limitations. See MULTI-HAZARD ACTION #5.

2) Completed and ongoing. See section 1.6.1 CONTINUED PUBLIC INVOLVEMENT.

3) Completed and ongoing. See 1.6.1 CONTINUED PUBLIC INVOLVEMENT and MULTI-HAZARD ACTION #5.

4) Completed and ongoing. See 1.6.1 CONTINUED PUBLIC INVOLVEMENT and MULTI-HAZARD ACTION #5.

5) Completed for flood hazard and ongoing for other hazards. See 1.6.1 CONTINUED PUBLIC INVOLVEMENT and MULTI-HAZARD ACTION #5. Also, city sends out annual letter to homeowners in floodplain.

6) Not completed due to resource limitations. See MULTI-HAZARD ACTION #5

7) Not completed. The Salem Natural Hazards Mitigation Committee decided that these items do not fit into the scope of the NHMP.

8) Not completed. See MULTI-HAZARD ACTION #7.

fill		
ed by		
•		
ysis		
ers		
rain		
;		
Action Item Status:		

1) Completed and further actions addressed in 2008 NHMP Update. See FLOOD ACTION #1.

2) Partially completed and in progress. Coordinate with the National Flood Insurance Program's

(NFIP) Community Rating System (CRS). See FLOOD ACTION #5.

3) Completed and ongoing. See LANDSLIDE ACTION #2.

Proposed Action It	em:			Alignment with Plan Goals:
Disaster Response - Emergency Manage	the City show ment Plan (E	uld continue to EMP).	o update its	
Rationale for Prop	osed Action	Item:		
• Disaster response is an important component to natural hazards planning that can save lives and property during a natural disaster. Coordinating disaster response efforts with the mitigation plan will ensure that the plan remains relevant to the larger community.				
Ideas for Implement	ntation:			
 The revised plan should include current resource lists and Emergency Operations Plan (EOP) annexes for each of the hazards discussed in this report. It should utilize the latest GIS mapping available and include a flood stage forecast map that relates the potential inundation areas and damage to specific flood forecasts. A list of appropriate response steps should be included for each flood stage level. Perhaps the EOP should also include the potential for hazardous materials spills. Additional guidelines can be found in FEMA's Community Rating System publications. The updated EOP should include a post-disaster recovery and mitigation annex/appendix with procedures that encourage property owners to incorporate retrofitting and mitigation measures and identifies where outside financial assistance would help. Guidance for this effort is available through FEMA and the Oregon State Police, Office of Emergency Management. The City should maintain liaison with the Oregon State Military Department, Office of Emergency Management with the intent of keeping informed about federal and state laws and regulations governing all aspects of post-disaster recovery and mitigation. 				
Lead Agency: Fire Departm		nent, Community Development Department, & Public		
Internal Partners: External Par		External Partners	:	
Timeline:		If available, estimation	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years) TPD				
Form Submitted by	v: Tech	nical Advisory	v Committee	
Action Item Status:				
Partially completed and portions are deferred to the 2008 NHMP Update.				

1) Completed.

2) Completed.

3) Not completed due to resource limitations. See MULTI-HAZARD ACTION # 7.

4) Completed and ongoing. See MULTI-HAZARD ACTION # 8.

$\mathbf{M} = \mathbf{M} + $				
Proposed Action Item:		Alignment with Plan Goals:		
Development Planning/Design/Permitting/Ma	intenance			
Paview the subdivision zoning and other ordi	nances and permit			
Review the subdivision, zoning and other ordi	if and and model			
review procedures to determine what changes,	in any, are needed			
to reduce or prevent losses from future natural	hazards events.			
Rationale for Proposed Action Item:				
Ideas for Implementation:				
1) Future development should be moved	away from identified	/mapped high hazard areas and/or		
transferring development rights from	high hazard areas to s	afer areas, especially in those areas		
where the risk to people and property	cannot be mitigated.			
2) Encourage landowners in the UGB to	provide set-asides for	r open space or less intensive		
development.				
3) Voluntary (with incentives) seismic up	pgrades of critical or	otherwise essential facilities located in		
high seismic risk areas.				
4) Evaluate a range of mitigation measur	es (e.g., less flammab	ble building materials, fuel breaks, etc.)		
to reduce urban-wildland interface fire	e risk.	-		
5) Broader use of all natural hazard over	lays (including a com	posite-hazard map) to better guide the		
planning and permit review process.				
6) Enhanced use of a watershed approach	n for new developmen	nt that includes contributions from		
upstream development and impact on,	for example, Mill Cr	eek as it passes through Salem.		
Lead Agency: Community Development Department (Planning and Building & Safety				
Divisions). Fire Department, and Public Works Department				
Internal Partners:	External Partners	:		
Timeline:	If available, estim	ated cost:		
<u>Short Term (0-2 years)</u> <u>Long Term (</u> 2-4 or more years)				
TBD				
Form Submitted by: Technical Advisor	y Committee			
Action Item Status:				
Partially completed and portions are deferred to the 2008 NHMP Undate.				
1) Not completed and will be addressed when the Salem Zoning Code is updated. See MULTI-				
HAZARD ACTION # 9.				
2) Completed and ongoing. See MULTI-HAZARD ACTION # 9 and FLOOD ACTION #2.				
3) Not completed and will be deleted. The Salem Natural Hazards Mitigation Committee decided				
that this item does not fit into the scope of the NHMP See EARTHOUAKE ACTION #1				
4) Completed and will be deleted This item is done by the Fire Department See WII DFIRF				
ACTION #1.				
5) Completed and ongoing. See MULTI-HAZARD ACTION # 10.				
6) Completed and will be deleted. This item is addressed in the Stormwater Master Plan. See				
FLOOD ACTION #3.				

Page A22

Proposed Action Item:			Alignment with Plan Goals:
Capital Improvement Programs (CIPs):			
Build on existing hazard mitigation efforts with the goal of improving the planning/budgeting process such that CIPs continue to contribute to hazard mitigation.			
T.J T			
Ideas for implementation	1:		
Possible measures to evaluate include securing hazardous areas for low risk uses: 1. Realigning or replacing roads and utilities when feasible in the course of regularly schedureplacement 2. Strengthening or replacing unsafe public structures (especially facilities critical to disaster post-disaster planning/response) 3. Prescribing standards for the design and construction of new public facilities in high haza 4. Suggested projects include those related to flood damage reduction, seismic upgrades of 'and 'non-critical' but essential public facilities, and retrofitting/relocation of structures in landslide prone areas. Lead Agency: Public Works Department, Fire Department, Police Department Community Development Department Internal Partners: External Partners:			low risk uses: the course of regularly scheduled ally facilities critical to disaster and w public facilities in high hazard areas. eduction, seismic upgrades of "critical" tting/relocation of structures in Department, Police Department, and ment :
· · ·			
Timeline:		If available, estimation	ated cost:
Short Term (0-2 years) Long	<u>term (</u> 2-4 or more years)	-	
Form Submitted by:	Technical Advisor	y Committee	
Action Item Status:			
Partially completed and portions are deferred to the 2008 NHMP Update.			
 Completed for some areas and is ongoing. See MULTI-HAZARD ACTION #1. Completed for some buildings including Fire Station, Civic Center, Emergency Operations Building and is ongoing. See MULTI-HAZARD ACTION #11. 			

3) Completed for some sites and is ongoing. See MULTI-HAZARD ACTION #12.

4) Completed for some areas and is ongoing. See FLOOD ACTION #1, EARTHQUAKE ACTION

#2, and LANDSLIDE ACTION #3.

Proposed Action It	em:			Alignment with Plan Goals:
Implement recommendations presented in the Stormwater Master Plan (SMP)				
Rationale for Proposed Action Item:				
These recommendations relate to review of policy and over 200 project-specific measures to better manage stormwater quantity/quality and flooding.				
Ideas for Implement	ntation:			
Lead Agency:	Lead Agency: Public Works Department			
Internal Partners: External Partners		External Partners	Partners:	
Timeline:		If available, estim	ated cost:	
<u>Short Term (0-2 years)</u> <u>Long Term (2-4 or more years)</u>				
TBD <u>5 years</u>				
Form Submitted by: Technical Advisory Committee				
Action Item Status:				
Completed and ong	Completed and ongoing. See FLOOD ACTION #3 in the 2008 NHMP Update.			
MITIGATION PLAN IMPLEMENTATION ACTION #13

Proposed Action It	em:				Alignment with Plan Goals:
Develop effective farming and forestry practices to reduce the impact of natural hazards such as flooding.					
Rationale for Prope	osed Ac	ction	Item:		
Ideas for Implement	ntation:	:			
Contact the Marion and Polk Soil and Water Conservation Districts to determine the potential for increased flooding due to farming and forestry practices in the various smaller watersheds. If there is a concern, encourage appropriate federal, state and regional agencies to help initiate safer agricultural and forestry practices.					
Lead Agency:			Public Work	s Department	
Internal Partners:				External Partners	:
Timeline:				If available, estim	ated cost:
Short Term (0-2 years)	Long Te	<u>erm (</u> 2-4	4 or more years)		
IRD					
Form Submitted by	Form Submitted by: Technical Adviso			y Committee	
Action Item Status	Action Item Status:				
Not completed and will be deleted. The Salem Natural Hazards Mitigation Committee decided that the action item was outside of the City of Salem and therefore outside of the scope of the NHMP.					

MITIGATION PLAN IMPLEMENTATION ACTION # 14

Proposed Action Item:			Alignment with Plan Goals:			
Increase Flood "Freeboard" to r	educe the impa	act of flooding.				
Rationale for Proposed Action	Item:					
Ideas for Implementation:						
If the current regulated flood elevations are too low, as determined by the City of Salem, consider requiring a higher protection level, rather than waiting for a possible new flood study. Consider a freeboard higher than one foot above the 100-year elevation in areas where past floods (including the 1996 floods) have been higher than the 100-year elevation. Marion County increased the freeboard to 2 feet. Check with the State of Oregon (Department of Land Conservation & Development) on this standard.						
Internal Doutnonse		Extornal Dantnong				
		External rarmers	•			
Timeline:		If available, estimation	ated cost:			
Short Term (0-2 years) Long Term (2 TBD Image: Constraint of the state of the	-4 or more years)					
Form Submitted by: Tecl	nnical Advisor	y Committee				
Action Itom Status:		-				
Not completed due to recover	limitationa	See ELOOD ACTIC	NI #4 in the 2008 NHIMD Lindete			
not completed due to resource	Not completed due to resource limitations. See FLOOD ACTION #4 in the 2008 NHMP Opdate.					

Proposed Action Item:			Alignment with Plan Goals:	
Inventory flood damage to damaged by flooding event	assess how many st s and which structu	ructures have been res are vulnerable.		
Rationale for Proposed A	ction Item:			
Ideas for Implementation	:			
Flood Mitigation Assistance Grant Programs, to determine the possibility of receiving funds for such an inventory. OEM has a general inventory. If funding is available, consider contacting property owners in order to develop an inventory with a focus on those areas subject to repetitive losses (i.e., two or more losses). If there are no funding sources immediately available, the City should consider maintaining contact with the owners of these properties and contacting them after the next flood or other disaster affects the same area. It is likely that new sources of funding would be available then.				
Lead Agency:	Public Work	s Department		
Internal Partners:		External Partners	:	
Timeline:		If available, estim	ated cost:	
Short Term (0-2 years) Long T	erm (2-4 or more years)			
TBD				
Form Submitted by:	Technical Advisory	y Committee		
Action Item Status:				
Partially completed and p	ortions are deferre	ed to the 2008 NHM	IP Update.	
Inventory of structures within the floodplain has been created. Three repetitive flood loss properties				

have been identified. Mitigation actions need to be identified. See FLOOD ACTION #6.

Form Submitted by:

MITIGATION PLAN IMPLEMENTATION ACTION #16

Proposed Action Item:				Alignment with Plan Goals:	
Maintain plan risk assessment.					
Rationale for Proposed Action Item:					
Although many features of a risk assessment have been addressed in preparation of this plan or are included in other Action items presented in this section of the NHMP, new (as of February 2001) draft OEM criteria for the risk assessment component of local plans are presented below (these draft criteria should be reviewed against OEM's Evaluation Criteria Checklist for Local Natural Hazards Mitigation Plan Review, January 2002). Many of these issues have been fully addressed in preparation of this NHMP. Other issues require additional work and should be evaluated and prioritized for implementation by the standing Mitigation Coordinating Committee, taking into consideration competing action items and budgetary considerations.					
Ideas for Implement	tation:				
 An estimate of type, including for hazardous of Salem are buildings in S and infrastruct presented in O mapping devo A map and di these propertit describe pote activities desi A summary of losses for eact various hazar Study conside can be found this action wo If not already mitigation. O 	of the type a ng residence s materials), jointly parti Salem. City cture. Action ORS 455 (se eloped. iscussion of ies. Action p ential activiti igned to red of potential i ch hazard typ rd or combin ered this from in the final ould be a may implement one method p	nd number of s, businesses, and infrastruc cipating in a g and County di n needed - exp ee Section 7.3 repetitive floo needed – Crea tes to be condu uce the number mpacts on res pe. Action need hazard sce m the Mill Cr February 2002 alti-year work ed, audit the e	structures within the critical facilities (hose cture (e.g., roads and grant to generate an in ivisions have listings band the inventory, ac of this NHMP) and a od loss properties and te map, conduct asse ucted to ensure comp er of NFIP targeted re- idents and the econo eded - Consider prepa- marios. The COE Sec eek flooding perspec 2 report. Public Worl item subject to fund ffectiveness of existi FEMA documents is	 community at risk for each hazard spitals, fire stations, and storage sites utilities). Marion County and the City nventory of unreinforced masonry of critical and other essential facilities dd facilities that meet the definitions add information to the existing d potential mitigation activities for ssment of mitigation activities, and, bliance with the NFIP including epetitive loss properties. my and an estimation of potential aring an estimate of losses under ction 205 Flood Control Feasibility tive and information used in that study ks Department notes that completing ing in the annual budget. ng code as they relate to hazard the Building Code Effectiveness 	
Lead Agency:		Public Works	s Department (for flo	ooding) and Community Development	
	Department			(Building & Safety Division).	
Internal Partners:			External Partners	:	
Timeline:			If available, estimation	ated cost:	
Short Term (0-2 years) Long Term (2-4 or more years) TBD					

Technical Advisory Committee

MITIGATION PLAN IMPLEMENTATION ACTION # 16 (CON'T.)

Action Item Status:

Partially completed and portions are deferred to the 2008 NHMP Update.

1) Partially completed. Critical facilities have been inventoried, but un-reinforced masonry buildings have not been identified. See MULTI-HAZARD ACTION #2 and EARTHQUAKE ACTION #1.

2) Partially completed. Due to restrictions on the dissemination of information about repetitive flood loss properties, a map will not be completed. Mitigation activities need to be identified. See FLOOD ACTION #5, "Ideas for Implementation."

3) Partially completed due to lack of resources. See MULTI-HAZARD ACTION #2

4) Not completed. The Salem Natural Hazards Mitigation Committee decided they do not have the power to change the building code and see this as outside of the scope of the NHMP.

MITIGATION PLAN IMPLEMENTATION ACTION #17

Proposed Action Item:			Alignment with Plan Goals:		
Include plan maintenance and Salem Natural Hazards Mitig					
Rationale for Proposed Act	tion Item:				
Reducing future losses from natural hazards is a long-term and complex process. Mitigation begins as early as possible so hazards are avoided, future vulnerability is not increased beyond acceptable risk levels, and feasible corrective actions are taken to reduce existing community vulnerability.					
Ideas for Implementation:					
 Establishes the method and schedule of monitoring, evaluating, and updating the mitigation plan within a three-year cycle, Identifies how the community will maintain public participation in the planning process, and Identifies a process for formal adoption of any plan revisions by the City of Salem. It is suggested that the Process outline and identify methods for obtaining funds to maximize benefits to all citizens within the community. However, this requirement is most important for areas that are State-designated as a "small and impoverished community" (note: Salem does not qualify as such a community). 					
Internal Doutnesse		External Destroya			
	Internal Partners: External Partners:				
Timeline:		If available, estimation	ated cost:		
Short Term (0-2 years) Long Term TBD	<u>m (</u> 2-4 or more years)				
Form Submitted by:	Technical Advisory	y Committee			
Action Item Status:					
Completed and deferred with adjustments in the 2008 NHMP Update.					

See section 1.5 PLAN IMPLEMENTATION and PLAN IMPLEMENTATION ACTION #3.

Appendix E. 2008 Plan Update Memo

This memo describes the changes made to the 2003 Salem Natural Hazards Mitigation Plan (NHMP) for the 2008 update of the Salem Natural Hazard Mitigation Plan. It serves as a record of major changes made for each NHMP section.



Memo

To:Federal Emergency Management Agency (FEMA)From:Oregon Partnership for Disaster ResilienceDate:June 19, 2008Re:List of changes to the 2003 Salem NHMP for the 2008 Plan Update

Purpose

This memo describes the changes made to the 2003 Salem Natural Hazards Mitigation Plan (NHMP) for the 2008 update of the Salem Natural Hazard Mitigation Plan. It serves as a record of major changes made for each NHMP section.

Project Background

In February 2008, the City of Salem contracted with the Oregon Partnership for Disaster Resilience (the Partnership) to update the 2003 Salem Natural Hazards Mitigation Plan. The Disaster Mitigation Act of 2000 requires communities to update their mitigation plans every five years to remain eligible for Pre-Disaster Mitigation (PDM) program funding, Flood Mitigation Assistance (FMA) program funding, and Hazard Grant Mitigation Program (HGMP) funding. The Partnership met with members of the Salem Natural Hazards Mitigation Committee (the Committee) to update the City's risk assessment, discuss the plan goals and action items in the 2003 NHMP, develop new goals and action items for the 2008 update, and review all changes made for the 2008 update prior to submittal to FEMA.

The Partnership and the Committee made several major changes to the 2003 NHMP. The major changes are documented and summarized in this memo for each section of the mitigation plan

2008 Update Changes by Section

The sections below only discuss *major* changes made to the 2003 Salem Natural Hazard Mitigation Plan for the 2008 plan update. Major changes include replacement or deletions of large portions of text, changing the location of sections, and new additions to the plan. If there are no sections that are addressed, then it can be assumed sections in the 2008 update are similar to like sections in the 2003 NHMP.

Cover Page and Acknowledgments

- 1. The Cover Page for the Salem NHMP has been revised to include 2008 update information and the agencies involved in developing the plan update.
- 2. The Acknowledgments have changed to reflect the participants involved in the 2008 update. The participant list in the 2003 NHMP included representatives of agencies that provided input during the 17 meetings Golder Associates and Robert Olson Associates held to develop the 2003 Salem NHMP. For the 2008 update, members of the original Mitigation Coordinating Committee were invited to participate because they had the most experience with the natural hazards that affected Salem and were involved in previous mitigation planning meetings. The 2008 update built upon information included in the 2003 NHMP, so the majority of the work had been completed and the Committee did not see the need to invite all the original stakeholders for the update process.

Section 1.0 Introduction

Section 1.0 includes an introduction and purpose for the plan, summarizes the process for developing the 2003 NHMP and the 2008 update, identifies City departments responsible for maintaining and implementing the plan, describes procedures for prioritizing projects, and describes procedures for maintaining and reviewing the plan. The major changes in Section 1.0 include the following:

- 1. Section 1.2. "Section 1.2 Hazard Mitigation" was formerly "Section 1.3 Hazard Mitigation" in the 2003 NHMP. Most of Section 1.2 includes new information added by the Partnership and replaces out of date text found in the 2003 NHMP. The new text defines mitigation, gives examples of mitigation strategies, and lists federal programs that communities with mitigation plans are eligible for. These programs include the Pre-Disaster Mitigation (PDM) Program, the Flood Mitigation Assistance (FMA) Program, and the Hazard Mitigation Grant Program (HMGP).
- 2. Section 1.3. "Section 1.3 City of Salem" was formerly "Section 1.5 City of Salem" in the 2003 NHMP. The Committee updated the plans and programs in Section 1.3 for the 2008 update, omitting plans that were no longer relevant and including new plans and programs developed since 2003.
- 3. Section 1.4. "Section 1.4 Planning Process Summarized" was revised to include a summary of the planning process for the 2003 NHMP and the 2008 update. Section 1.4.1 combines sections "1.4.2 Technical Advisory Committee" and "1.4.3 Community Involvement" of the 2003 NHMP, and includes meetings the Mitigation Coordinating Committee held since 2003. "Section 1.4.2 Development of the 2008 Salem NHMP Update" is new information added to the plan describing the 2008 update process.
- 4. Section 1.5. "Section 1.5 Plan Implementation" is a new section added to the 2008 update. 1.5.1 identifies the Plan convener, 1.5.2 identifies Coordinating Body members and their responsibilities, and 1.5.3 Project Prioritization discusses the project prioritization process and summarizes the STAPLE/E approach, which is discussed further in Appendix B. Elements of Section 1.5 were addressed in the 2003 NHMP in Actions 2 and 17, but the Plan did not have a comprehensive section that discussed who was responsible for the Plan and how projects would be prioritized.
- 5. Section 1.6. "Section 1.6 Plan Maintenance" is a new section added to the 2008 update. Section 1.6 discusses who will be responsible for implementing the plan; section 1.6.1 discusses how the public would continue to be included in updating the Salem NHMP; section 1.6.2 discusses the Five-Year Review process. The 2003 NHMP had maintenance information included as Action # 17, but did not clearly delineate how the plan would be updated and what the maintenance process would be. However, some of the information included in Action # 17 has been incorporated into Section 1.6.
- 6. Section 1.7. "Section 1.7 Natural Hazard Mitigation Plan Components" was originally included in the 2003 NHMP, but changed for the update to reflect the 2008 NHMP structure. The major changes include content in Section 4.0 which discussed General Mitigation Measures in the 2003 NHMP, but now includes the Risk Assessment, and Section 5 which now includes new action items developed for the 2008 update.

Section 1.0 Deletions

For the 2008 update, several sections from the 2003 NHMP have been removed because of outdated information or because they are no longer relevant to the 2008 update. The removed sections are the following:

- 1. Quote from former Mayor Swaim located above "Section 1.0 Introduction" deleted.
- 2. "Section 1.2 The Planning Context" found in the 2003 NHMP discussed the Salem Futures Project that was being developed in 2003. However, since Salem Futures is no longer in existence, the Salem Natural Hazard Mitigation Committee recommended that references to the project be removed.

Section 2.0 Natural Hazards

Section 2.0 "Natural Hazards" discusses the prominent natural hazards that affect the State of Oregon and the City of Salem and which hazards have been mapped. This section discusses flood, earthquake, landslide, severe weather, urban-wildland interface fire, volcano, drought, and hazardous materials. For 2008 update, most of the changes involved adding new information relevant to the City of Salem. The changes made to each section include the following:

- 1. **Section 2.1.1 Flood Hazard.** For the 2008 update, additional information was added to this section to discuss how flood hazards affect the City of Salem, information on the Community Rating System, and how many structures, by zone, are located in the floodplain.
- 2. Section 2.1.2 Earthquake Hazard. For the 2008 update, additional information was added to this section to discuss how earthquake hazards affect the City of Salem.
- 3. **Section 2.1.3 Landslide Hazard.** For the 2008 update, information gathered from the March 19, 2008 hazard identification meeting was added to the landslide hazard section.
- 4. Section 2.1.4 Severe Weather. The Severe Weather Hazard Section had recent information relating to severe weather in the City of Salem added to it. The information came from the NOAA Climatic Data Center and information from the March 19, 2008 hazard identification meeting.
- 5. Section 2.1.5 Urban Wildland Interface Fire Hazard. For the 2008 update, information gathered from conversations with Salem Fire Department staff was added to this section.
- 6. **Section 2.1.6 Volcano Eruption Hazard.** Additional information about volcanic eruptions from Mount Hood and Mount Jefferson and their impact on the City of Salem included in this section.
- 7. Section 2.1.7 Drought Hazard. The 2003 NHMP did not include drought. The Salem Natural Hazard Mitigation Committee decided to include the drought hazard in the 2008 update. The Partnership added information about drought as it relates to Oregon and the City of Salem.
- 8. Section 2.1.8 Hazardous Materials. The 2003 NHMP included information on hazardous materials. The 2008 update also included hazardous materials information, but added additional information gathered from the March 19, 2008 hazard identification meeting.
- **9.** Section 2.2 Mapped Hazards. Section 2.2 in the 2003 Salem NHMP discussed mapped hazards and included a "Section 2.2.3 All-Hazards Map" that explained the methodology for developing the "all hazards" map. Only the first part of Section 2.2 was retained for the 2008 update, and Section 2.2.3 was moved to Appendix F Natural Hazard Maps.

Section 2.0 Deletions

The 2003 NHMP included "Section 2.2.1 Hazard Rankings" which described how the committee ranked the natural hazards in the plan. This section was removed from Section 2.0 in the 2008 update, revised, and placed in Section 4.0 as "Section 4.2 Hazard Assessments and Rankings." Section 4.2 now includes definitions for hazard assessments, describes the methodology for developing the assessments, and how the natural hazards are ranked. See Section 4.2 in this memo for more information.

Section 3. Goals and Objectives

Section 3 discusses the goals and objectives for the Salem NHMP. The 2003 Salem NHMP included the following three goals:

- 1) Lives should be protected.
- 2) Existing properties should be protected.
- 3) Future development should be protected.

Each goal also had its own set of objectives.

For the 2008 update, the Salem Natural Hazard Mitigation Committee, together with the Partnership, revised the goals to read like the following:

- 1) Protect lives.
- 2) Protect existing and future development.
- 3) Increase cooperation and coordination among stakeholders.
- 4) Reduce economic loss.
- 5) Protect the natural environment.

(Objectives to be included here once a decision is made)

Section 4.0 All-Hazards Risk Assessment

Section 4.0 describes the purpose of the Risk Assessment, the methodology and definitions for the probability and vulnerability assessments, and includes a Risk Assessment table that describes the extent, the threat, the hazard history, probability and vulnerability assessments, and the rankings for all the hazards addressed in the Salem NHMP. In the 2003 NHMP, the risk assessment information was in Appendix 7.2, but is now part of Section 4.0.

- 1. Section 4.1 Purpose. Section 4.1 Purpose is an addition made in the 2008 update and describes the purpose of the risk assessment and the hazards addressed by the plan.
- 2. Section 4.2 Hazard Assessments and Rankings. Portions of Section 4.2 are taken from the 2003 NHMP "Section 2.2.1 Hazard Rankings." Section 4.2 discusses the methodology used for the 2003 NHMP hazard rankings. Information added for the 2008 update includes definitions of vulnerability and probability assessments, a description of how the Committee decided probability and vulnerability assessments, and how natural hazard rankings for the 2008 update were developed.
- 3. Section 4.3 Risk Assessment Table. The format and much of the information found in the risk assessment table is taken directly from the 2003 NHMP. For the 2008 update, several changes were made to update the table. These changes include:
 - Moving the Risk Assessment Table from Appendix 7.2 in the 2003 NHMP to Section 4.0 in the 2008 update. This makes the Risk Assessment part of the plan, rather than an Appendix to the plan.
 - Updated hazard history using information gathered from the March 19, 2008 Salem Natural Hazards Mitigation Committee meeting. Hazards addressed include: Flood, Landslide,

Earthquake, Volcanic Eruption, Severe Wind and Ice Storms, Urban Wildland Interface Fire, and Hazardous Materials.

- Added a column to the table to address the drought hazard.
- Removal of the hazard assessments and rankings found in the 2003 NHMP and replacing them with new probability and vulnerability assessments for the 2008 update. In the 2003 NHMP, the methodology for obtaining the hazard assessment scores was unclear. The 2003 plan also pre-empted assessments completed for the Polk County and Marion County Natural Hazard Mitigation Plans. For the 2008 update, new probability and vulnerability hazard assessments with a clear methodology that are linked with hazard assessments found in the Polk County and Marion County Natural Hazard Mitigation Plans are included. For more information, see Appendix G Public Process.
- Inclusion of new hazard rankings as described in Section 4.2.

Section 4.0 Deletions

In the 2003 Salem NHMP, Section 4 included a description of Mitigation Measures that communities can take to reduce vulnerability of natural hazards, but most of this information has been incorporated into the plan or removed because new additions to the 2008 update made the information redundant. In its place is the Risk Assessment information for the City of Salem. The Sections found in the Section 4.0 of the 2003 NHMP included the following:

- Section 4.0 introductory paragraph;
- "Section 4.1 General Measures Available to Reduce Future Losses;"
- "Section 4.2 Other Considerations" which included 4.2.1 Cost/Benefit Analysis for Hazard Mitigation, 4.2.2 Concurrent Hazards and Compound Disasters, 4.2.3 Multiple Objective Management;
- "Section 4.3 The City of Salem's Current and Prospective Mitigation Measures."

Section 4.1 of the 2003 NHMP had information that was largely incorporated into the action items of the 2008 update. As a result, information in Section 4.1 was redundant and removed. Section 4.2.1 Cost/Benefit Analysis and Section 4.2.3 Multiple Objective Management were removed because this information is addressed in Section 1.5.3 Project Prioritization Process and in Appendix B STAPLE/E Approach of the 2008 update, Finally, Section 4.2.2 Concurrent Hazards and Compound Disasters had information discussed in Section 2.0 of the 2008 update, and was removed. Section 4.3 was removed because this information was included in Section 5.0 Action Items in the 2008 update.

Section 5.0 Mitigation Action Items

Section 5.0 of the 2008 update defines action items and lists the natural hazards they address, explains how action items are coordinated with Existing Plans and Programs, describes the different categories found in the action item worksheets, includes an action item matrix, and lists 33 action items.

- 1. Section 5.1 Updated Mitigation Action Items. Section 5.1 is a new addition to the 2008 update that defines action items and lists the hazards that the action items address.
- 2. Section 5.1.1 2008 Action Item Update. This is a new addition to the 2008 update and describes how the Partnership and the Salem Natural Hazard Mitigation Committee developed the action items for the 2008 update.
- 3. Section 5.1.2 Coordination with Existing Plans and Programs. This section is a new addition to the 2008 update and describes how the Mitigation Plan is coordinated with existing City plans and programs.

- 4. Section 5.2 Action Item Worksheets. Section 5.2 describes action item worksheets and is a new addition to the 2008 update. To organize the different elements of an action item, the Partnership placed all the new action items into action item worksheets. Section 5.2 describes in detail the different action item categories.
- 5. Action Item Matrix. The action item matrix is a new addition in the 2008 update. The matrix summarizes the major elements of each action item and how they relate to the plan's goals.
- 6. Action Items. The 2008 update includes 33 action items. They are a combination of new action items addressing issues committee members identified in the March 19 hazard identification meeting, and action items from the 2003 NHMP that had not yet been completed. The 2003 actions that are included in the update as current action items have been revised as needed. The 2003 action items can be found in Appendix F.

Appendices

The Appendices include supporting information for the Salem NHMP. The 2003 NHMP numbered the appendices as section 7, however for the 2008 update, the appendices were assigned a letter. With the exception of Appendix 7.2, the same appendices found in the 2003 NHMP remain, with some edits made to them. Additional appendices were also added for the 2008 update.

- 1. Appendix A. Additional Federal and State Government Mitigation Information. This appendix was included in the 2003 NHMP as Appendix 7.1 but had outdated information. The Partnership updated names of agencies that had changed, removed references to the 1996 floods and landslides, and removed outdated rules relating to the Disaster Mitigation Act of 2000 that had been updated over the past five years.
- 2. **Appendix B. STAPLE/E Approach.** Appendix B is an addition to the 2008 update. The 2003 Plan did not include information about the STAPLE/E Approach, which allows committees to quickly evaluate mitigation activities.
- 3. **Appendix C. ORS 455.477**. Appendix C is an excerpt from ORS 455.477 that regulates certain structures vulnerable to earthquakes and tsunamis. It was included in the 2003 NHMP as Appendix 7.3 and remains in the 2008 update with no changes.
- 4. **Appendix D. Benefit-Cost Methodologies.** Appendix D describes different methodologies for conducting cost-benefit analyses to evaluate the cost effectiveness of Proposed Hazard Mitigation Measures. Appendix D was included in the 2003 NHMP as Appendix 7.4, and remains in the 2008 update with no changes.
- 5. Appendix E. 2003 Salem NHMP Action Items. Appendix E includes all of the action items from the 2003 NHMP incorporated into action item worksheets. Formerly in Section 5.0 of the 2003 NHMP, the action items as an appendix are a new addition in the 2008 update. The action item worksheets are also an addition to the 2008 update and they help organize different elements of each action item. Included with the worksheets is a status bar indicating whether the action item has been completed or not completed. For those actions that have not been completed, the status bar indicates whether the action will be deleted or incorporated into the update and why. The actions that are incorporated into the 2008 update are referenced in the status bar and located in Section 5.

- 6. **Appendix F. Natural Hazard Maps.** Appendix F includes maps showing natural hazard impacts in the City of Salem. The 2003 NHMP included maps within the body of the report. However, since the maps are referenced throughout the plan, it made more sense to include them in an appendix. The maps in the 2008 update are similar to the maps in the 2003 NHMP but they include new information, such as urban growth boundary extensions. In addition, Section 2.2.3 "All Hazard Map" found in the 2003 NHMP has been moved to Appendix F because it discusses the methodology for creating the All-Hazard map.
- 7. **Appendix G. Public Process.** Appendix G documents the announcements, agendas, minutes, and sign-in sheets for meetings regarding implementation, maintenance, and updating of the Salem NHMP. Appendix 7.5 of the 2003 NHMP included a public outreach announcement for the plan and was the only record of a meeting in the 2003 NHMP. Appendix G includes the public outreach announcement, as well as the agendas of meetings that have taken place since 2003, and meeting documents from the 2008 Salem NHMP update.

Appendix F. Public Process

This appendix contains announcements, agendas, minutes, and sign-in sheets documenting the public process that has taken place since the inception of the Salem Natural Hazards Mitigation Plan. As noted in section 1.6.1 Continued Public Involvement, the City of Salem is dedicated to involving the public directly in the reshaping and updating of the Salem Natural Hazards Mitigation Plan. Moving forward, the convener will be responsible for documenting the outcome of the semi-annual meetings in this appendix.

Materials from the following meetings are included in this appendix:

- Hazard Mitigation Open House Announcement, December 15, 1999;
- Technical Advisory Committee Meeting, September 9, 2005
- Technical Advisory Committee Meeting, April 4, 2006
- 2008 NHMP Update Kickoff Meeting, February 15, 2008
- 2008 NHMP Update Meeting, #1, March 19, 2008
- Interview with Joe Parrot, April 16, 2008
- 2008 NHMP Update Meeting, #2, April 30, 2008
- 2008 NHMP Update Meeting, #3, May 22, 2008

Hazard Mitigation Open House*

15 December 1999 4:30 p.m. - 7:00 p.m.

Al Loucks Lecture Hall Auditorium

City Library 585 Liberty St. SE Salem, Oregon

AGENDA

Open House----time to browse around and view printed and mapped materials and ask questions of Salem staff and Golder and Associates personnel (consultants for the project).

- **5:30 p.m.** Brief presentation on background, purpose of study, schedule, review of written and mapped material, together with questions, comments, and suggestions.
 - · Introductions Cecilia DeSantis Urbani
 - Purpose of Study and Schedule Curt Schneider
 - Present Hazard Information Working Paper #1 Golder and Associates
 - · Questions and Answers Golder and Associates and Salem staff
- Open House continues
- Adjourn

*Purpose of the Open House is to present: (1) area hazards inventory information and mapping (HIP #1) available to the public; (2) receive feedback as to the inventory information; and (3) to seek any comments citizens may have on goals, objectives and direction the study should follow (within the constraints of the grant).

CECE AGENDA 米巴川市 CHULLE B STRE Daws UNIC NATURAL HAZARDS MITIGATION Workgroup M GOTERRY September 9, 2005 - 1330 - 1430 RSTEVENSON LOCATION: PW Salmon Run 1. Additions to the agenda Stevenson 2. Natural Hazards Mit. Workgroup KEAR (AWARENESS) RISK-- PAPELS OPS AS NOUCESSARY AU - AN FREILITS ENG (Group 3. Make up of Work group 3BD HURS 4. Frequency of meetings NO ZAD THIS HIT Group IST MORN MONTHEN ANNURUY AT UGAST THEN DECEMBER Group 5. Mitigation Steps · YOU GUYS AND GUING TO RELIEW 17 POINTS - KEBP . WIMING THE ATTONS NOODED POR FUNDE IMPREMENTATION · ACTIVITYS NOTED AR PROGRESS IN CAUGH AREAS Group 6, Rank Steps COMMUNITY BIRS - LORD ASSUMPTIONS SHOUS SITUATION CRCZ BUSLICESOS ANZ IMPORT HOUSEN RETIND 7. Adjournment 14-28 8. Next meeting: TBD

Eeb 27 2008 9:53 P. 02

MITT AUTTEN COWN CALK L9X:203-282-8014

--

0

RESTER

AGENDA	2, 3, 0,0,0,0,0
NATURAL HAZARDS MITIGATION Workgrou	10 P. URBANI
4	s, downs ,
April 🗶 2006 - 1000 - 1100	M GUITERBA J
LOCATION: PW 325	B. KOHUMEY 22~
	K. Rais/
Action tome - progress?? Follow-up to the Action items in th	S TONDAHL CALLOD
Action Remains - progress r. Tonow-up to the retient terms in the	rdivisting
Committee (MCC) which will be a technical committee - Designate a standing willigation Cool Committee (MCC) which will be a technical committee made up of staff from city dep gencies. Expanded GRAP - ONE25?	Leizer Trans
Action 3: Capital Planning Commission - Meet with the Capital Planning Commissio oles it plays in mitigating natural hazards, especially for State of Oregon properties of br which it has jurisdiction.	on to determine what DAS or others in Salem
Lead agency: Community Development Department Deadline: Within 6 months after accepting this plan.	
Action 4: Update "All-Hazards" Ranking - Obtain additional input from stakeholder	rs (cities, counties, milling from My Druk
Lead agency: Community Development Department and Fire Department Deadline: to be determined (tbd)	
	ning to better define
Action 5: Refine Existing Hazards Mapping - Consider additional research and map	
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk, — CRINCAL WRLAS WORKE AYER, "C Lead Agency: Community Development Department and Fire Department	
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk, — COLACAL WRLAS WORKE WYER, "C Lead Agency: Community Development Department and Fire Department Deadline: tbd	
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk. — CRINCAL WRLAS WORKE WYEL C Lead Agency: Community Development Department and Fire Department Deadline: tbd	and continue to contact
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk, — OR REAL WRLAS TWO THE AYES, "C Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should reach of the identified facilities to:	d continue to contact
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk, - COLICAT WRLAS TWO AND AYER, -C Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to:	d continue to contact
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk, - CDINCAL WRLASTWORD - AYEL. Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans,	d continue to contact
 Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk OPINATION WRLAS WORKER AYES Community Development Department and Fire Department Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: Develop a list of contacts and telephone numbers, Determine if they have their own emergency response plans, Determine any special coordination that will be needed during a disa 	d continue to contact
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk, - OR The WRLAS TWO FOR A YOU, - C Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans, 3. Determine any special coordination that will be needed during a disa Reminder contacts should be made and the list should be updated at least annually Lead Agency: Fire Department and Public Works Department Deadline: tbd	d continue to contact
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk. — CRITCAL WRLASTWCINE AYE. ~ Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans, 3. Determine any special coordination that will be needed during a disa Reminder contacts should be made and the list should be updated at least annually Lead Agency: Fire Department and Public Works Department Deadline: tbd	d continue to contact ester, and
 Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk OR TOP OF WRLAS TWO AND ANY TOP - ANY TOP -	d continue to contact ester, and
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk ORMAN WELLAS WORKS WAYES, "C Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans, 3. Determine any special coordination that will be needed during a disa Reminder contacts should be made and the list should be updated at least annually Lead Agency: Fire Department and Public Works Department Deadline: tbd Action 7: Public Outreach - Inform the public about the benefits of hazard mitigation include: Lead agency: Community Development Department and Fire Department Deadline: tbd MW-AS FEQUESTED - WMWAG - WMWA	and continue to contact d continue to contact ester, and on. Possible actions (POCIATIN) NEEDED (POCIATIN) NEEDED
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk CREACE AGEN AGEN AGEN, - CREACE Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans, 3. Determine any special coordination that will be needed during a disa Reminder contacts should be made and the list should be updated at least annually Lead Agency: Fire Department and Public Works Department Deadline: tbd Action 7: Public Outreach - Inform the public about the benefits of hazard mitigation include: Lead agency: Community Development Department and Fire Department (CFA Deadline: tbd Action 8: Disaster Awareness/Warning - The City should research and review the fe following in consultation with the National Weather Service and/or other agencies in	and continue to contact aster, and aster, aster, and aster, aster,
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk CRUTCAC URLASTWCATE AYEL Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans, 3. Determine any special coordination that will be needed during a disa Reminder contacts should be made and the list should be updated at least annually Lead Agency: Fire Department and Public Works Department Deadline: tbd Action 7: Public Outreach - Inform the public about the benefits of hazard mitigation include: Lead agency: Community Development Department and Fire Department (CETT Deadline: tbd Action 8: Disaster Awareness/Warning - The City should research and review the fe following in consultation with the National Weather Service and/or other agencies in Lead Agency: Fire Department, Community Development Department Department, & Public W Deadline: tbd	and continue to contact attack aster, and aster, aster, and aster, aster, as
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk CRL (RAC (URLAS MUCTURE AYC) - C Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans, 3. Determine any special coordination that will be needed during a disa Reminder contacts should be made and the list should be updated at least annually Lead Agency: Fire Department and Public Works Department Deadline: tbd Action 7: Public Outreach - Inform the public about the benefits of hazard mitigation include: Lead agency: Community Development Department and Fire Department (CFA Deadline: tbd Action 8: Disaster Awareness/Warning - The City should research and review the fe following in consultation with the National Weather Service and/or other agencies in Lead Agency: Fire Department, Community Development Department Department, & PUDIC W Beadline: tbd Action 9: Disaster Awareness/Warning - The City should research and review the fe following in consultation with the National Weather Service and/or other agencies in Lead Agency: Fire Department, Community Development Department, & Public W Beadline: tbd	and continue to contact aster, and aster, aster, and aster, aster, as
Action 5: Refine Existing Hazards Mapping - Consider additional research and map hazards and property at risk COLMAN WRANG - AYER - CALL Lead Agency: Community Development Department and Fire Department Deadline: tbd Action 6: Facilities at Risk & Emergency Preparedness/Response - The City should each of the identified facilities to: 1. Develop a list of contacts and telephone numbers, 2. Determine if they have their own emergency response plans, 3. Determine any special coordination that will be needed during a disa Reminder contacts should be made and the list should be updated at least annually Lead Agency: Fire Department and Public Works Department Deadline: tbd Action 7: Public Outreach - Inform the public about the benefits of hazard mitigation include: Lead agency: Community Development Department and Fire Department (CFA Deadline: tbd Action 8: Disaster Awareness/Warning - The City should research and review the fe following in consultation with the National Weather Service and/or other agencies in Lead Agency: Fire Department, Community Development Department Department, & Public W Action 9: Disaster Response - The City should continue to update its emergency op Lead agency: Fire Department, Community Development Department, Public W	and continue to contact aster, and aster, aster, and aster, aster, a

Action 10: Development Planning/Design/Permitting/Maintenance - Review the subdivision, zoning and other ordinances, and permit review procedures to determine what changes, if any, are needed to reduce or prevent losses from future natural hazards events. Lead Agency: Community Development Department (Planning and Building & Safety Divisions), Fire ROTEI THE WOSS · FLOOPING - DOWNS Department, and Public Works Department properties Deadline: tbd Action 11: Capital Improvement Plans (CIPs) – Build on existing hazard mitigation efforts with the goal of improving the planning/budgeting process such that CIPs continue to contribute to hazard mitigation. * 507m unit Lead agency: Public Works Department, Fire Department, Police Department, and Community • SEISMIC Development Department · BR40635 Deadline: tbd Action 12: SWMP Recommendations - Implement recommendations presented in the Stormwater Master Program Plan (SWMP). · CiP י הכבר עבר. הניחורי Lead agency: Public Works Department Deadline: tbd · RUER DOM Action 13: Farming and Forestry Practices - Contact the Marion and Polk Soil and Water Conservation Districts to determine the potential for increased flooding due to farming and forestry practices in the various smaller watersheds. - MARION CNIY - LITTLE PUDDING Lead agency: Public Works Department . TMOL ISSUES W) Deadline: tbd " MSO Paul WC Action 14: Increase Flood "Freeboard" - If the current regulated flood elevations are too low, as determined by the City of Salem, consider requiring a higher protection level, rather than waiting for a . FEMA PIRM MAPS · GENERISAND PROTECTOJ possible new flood study. Lead agency: Public Works Department . PLOOD INS. Deadline: tbd Action 15: Inventory Flood Damage - City staff should first check with potential funding sources, including FEMA's Hazard Mitigation and Flood Mitigation Assistance Grant Programs, to determine the possibility of receiving funds for such an inventory. . Ree anon

Lead agency: Public Works Department .SAVEN (NOTTHAL PAR PROPERT (?) Deadline: tbd

Action 16: Risk Assessment - Although many features of a risk assessment have been addressed in preparation of this plan or are included in other Action items presented in this section of the NHMP, new (as of February 2001) draft OEM criteria for the risk assessment component of local plans are presented below (these draft criteria should be reviewed against OEM's Evaluation Criteria Checklist for Local Natural Hazards Mitigation Plan Review, January 2002).

22 O.SIGTUST

Lead agency: Public Works Department (for flooding) and Community Development Department (Building & Safety Division).

Deadline: tbd

Action 17: Plan Maintenance and Implementation - Reducing future losses from natural hazards is a long-term and complex process. Mitigation begins as early as possible so hazards are avoided, future vulnerability is not increased beyond acceptable risk levels, and feasible corrective actions are taken to reduce existing community vulnerability.

Lead agency: Community Development Department Deadline: tbd

7. Adjournment

8. Next meeting: TUNE 06

P. 04



Meeting:Salem Natural Hazard Mitigation KickoffDate:February 15, 2008Time:10 am to 11 30 amLocation:City of Salem

AGENDA

1.	Introductions	(5 minutes)
2.	Overview of Plan Needs	(10 minutes)
3.	 Clarification of City Responsibilities a. Notification and Invitation to MCC members for plan update meetings b. Arranging meeting place c. Provide hazard history events and update maps 	(10 minutes)
4.	 Meeting Representatives (MCC) a. Any additional stakeholders not previously represented that should be at the t b. Date for Kickoff meeting c. History of MCC meetings since plan adoption, has plan been used? 	(10 minutes) able now
5.	Community involvement a. Past community involvement b. Public involvement in update process	(10 minutes)
6.	Vulnerability assessmentsa. Are current vulnerability assessments adequate?b. What was the methodology for determining vulnerability assessments?c. How should vulnerability assessments be changed?	(10 minutes)
7.	Probability assessments d. Inclusion of Marion County probability assessments	(10 minutes)

8. Next Steps



Meeting: Salem NHMP Update Meeting # 1

Date: March 19, 2008

Time: 9 am to 11 am

Location: Salem

AGENDA

1.	. Introductions				
 Project Overview Partnership for Disaster Resilience information (Krista) 5 Year Update Requirements (Krista) OPDR Recommended Revisions (Gregoor) 					
3.	Hazard Identifi a. Preser i. ii. iii. iv. v. v. vi. vi. vii. vii	ication and Vulnerability/Probability Assessment Intation and identification of hazard history and mitigation activities (Nick Flood Landslide Earthquake Volcanic Eruption Severe wind & ice storm Urban-Wildland Interface Fire Hazardous Materials Incidents Drought?	(1 hour 20 min) & Gregoor)		

4. Next Steps: Goals & Action Items Meeting (Krista)

(10 minutes)



Meeting Minutes

Meeting:Salem NHMP Update Meeting # 1Date:March 19, 2008Time:9 am to 11 amLocation:Salem

1. Meeting attendees (Natural Hazards Mitigation Workgroup)

-Roger Stevenson, City of Salem Emergency Management
-Cliff Sabin, Salem Hospital
-Lyle Gembala, Salem Police
-Mike Gotterba, Salem Public Works
-Sara Jondahl, Salem Public Works
-Bruce Bishop, Salem Planning Division
-Ken Boley, City of Salem
-Steve Downs, Salem Public Works
-John Vanderzanden, Marion County Emergency Management

-Krista Dillon, Partnership for Disaster Resilience (PDR)-Nick Kraemer, PDR-A. Gregoor Passchier, PDR

2. Project Overview

-Partnership for Disaster Resilience Overview

Oregon Partnership for Disaster Resilience (OPDR) is an applied research center housed at the University of Oregon's Community Service Center and has been hired to work with the City of Salem to update the Salem Natural Hazards Mitigation Plan (NHMP). Communities are required to update their mitigation plans every five years if they want to remain eligible for Federal Pre-Disaster Mitigation (PDM) Funding and Flood Mitigation Assistance (FMA) program funding. Salem's current version of the mitigation plan is set to expire in May 2008.

-5 Year Update Requirements

There are four components communities must address in their mitigation plan updates:

- Planning process: Describes the process communities use to review and analyze each section of the plan and how the public remained involved during the plan maintenance process over the previous five years.
- Risk Assessment: The risk assessment section addresses and incorporates the following elements:
 - 1. Newly identified hazards that pose a more significant threat than was previously apparent;
 - 2. Occurrences of hazards addressed in the previously approved plan;



- 3. New occurrences of hazard events;
- 4. New hazard-related information or hazard information related to profiling hazards, such Flood Insurance Program maps or studies, HAZUS studies, or reports from other Federal or State agencies that describe location, extent, probability, or previous occurrences of hazards.
- 5. Deficiencies identified in the previously approved plan, and if not resolved, must be addressed in the updated plan accompanied by an explanation of why they remain and an updated schedule to resolve the issue.
- 6. Updated maps consistent with new hazard information.
- 7. Description of any changes, clarifications, or refinements to the overview summary described in the previously approved plan. It shall continue to include, by type of hazard, a general description of the types of structures affected by the hazard.
- 8. In addition, the community should take into account the following when updating its vulnerability assessment:
 - Updates to inventories of existing structures in hazard areas, including structures located in annexed areas.
 - Potential impacts of future land development, including areas that may be annexed in the future.
 - New buildings that house special high-risk populations (i.e., elderly, low-income, disabled).
 - Completed mitigation actions that reduce overall vulnerability.
- Mitigation Strategy: Reaffirms or updates plan goals based on current conditions, including the completion of mitigation initiatives, an updated or new risk assessment, or changes in State priorities.
- Plan Maintenance Process: Provides an analysis of whether the previously approved plan's method and schedule for monitoring, evaluating, and updating the plan worked, and what elements or processes, if any, were changed; and the method and schedule to be used over the next five years to monitor, evaluate, and update the plan. The updated plan shall describe how the community will involve the public during the plan maintenance process over the next five years.

-Additional OPDR Recommended Revisions

- OPDR found the following additional update needs for the Salem NHMP as per the Local Hazard Mitigation Plan Review Crosswalk used by FEMA to evaluate whether the plan meets requirements set forth by the Disaster Mitigation Act of 2000.
 - 1. Update the hazard description information and how they affect the City of Salem
 - 2. Include probability assessments and update vulnerability assessments into the Salem NHMP.
 - 3. Elaborate on the impacts of the natural hazards on the community.
 - 4. Include measures for monitoring, updating, and evaluating the plan
 - 5. Describe how the NHMP will be integrated into existing plans and policies in the City of Salem.

Additional information for the NHMP includes:

- 1. Updating hazard history
- 2. Revisiting goals, objectives, and action items.
- 3. Develop new action items as needed.



- This information will be gathered through three meetings: March 19 (Hazard History), April 30 (Goals and Action Items), May (date tbd, Final Review of Plan Changes)
- 3. Hazard Identification and Vulnerability/Probability Assessment

PDR staff facilitated a hazard identification workshop with committee members to discuss new hazard events that have occurred since 2003 in the City of Salem. In addition, committee members discussed probability and vulnerability assessments for each natural hazard that will be included in the update natural hazard mitigation plan.

Flood

<u>History</u>

- -December 1996 and 1997, localized flooding, especially in Keizer
- -Every year there is an opportunity for flooding
- -Flooding issues on McGilchrist Avenue & Pringle Road

Mitigation/Improvement Activities

- -Salem has Digital FIRM maps
- -Salem will be joining the Community Rating System (CRS) program in May as an 8
- -Installed a flood alarm system
- -Weekly updates on water levels are held in the spring

Projects Completed

-ODOT raised I-5 at Mill Creek to avoid flooding

-Bridge replacements at Mill Creek and Center Street, Shelton at 25th

-Annual stream cleanups and Public Service Announcements help with education and outreach to stop people from throwing waste into the rivers

-Completed road projects to upsize culverts in West Salem

-Salem is experimenting with using Local Improvement Districts to pay for capital improvements in neighborhoods

-Salem has pre-established sandbag areas for the public

-

Plans and Agreements

-There is an evacuation plan in place for the mobile home park

-The City of Salem has mutual aid agreements between other cities, Marion County, and the State.

Stormwater Runoff

-Completed a regional stormwater detention study

-Had extensive public involvement in the development of the stormwater plan

-There is a stormwater design standard

-Salem has a Stormwater Master Plan, completed in 2000

-The Fairview development in Northwest Salem has porous pavements on the 32 acre development to lessen stormwater runoff



WWW.OREGONSHOWCASE.ORG

-The Hospital has issues with debris piling up in Pringle Creek. They are currently building a new hospital across the street further away from flood issues. New construction ideas include parking on the first floors and building a retention wall along Pringle Creek

Flood Probability/Vulnerability Assessments Probability-High Vulnerability-High

Landslide

Hazard History

-2005, South River Road landslide, near S. Owen Street, blocked off traffic to the city, but no homes were damaged.

Mitigation Activities/Improvements

-Heath Street tried to develop a mitigation plan but the homeowners did not buy in to it -Inclinometers are used to measure movements of hillsides, such as along Heath Street

-Reservoirs are being relocated away from landslide hazards

-Salem has an Erosion Control Program developed in 2001 (SRC 75) and a Landslides Hazard Ordinance developed in 2000 (SRC 69)

-Salem has a Tree Preservation program

Landslide Probability/Vulnerability Assessments Probability-High Vulnerability-Medium

Earthquake

Hazard History No new earthquake hazard history

Mitigation Activities/Improvements

-Reservoirs and pump stations are vulnerable, and some reinforcements are being completed for these -The fire station was recently upgraded to fit seismic codes

-Community Emergency Response Teams (CERT) currently do outreach activities with neighborhoods, including earthquake information.

-Salem is building a new computer center to meet seismic codes

-The Operations Section (managed by Public Works) has a new Emergency Operations Center (EOC)

-The 911 and EOC moved out of City Hall to a new seismically safe location

-Most unreinforced masonry buildings downtown are being brought up to current code when they are redeveloped

-Bridges are being upgraded to meet seismic codes, nine upgrades completed in the past four years -The new hospital will be built to seismic standards

-Seven schools are vulnerable to earthquakes and have a high collapse potential. The school district has a construction plan and is working on a bond to complete deferred maintenance and upgrading schools.



Earthquake Probability/Vulnerability Assessments Probability-High Vulnerability-High

Volcano

Hazard History 2004 a small eruption on Mt. St. Helens.

Mitigation Activities/Improvements

-Air handling systems at critical facilities are vulnerable to ash infiltration -There's a pre-treatment facility at the water treatment plan to prevent ash from infiltrating the water. -An eruption from Mt. Jefferson could present future drinking water issues.

<u>Volcano Probability/Vulnerability Assessments</u> Probability-Moderate Vulnerability-Low

Wind and Ice Storms

Hazard History

-In 1968 a large snow event in Salem

-December 2007 had a windstorm, no significant damage

-Snow and ice are annual occasions in the City of Salem, most recent in January 2008

Mitigation Activities/Improvements

-Salem has tree removal programs in place, and utilities are responsible for trimming trees near power lines

-Salem has a City Arborist who helps manage trees on city properties

-Utilities are preferred to be placed underground, but this is not a requirement

-A tree ordinance regulates how trees are maintained

-The new radio tower on Skyline Road is designed for high winds

-Backup generators are installed at critical facilities including the EOC and pump stations in case of a power outage.

-The downtown utility master plan is currently being developed.

Wind and Ice Storm Probability/Vulnerability Assessment Probability-High Vulnerability-High

Urban Wildland Interface Fire

Hazard History

-A constant medium-threat level in South Salem, especially around the Salem/Turner area



<u>Mitigation Activities/Improvements</u> -CWPP completed for Marion County in 2007 -Polk County currently developing a CWPP -Chief Parrot would have ideas for mitigation activities -Improve water access to West Salem by taking over Illahe and Orchard Heights private water for fire fighting.

<u>Wildfire Probability/Vulnerability Assessments</u> Probability-Moderate Vulnerability-Moderate

Hazardous Materials

Hazard History

-2004 a gas spill occurred near the Detroit Reservoir

-1976 a major oil spill occurred at Boise Cascade into the Willamette River

-Trucks and vehicles on I-5 have had spills in the past, Chief Godfrey and Parrot would have a history on events.

Mitigation Activities/Improvements

-A large railroad goes through the Capital area carrying hazardous materials, a potential mitigation action would be to keep people off and away from the railroads -Hazardous materials spills largely impact transportation

<u>Hazardous Materials Probability/Vulnerability Assessments</u> Probability-Moderate Vulnerability-High

Drought

<u>Hazard History</u> -Summer time, some years with low stream flows. -Problems occurs when reservoirs dump large quantities of water as they did in 2001

Mitigation Activities/Improvements

-Completing water conservation outreach within the community through the City's Environmental Protection Assistance Program

Drought Probability/Vulnerability Assessments Probability-Moderate Vulnerability-Moderate

4. Next Steps: Goals & Action Items Meeting

The next meeting will discuss the goals and action items for the plan, and committee members agreed to meet on April 30 from 9 to 12 at the Salem Emergency Operations Center.

	Natural Hazard	s Mitigation Work Group		
Attendance	Location: Salem EOC	3/19/2008	Time: 1400	
First	Last	Agency	<u> </u>	Email
ROGETZ	STEVENSON	CLEY OF SALON BIGHT	503-763-3331	BABIERSON CONVORSE BM. N.C. (
Ciff	Sabin	Salem Huspital	503-561-3074	Cliff. Sabin @Salembospital, Org
Kinsta	Dillon	Partnership UD	541.346.3588	Kristam@ vorcgon.edu
Lyle	GembAla	SAlem Police	503-588-6026	IgembA/A@ atypfsalem. net
MIKE	GOTTERBA	PUBLIC WONKS	503 588 6347	MGOTTES bald CITYOF SALPM. NO T
Sava	Jondall	City Public Works	503 588 6211	Sjondahl@ cityof salem. net
BELE	BISHOP	CZAY PLANNIZNIA DEC.	503-588-6173 +758	bbishpe city of salem. not
Grender	Passelwer	PDR	541-346-0259	adriamp @ woregonedu
Kin	Roley	City of sola	503-588-6386	Koolow a scitudselin enet
Steve	Dawns	Scien Public evenus	503-588-6211	Sdaws C Chy of Schem. nel.
JOITW	VANDERZANDEN	MCEM	503-365-313	3 JVanderzander@ Co. MARION. OR
L-		· · · · · · · · · · · · · · · · · · ·		
	· ·			
	· ·			

JOE PARROT INTERVIEW SUMMARY-APRIL 16, 2008, 10 AM

WILDFIRE

WILDFIRE HISTORY EVENTS

-most of the fires are brush fires in wild lands on the fringes of Salem and are minor and insignificant, wildfire not considered to be a big risk -vulnerable areas have been mapped by the Oregon Department of Forestry's Community at Risk Program where they mapped the entire state for areas at risk to wildfires. The red shaded areas are areas with the highest scores -The most significant wildfire occurrence was 2 years ago with a 14 acre field fire in south Salem by Battle Creek Road south of Keubler, started by fireworks.

VULNERABLE WILDFIRE LOCATIONS AROUND SALEM

-The areas with the highest risk have the highest slopes, the right fuels, and values at risk (meaning valuable property adjacent to wildland areas) -Areas at risk are indicated on the ODF's Community's at Risk Program maps -The wildland urban interface areas are within the South Salem Fire Department's jurisdiction

CURRENT AND FUTURE MITIGATION ACTIVITIES

-Currently not doing much mitigation for wildfire -Future mitigation activities would be limited to educational efforts, which are outlined in the CWPP, such as promoting 100 ft defensible space, fuels modification or reduction, and promoting noncombustible roofs. Currently there are few worries about roofing material.

COMMUNITY WILDFIRE PROTECTION PLANS

-Marion County currently has a wildland urban plan which discusses the Salem area and provides action items for wildfire mitigation.

-The plan largely speaks to the county public education efforts outside of the city

GROWTH AND DEVELOPMENT

Growth and development currently reduce the risk of wildfire events in Salem because they destroy the natural landscape of the area. Salem does not develop in a pattern with isolated homes throughout the forest, but in a more cohesive pattern.

HAZARDOUS MATERIALS

HAZARDOUS MATERIALS HISTORY

-Hazardous materials occur in the City of Salem on a regular basis, they are all over the city all the time

-Will send a report documenting hazardous materials incidents over the past five years

VULNERABLE AREAS

-Trucking routes along I-5 and Highway 20 that run through Salem are vulnerable because there are no restrictions placed on these routes. -Two railroad lines run through the City of Salem, one to the west is not an issue because they usually haul logs, but the one that runs through downtown Salem near the Capital area are a concern because there are approximately 12,000 cars of hazardous materials that pass through the city each year. -Fixed facilities with hazardous materials are located throughout the City

MITIGATION ACTIVITIES

CURRENT

-The City enforces the International Fire Code which regulates building construction and activities related to hazardous materials -Hazardous inspections occur in homes and in businesses -State has a reporting program that businesses with hazardous materials must discuss, and the Fire Department has the authority to regulate hazardous materials.

-DEQ also has their own program for regulating hazardous materials

FUTURE

-There are mitigation opportunities for alert and warning systems -The city is in the process of developing a reverse 9-11 system where the city can use GIS to plot areas at risk and make phone calls to affected areas and institute shelter in place to protect people from hazardous materials. -Emergency Management has been involved in mapping flood prope areas to

-Emergency Management has been involved in mapping flood prone areas to provide improved warning systems should flood levels rise

IMPACTS OF GROWTH AND DEVELOPMENT

-growth and development increase the consequences of hazardous materials events because as the population grows, the more people are at risk. They are working to a) prevent events from happening through monitoring, b) know what to do if something should happen, and c) have an effective response system in place.



Meeting:Salem NHMP Update Meeting # 2Date:April 30, 2008Time:9 am to 12 pmLocation:Salem Emergency Operations Center

AGENDA

1.	Introductions	(5 minutes)
2.	Meeting Overview (Gregoor)	(5 minutes)
3.	Discussion/evaluation of Goals and Objectives (Gregoor)	(20 minutes)
4.	Evaluation of current action items (Nick & Gregoor)	(60 minutes)
Br	eak	(10 minutes)
5.	Presentation of new Action Items (Nick & Gregoor)	(60 minutes)
6.	Presentation of Plan Maintenance and Implementation (Nick)	(10 minutes)
7.	Next Steps: Development of Plan and Final Plan Review	(10 minutes)



Meeting: Salem NHMP Update Meeting # 2

Date: April 30, 2008

Time: 9 am to 12 pm

Location: Salem Emergency Operations Center

AGENDA

1. Introductions

Good morning, welcome to the second meeting for the Salem Natural Hazard Mitigation Plan update. For those not here previously, my name is Gregoor Passchier and I am a research intern that works for the Partnership for Disaster Resilience who has been contracted to complete the update to the Mitigation Plan. We are an organization based at the University of Oregon that has completed Natural Hazard Planning work throughout the State of Oregon. The update Salem is completing is a FEMA requirement that must be completed every five years for communities to remain eligible for PDM and FMA funding. Our goal is to send a completed mitigation plan update to FEMA by the end of May. But before we begin I would like to go around the room and have everyone introduce themselves.

2. Meeting Overview

There are four main objectives we would like to discuss today. They are:

- Look at current plan goals and objectives and develop any updates to these goals.
- Look at plan action items and the current status of these action items
- Review new action items the Partnership developed based on our previous meeting and through conversations with staff
- Discuss plan maintenance and implementation measures the Partnership has developed based on information in the old plan and other ideas we incorporated into the plan.
- 3. Discussion/evaluation of Goals and Objectives (Gregoor)
 - Plan goals serve to guide the direction of future activities aimed at reducing risk and preventing future loss from natural hazard events.
 - The Salem NHMP identified three main goals each with their own objectives. They are indicated on your handout. In bold below each objective are the current plan's action items that correspond with each of the goals. For a goal and objective to be effective and implemented, it should have a corresponding action item.
 - What we would like to talk about today is whether the goals and objectives listed here are still relevant to the plan or if they have been met and should be revised. If so, we need to identify new objectives and goals. I would like to go through each of the goals and the objectives and ask you whether any of these have been met, and if so, we need to establish new goals and objectives for the plan.
 - In evaluating the goals, we came up with two areas that we feel should be addressed:
 - There is one objective that does not have an action item associated with it. All goals and objectives should have an action item designed to implement it, so we either need to develop an action item addressing this objective, or we need to get rid of this objective and formulate a new one.
 - In addition, there are no goals addressing economic loss and discussing collaboration among different agencies, and we feel there should be some goals that address these issues
 - Potential goals include: (1) Increase cooperation and coordination among local, state, and federal agencies. (2) Reduce economic loss.
 - Three questions:

(5 minutes)

(5 minutes)

(20 minutes)



(50 minutes)

- Do you feel that the goals are relevant to the plan and should they be changed?
- Do you feel that the objectives are relevant, and useful, and should they be kept or altered?
- What are your thoughts on Objective 2.1?
- What do you think about the two additional goals we are proposing?
 - If you agree, we can add these as goals and we will assign you homework to send us
 objectives for the mitigation plan.

4. Evaluation of current action items (Nick & Gregoor)

- The next part of the mitigation plan is to evaluate the current action items in the Salem NHMP and whether or not they have been implemented and what should be done with them in the future. Action items are an important part of a mitigation plan because they provide detailed recommendations for activities that local departments, citizens, and others can engage in.
- The Salem NHMP has 17 action items that focus on city policy actions and mitigation implementation measures.
- We took the action items in your plan and put them in the action item forms that the Partnership uses to help organize different elements of an action item and help to fill in the blanks that are missing. One of the additional benefits of having this form is that if communities want to write a grant to implement an action item, they have all the relevant information.
 - o I want to briefly explain the main components of the action item form:
 - 1. Action item statement: this is the action item statement that addresses the specific issue or natural hazard
 - 2. The statement should align with plan goals to show how the goals are being implemented in the plan.
 - 3. The rationale explains why the action item is important and provides the reasons for why you are presenting the plan
 - 4. The ideas for implementation discuss the methods a community can use to implement the action item, which could include coordinating with other organizations, etc.
 - 5. The coordinating body is the person is responsible for implementation of the action item
 - 6. The coordinating body works together with internal partners which would include local and state organizations,
 - Many of these are blank, but we will follow-up with the listed coordinating organization to determine internal partners and external partners.
 - 7. External partners which would include federal agencies
 - 8. Then there is information that discusses the timeline of events, etc.
- Now I would like to go over each of the action items one by one, together with Nick Kraemer, and see which ones have been completed and which ones have not. For those that have not been completed, we need to decide whether they should be kept in the plan, deferred, or deleted from the plan, and why? The "why" part is a FEMA requirement that needs to be included for each action item.
- There are a few blank spaces included in the form, but we will contact the coordinating organization to determine the external and internal partners needed for the action item.
- We also have to keep a tight timeline for each of these, 3 minutes per action item
- (Nick and Gregoor will review each action item one by one to review status)

Action Item # 2:

Does the current MCC stand? Is this the name that you want the committee to be?

Action Item # 5

Has there been work completed on estimating the number and types of buildings affected by natural hazards and have these been mapped?

Action Item # 6 Has the city addressed any of the ideas of implementation listed in this action item at all?

Action Item # 7 Has the city addressed any of the ideas for implementation listed in this action item?

Action Item # 16 Address all the ideas for implementation, has the city updated its vulnerability assessment of structures? If not, WE NEED TO DEVELOP A MITIGATION STRATEGY FOR UPDATING THESE FOR THE NEXT UPDATE!

Action Item # 17 Do you have a plan implementation process, or is the one that we have adequate?

Break

- 5. Presentation of new Action Items (Gregoor & Nick)
 - The next task on the agenda is to present new action items for the city of Salem, gather your input, and provide additional action items that we may be missing. The action items are part of the third handout.
 - Have a broad range of action items because FEMA requires that communities have a comprehensive range of action items that address all hazards.
 - We based these action items on information gathered from the March 19 meeting and are organized around natural hazards that can affect the city of Salem.
 - There are four main questions that we will be discussing:
 - 1. Is the action correct/relevant
 - 2. Are there any changes that we should make?
 - 3. Is the coordinating body correct?
 - 4. Who would the partners be?
 - Nick and I will briefly review these (3 minutes for each one), and if you have any input you would like to include, please let us know
- 6. Presentation of Plan Maintenance and Implementation (Nick) (30 minutes)
- 7. Next Steps: Development of Plan and Final Plan Review
 - New meeting date and time for the final plan review.
 - Update on the Maps?
 - Update on floodplain information? •



(10 minutes)

(50 minutes)

(10 minutes)

	Natural Hazard	ls Mitigation Work Group		
Attendance	Location: Salem EOC	4/30/2008	Time: 0900	
First	Last	Agency	Telephone	Email
KEGER	STREASON	CUTY OF SACION EMGN	5037633331	PS/DEMSon Lechyops ALEM. NET
Charlie	Crawford	Salen Hapital	503-561-2073	Charlie - Crawford Osalembortalme
Giegoer	Passdurer	University of Gregon	541-346-0259	adriaump Burgon edu
JOHN	VANDERZANDEN	Marion County	503-365-3133) Vanderzander @ Co. marion. O.R. us
Lyle	Gembala	Salem tolice	503-588-602	6 1genbald Deityspalem not
FRIE	BISHOP	PLANNIZNIA DON. (COS)	513-588-6173	bishope cited solom. not
Fin	holey	City of Salen A	503-588-6386	Kroley@ city ofsalm.net
Steve	Dawns		503-588-621	Solum Cc. Lycf. Schem net.
Sava	Jondall	wir PW	503-588-6211	sjondahl@cityofsalemnet
JDE	PAPPERT	CITY OF SALEM	93-588-62.45	- ``
		1		
	-			
4/30/2008				1



Meeting:Salem NHMP Update Meeting # 3Date:May 22, 2008Time:9 am to 12 pmLocation:Salem

AGENDA

- 1. Presentation and discussion of revised Plan sections
- 2. Plan approval process

(45 minutes)

(10 minutes)


Meeting:Salem NHMP Update Meeting # 3Date:May 22, 2008Time:9 am to 12 pmLocation:Salem City Hall, Community Development Conference Room #305, 555 Liberty Road SE, Salem OR 97302

AGENDA

I'd like to thank everyone for taking the time out to review the plan and provide us with comments. Today we are reviewing the changes made to the Salem NHMP. Our main goal for today is to have approval from the committee on a finalized version of the mitigation plan to send to FEMA for pre-approval. This will include a discussion of what to do with the goals and objectives, and then we will be going through the new plan section by section to review all the changes that we have made and why we made them. We need to have agreement on all changes before we can proceed further.

1. Discussion of goals and objectives

(30 minutes)

At our last meeting we talked about changing the goals and objectives for the mitigation plan. Originally the plan had three goals:

- 1. Lives should be protected.
- 2. Existing properties should be protected.
- 3. Future development should be protected.

We revised the goals to be more direct, and combined goals two and three. Our new goals are:

- 1. Protect lives.
- 2. Protect existing and future development.
- 3. Increase cooperation and coordination among stakeholders.
- 4. Reduce economic loss.
- 5. Protect the natural environment.

The question that we have is what to do with the objectives. After our last meeting, we decided that the objectives need to be revised and further simplified. We did this, but looking back at the objectives, they sound like action items that should be included in the plan. In many of the plans that we have developed, the goals usually do not include objectives because they often act like action items. That is why we proposed to only have goals in the mitigation plan.

We have two questions we need answered:

- 1. Should we include objectives in the mitigation plan or not?
- 2. If we are keeping the objectives, what do you think about the ones we have developed? Should they be revised?



4. Presentation and discussion of revised Plan sections

Now I would like to go over all of the revisions that we have made to mitigation plan, which we have documented in the attached memo. The memo serves as the primary record of revisions to the plan. We will go through each section of the mitigation plan, discuss the changes that were made, and why we made those changes to the plan. Nick and I will tag team, discussing each section.

5. Plan approval process

We will go back and make any changes to the plan needed so far. What we would like to know is if he committee or just Joe and Roger want to see a final draft before we give it to FEMA for a final review.

We intend to give this document to FEMA by the second week of June once we have made all the necessary revisions. Once FEMA has the plan, they have 45 days to review it and give us pre-approval. Unfortunately, the FEMA reviewer will be leaving from May 22 to June 24 and will not be able to provide pre-approvals of mitigation plans during that time. What this means is that we will not be getting pre-approval until June 24. Dennis Sigrist from Oregon Emergency Management has been notified of this, and he is working with FEMA to get Salem's NHMP early on in the queue and make our plan a priority.

Once we have pre-approval, the Salem City Council needs to adopt the mitigation plan by resolution. The resolution will be sent on to FEMA, and they will give their final approval to the mitigation plan.

6. Next Steps

Are there any other questions that the committee has?

(120 minutes)

(10 minutes)

(5 minutes)